Uniwersytet Mikołaja Kopernika w Toruniu Wydział Filozofii i Nauk Społecznych

> Olgierd Borowiecki nr albumu: 503171 Szkoła Doktorska Nauk Społecznych

> > Rozprawa doktorska

The conceptual framework of information processing in action

Opiekun pracy dyplomowej dr hab. Anita Pacholik-Żuromska, prof. UMK

Toruń, 2023

Table of Contents

Table of Contents			
Introdu	ction	3	
1. Th	ne description of the biological body in the physical space	14	
1.1	Information processing by the biological body	15	
1.2	Processing information from the receptors	17	
1.3	Biomechanics of the biological body	22	
1.4	Summary		
2. Co	onsciously experienced surroundings of multiple potential futures	27	
2.1	Affordances toward the potential futures		
2.2	Surroundings denote environment and niche combined		
2.3	Exploration-exploitation spectrum of affordances		
2.4	Decision-making and action execution in the striatum		
2.5	Summary	44	
3. Inj	formation processing behind the conscious experience		
3.1	Conscious, coherent, effortless experience of subjective construal	46	
3.2	Gestalt areas compute the experience		
3.3	Summary	54	
4. Tu	vo kinds of spaces for the future are stored in memory	55	
4.1	Two interpretations of the free energy principle	56	
4.2	Two memory systems for encoding the statistical regularities	62	
4.3	The entirety of memory is procedural (i.e., discrete, sequential)	70	
4.4	The memory systems cooperate and compete	74	
4.5	Summary	80	
5. Th	ne n-dimensional spaces of objects and agents		
5.1	Borders and dimensions of the spaces in the HPC-dependent network		
5.2	The spaces computed by the large-scale brain networks	89	
5.3	The inferred space is investigated in the Theory of Mind (ToM)		
5.4	Summary		
6. Po	tential applications of the conceptual framework	107	
6.1	Encoding and retrieval of information, not behavior		
6.2	Dopamine contributes to learning and motivation in the networks		
6.3	Agency over planned intentions		
6.4	Open questions		
Conclusion1			
Figures and Tables			
Bibliography			

Introduction

This work begins with the claim that animals fundamentally process physical, chemical, and biological information (Oyama, 2000). Information can be understood in many ways, such as a DNA code, action potentials propagated from the sensory receptors or activity of the large-scale brain networks. Information is thus a capacious term that is further operationalized into three information channels. Sensory information involves information "from the outside"; somatosensory information involves information about the state of the body, while vestibular information reflects the body's locomotion (Figure 1). For example, sensory information includes chemical information attracting and repealing simple organisms or visual information reaching the human eye (Milner, 2017). Somatosensory information, on the other hand, provides information such as piercing a single cell's membrane or the skin's temperature (ten Donkelaar et al., 2020), while vestibular information encloses gravitational information about a change in velocity (acceleration/deceleration) and angular momentum of the head (Day & Fitzpatrick, 2005). It is claimed here that the necessary and sufficient conditions for describing a biological body acting in the physical space are to treat it as a system processing sensory, somatosensory, and vestibular information.



Figure 1. The biological body in the physical space is fully described with sensory, somatosensory, and vestibular information. Sensory information is smell, hearing, and vision; somatosensory information is pain, pressure, temperature, and proprioception through stimulating muscles and joints; vestibular information is spatial orientation. The flow of information in these channels fully characterizes the biological body acting in the physical space.

Nevertheless, it should be observed that understanding living beings in terms of information processing systems ignores the everyday phenomenological experience even though we experience qualia, not information processing per se. To address this gap, a phenomenological experience specific to action will be analyzed. The action-oriented qualia are experienced as affordances - possibilities for action that lead to the plurality of hierarchically organized potential futures. The rich landscape of affordances is called here the **surroundings**, which reflects Gibson's (2014a) understanding of the term environment and Levontin's (2000) understanding of the term niche (**Figure 2**). The surroundings thus encompass the experience of the potential futures, such as locomotion or reaching an object, and much thicker in time actions such as finishing a book or contributing to the civilizational goals. The temporal thickness of an organism depends on its phylogeny and ontogeny, encapsulating the entirety of the organism's history (Friston, 2018). Taking action toward a specific future thus entails following affordances experienced as leading toward this future.



Figure 2. Phenomenological surroundings of multiple potential futures. At its current state (now), an organism experiences possibilities for actions toward possible ends known as affordances. These actionoriented qualia are hierarchically organized paths toward the future constituting the surroundings (rich landscape of affordances). Humans have the most robust temporal thickness ("cognitive light cone") that can reach even beyond their lifespan, i.e., people are capable of experiencing the distant future and acting upon it.

The discrepancy between processing the information in the biological system and an organism's phenomenological experience is known as the mind-body problem (J. R. Anderson, 2007; Frith, 2007; Wiese & Friston, 2021). It implies that at some point in the phylogeny, the experience of affordances became a proxy for balancing the information toward the potential futures (**Figure 3**). In the case of humans, the conscious experience is statistically independent of the flow of sensory, somatosensory, and vestibular information and arises from the information processing in the posteriorly located set of brain areas known as "gestalt areas" (Dehaene & Changeux, 2011; Lieberman, 2022; Tononi & Koch, 2015). This idea relates to the distinction between the "remembering self" that remembers and plans the future and the "experiencing self" actually living life (Kahneman & Riis, 2005; Zajchowski et al., 2017). Sensory, somatosensory, and vestibular information propagated to the gestalt areas results in experiencing the physical world. In contrast, information dissociated from these inputs that reach the gestalt areas results in experiencing the affordances toward the more distal futures "in mind".



Figure 3. Experience of surroundings. Potential futures are experienced with affordances. Once affordances are followed, they correspond to the specific flows of information in the sensory, somatosensory, and vestibular channels. The experience is computed in the gestalt areas that receive information varyingly correlated with the ongoing flow of sensory, somatosensory, and vestibular inputs.

The experience of action-oriented qualia processed by the gestalt areas includes the immediate and the far future. This conscious experience of surroundings is composed of two models: (I) the ecological-enactive model connected with the ongoing flow of information from the biological body processing physical space and (II) the Bayesian model dissociated from this flow (Bruineberg, Kiverstein, et al., 2018; Constant et al., 2021) (Figure 4). Here, these models are interpreted as biomechanical and inferred spaces computed with specialized brain networks that become conscious when coupled with the gestalt areas. The spaces are composed of information already transducted from the receptors and along the nervous system; thus, the spaces are *representational*. On the one hand, representations are directly coupled with the ongoing stream of the incoming information (i.e., perceiving representations of the encountered physical world); on the other hand, representations are decoupled from this information and denote latent causes behind the incoming information. In addition, the spaces are investigated in psychology and neuroscience as memory systems, which in advanced organisms are divided into "procedural", and "declarative" (Poldrack & Packard, 2003a; Squire, 2004; Ullman, 2016). The procedural memory makes associations within the ongoing flow of information, while the declarative memory makes interdimensional associations (Hikosaka et al., 1999; Keele et al., 2003). The architecture of information processing behind the memory systems reveal the insights about organization of the brain networks.



Figure 4. Two models for the future. Experience of surroundings originates from highly processed information entering the gestalt areas. Thus, the entirety of the experience is representational. The content for the sake of action involves experiencing the actual flow of information (e.g., driving a car) or being dissociated from it (e.g., planning a business meeting while driving a car). The latter case is exemplified as an inference about the intentional states of other agents that compose the social world.

A biological body in the physical space can be described in terms of exchanging kinetic and potential energies in the biomechanical space (Knudson, 2021) (**Figure 5a**). The dimensions of the biomechanical space are enclosed in the sensory, somatosensory,

and vestibular information. Complementary, the dimensions of the inferred space are enclosed in agents' social interactions (Gollwitzer, 1999; Pacherie, 2008; Tomasello, 2018) (**Figure 5b**). The brain networks that compute the two spaces train each other in organism's ontogeny (M. Chen et al., 2022). The action-oriented experience of the surroundings is mediated by the "control" network that partially overlaps with the gestalt areas (Uddin et al., 2019), and gates the access of the networks calculating the biomechanical and inferred n-dimensional spaces to the gestalt areas (Dixon et al., 2018). The creation of the inferred space has been experimentally investigated in research on the Theory of Mind (J. R. Anderson et al., 2004; Frith, 2007; H. L. Gallagher & Frith, 2003; Goel et al., 1995).



Figure 5. Objects and agents exchange information in time. The x and z axes denote the position in the space of information exchange. a) Example of the billiard balls exchanging kinetic and potential energies in the physical space. b) Example of a stationary agent (green), a predator (blue), and a prey (red) exchanging information in the n-dimensional inferred space. The dimensions of the latter space depend on the interactions between the agents that are modeled with their potential future actions (e.g., how agile other agents are).

The information processing behind the experience of the surroundings involves simultaneous and independent encoding and retrieval of the biomechanical and inferred spaces in the respective networks. Such simultaneous and independent information processing is identified in multiple dual theories of memory developed in psychology and neuroscience (Dolan & Dayan, 2013; Geerts et al., 2020a; Poldrack & Packard, 2003a; Squire, 1992). Here, these insights are reviewed in light of the large-scale brain networks that underpin the experience by processing information in a highly organized way (M. Chen et al., 2022; Dixon et al., 2018; C. Murphy et al., 2018; Uddin et al., 2019). The phylogenetic and ontogenetic interactions between the networks constitute the information flow in the entire system. To explore the organization of the information processing are postulated:

- Information processing is **distributed**.
- Information processing is **future-oriented**.
- Information processing is **memory-guided**.

The idea of **distributed information processing** has emerged in anthropology with the recognition of the social aspects of human cognition, presented by Vygotsky and Cole (1978), and Minsky (1988). Later, Hutchins (2000) distinguished at least three types of distributed information processing. Firstly, information processing is often distributed among members of a social group. Secondly, information processing requires the animal to coordinate the internal and external states. Lastly, earlier events can impact later events, resulting in the distribution of processes through time (Hutchins, 2000). Put simply, the distribution of the information processing systems means that the systems enclose physical objects and social agents and are embedded in time.

The **future-oriented information processing** is a consequence of the evolutionary processes. Adaptive systems anticipate their future to navigate it effectively (Bruineberg, Kiverstein, et al., 2018; A. Clark, 2013; Friston & Stephan, 2007). Navigating the surroundings involves navigating toward and away from specific futures, which is implemented by two evolutionary ancient motivational circuits: (I) reward-based and (II) misery-avoidant, processed by the nucleus accumbens and the amygdala, respectively (Loonen & Ivanova, 2015). Orientation toward the future is fundamental, given that surprise may imply receiving somatosensory information that will result in an organism's death.

The memory-guided information processing is a consequence of orientation toward the future. To construct accurate predictions, the system must store information about the already encountered statistical regularities. Memory is organized hierarchically, which is reflected in numerous frameworks such as Bayesian inference (Friston & Stephan, 2007), hierarchical processing of affordances in the Feedback Control Theory (Pezzulo & Cisek, 2016), a capacity to hold an optimal grip in the field of affordances in the Skilled Intentionality Framework (Rietveld et al., 2018), also as Cognitive Control given the broad context (Badre, 2008) and executive control over selecting actions and thoughts relative to internal intentional states (Koechlin & Summerfield, 2007). Complex hierarchical actions are understood as supervised movement sequences (Desrochers et al., 2016), which require cognitive control over action selection (Badre et al., 2009). In a branch of study focusing on the hippocampus, such actions are named "cognitive skills" (Cohen & Squire, 1980) and are associated with planning and counterfactual thinking (Squire, 2009b). The hierarchical organization of actions is also articulated in psychology and economy as a capacity for delayed gratification (Kahneman, 2013). Thus, this type of memory organization is widely recognized across the literature.

The three principles lead to the developing of a coherent and interdisciplinary conceptual framework of information processing in action. Fundamentally, organisms are systems processing information according to the above principles. The experience of the surroundings originates in the gestalt areas that receive information from the biomechanical space computed in the "attention" network and inferred space calculated in the "default" network decoupled from the incoming flow. The networks shape each other's ontogenetic development (M. Chen et al., 2022) and are varyingly active depending on task demands (Dixon et al., 2018). Thus, the spaces computed in the networks significantly influence each other. Inference of the space of hidden causes behind the experience changes the meaning of the incoming information (Babayan et al., 2018; M. Chen et al., 2022; Nour et al., 2018), which is captured as a subjectivity of the experienced meaning (Ross & Ward, 1996). Given the complexity of the mutual interactions of the networks computing the spaces, this work simplifies the biomechanical inferred spaces entering the gestalt areas as the experience of the surroundings (rich landscape of affordances). The information processing the both spaces is fundamentally future-oriented, and thus also hierarchical, as the hierarchical organization of the predictions is more efficient.

The entirety of the conscious experience is representational as it originates from the processed information reaching the gestalt areas. The conscious experience for action is an experience of the surroundings that is computed from the biomechanical and inferred spaces entering the gestalt areas (**Figure 6**). The biomechanical space is calculated with the "attention" network (Ptak et al., 2017; Uddin et al., 2019). The inferred space is computed with the "default" network, decoupled from inputs (C. Murphy et al., 2018) and processing social information (Buckner & DiNicola, 2019; Li et al., 2014; Uddin et al., 2019; Yeshurun et al., 2021). During the ontogenetic development, the networks train each other in multiple ways (M. Chen et al., 2022; Dixon et al., 2017; Gao et al., 2013; Spreng et al., 2016). The activity of the two networks is controlled with a third "control" network, which includes some of the gestalt areas and mediates executive functions (Dixon et al., 2018; Uddin et al., 2019). The computations in the networks parallel the information processin in the two memory systems.



Figure 6. The conceptual framework of information processing in action. The conscious experience of the surroundings is computed from the biomechanical and inferred spaces. The former space is computed in the "attention" network, the latter in the "default" network. The dimensions of the biomechanical spaces consist of the sensory, somatosensory, and vestibular information, while the dimensions of the inferred spaces is abstract associations such as social interactions. The networks interact independently from the gestalt areas thus, their training during ontogeny remains to be investigated.

The conceptual framework developed in this work attempts to establish the interdisciplinary dialogue between cognitive sciences. While scholars trained in narrow

specializations investigate the structure of problem spaces, various kinds of attention and memory systems, semantics and syntax, and neurotransmitter systems; only some attempt to relate these research fields. As a result, the deep philosophical questions about the nature of existence, reality, or consciousness still need to be answered (Vervaeke et al., 2017). These questions arguably can only be asked understandably with coherent interdisciplinary terms. This is vital since existence, reality, or consciousness transcend things investigated in specific disciplines. Yet, no definition of existence, reality, or consciousness would enclose all these perspectives. This interdisciplinarity is needed to understand our place in the world and ourselves in terms of biology and psychology. The first step toward this understanding is developing a conceptual framework that would connect the disciplines in a genuinely interdisciplinary way. Once this understanding is achieved, more accurate questions concerning cognitive processes can be asked, potentially leading to a more in-depth understanding of human nature.

The search for interdisciplinarity in exploring what cognition is started its formal history at the first meeting of the Cognitive Science Society in 1979. At its dawn, the project aimed to establish a collaboration between scholars from various disciplies to understand how the mind emerges from information processing in the brain. The disciplines recognized in the society included computer science, which had been heavily influenced by cybernetics from the 1930s and '40s. Pioneers such as von Neumann, Minsky, and Turing contributed to developing the first computers. Today, cybernetics ideas are implemented in optimal control theory (Todorov, 2004) or predictive accounts of the brain (Friston, 2012). In psychology, flexible latent learning (Tolman, 1948) and language creativity (Chomsky, 1959) were contrasted with inflexible behaviorist accounts. Psychological research on memory began in 1952 with studies on the patient H.M. by Brenda Milner, which led to the discovery of two separate long-term memory systems known as declarative and procedural memory (Squire, 2009a). These observations inspired the progress in neuroscience, which has been experiencing tremendous development since the discovery of the DNA structure in 1953. In linguistics and anthropology, scholars such as Sapir and Whorf recognized that language shapes thoughts (Wolff & Holmes, 2011) and that understanding a given ethnic group requires cross-cultural translation of a broader context (Buden et al., 2009). Philosophy makes one ask fundamental questions and provides tools for reasoning such as abductive cognition, reasoning from first principles, and ontological methods for categorizing terms

used in particular disciplines and identifying which tokens of observations shall be interpreted as similar enough to be classified into a specific category.

This work connects the ideas from various disciplines of cognitive sciences to develop a broad understanding of human mental processes. In the last century, physics described general relativity and quantum theories, chemistry classified atoms and substances, and biology discovered evolution. The author urges similar efforts in sciences investigating cognitive processes and profound questions about the mind, consciousness, and our place in the universe. This effort requires recruiting onboard scholars from numerous disciplines as our civilization develops at an unprecedented pace toward the unknown future. We ought to be able to formulate precisely the deep existential questions and launch an effort in search of the answers.

The need is existential. To identify the direction where the next generations will find themselves, we must classify the mechanisms that influence our decisions and actions on numerous levels of complexity, from gene expression to electrical and chemical communication to social interactions. These levels are connected because each of us embodies a simultaneous existence on all these levels, yet they still need to be comprehensively addressed in scientific writings.

Today's fragmented knowledge is hardly accessible, given the specificity of language. The sheer number of concepts used to explain cognitive phenomena is baffling. The same phenomenon is captured with different labels, and the same labels denote diverse mechanisms. Navigating through this equivocacy requires such training that the sense of understanding is lost along the way.

The need for understanding has been pursued by such people as Hume, Descartes, Nietzsche, Heidegger, Kant, Jung, Darwin, James, Turing, Piaget, and Gibson. Naturally, the progress in this direction is ongoing. Contemporary scholars such as Sapolsky, Letvin, Wolfram, Vervaeke, and Bach pursue these questions. Nevertheless, this work argues that more is needed. Thus, this need is addressed by bridging the perspectives in a search for a genuine understanding of existence.

The framework developed here tackles the mind-body problem by treating animals as systems processing information in a **distributed**, **future-oriented**, and **memory-guided** manner. Then, a subset of this information processing related to action acquisition and execution is reviewed. The conscious experience of surroundings for action is linked to the information processing in memory systems, large-scale brain networks, and neurotransmitters' modulation of this information processing. The framework explains how one infers everyday beliefs as part of one's experience by relating the conscious experience to the activity of brain networks. The review of the brain networks explores the information exchange within and between the brain networks. The analysis of the phenomenological evidence links the experience to the brain's activity, finding a potential home for the mind.

1. The description of the biological body in the physical space

The biological body in the physical space is operationalized in this chapter according to biomechanical and neuroscientific evidence. The processed information is conveyed by the sensory, somatosensory, and vestibular receptors that an organism is equipped with – its Umwelt. These information-processing channels are investigated in biomechanics (Knudson, 2021) and are grounded in animal brain regions underlying computation of body schema, tool use, and peripersonal spaces (Làdavas, 2002; Lopez, 2015; Maravita & Iriki, 2004; Serino, 2019), including parietal, frontal, and subcortical areas. As a result, the observations of the biomechanics of living beings and the neurobiological information behind these overt behaviors comprehensively describe a biological body in the physical space.

1.1 Information processing by the biological body

A biological agent is described with three information processing channels: sensory, vestibular, and somatosensory (Mergner & Rosemeier, 1998; Nashner, 2014) (Table 1).

Modality	Dimensions	Experience		
Somatosensory channel				
	sharp cutting pain	sharp cutting pain		
Pain	dull burning pain dull burning pain			
	deep aching pain	deep aching pain		
Tomporatura	warm/hot	warm/hot		
i emperature	cool/cold	cool/cold		
	itch/tickle and crude touch	itch/tickle and crude touch		
		touch		
Touch	discriminative touch	pressure		
		flutter		
		vibration		
		muscle length		
	Position: Static Forces	muscle tension		
		joint pressure		
Proprioception	Movement: Dynamic Forces	muscle length		
		muscle tension		
		joint pressure		
		joint angle		
Vestibular channel				
Spatial	linear motion	acceleration / deceleration		
orientation	angular motion	rotating head		
orientation	tilt	nodding head		
	vestibular-ocular reflex	keeping the gaze fixed despite body		
Balance	vestionini ocului fellex	movements		
200000	vestibulospinal reflex	keeping the mass' centre of the body		
		balanced		
Sensory channel				
	olfaction	smell		
Chemical	gustation			
	trigeminal chemosensory system	detection of chemicals with smell		
Mechanical	cochlea's structures' vibration	sound		
Electromagnetic	electromagnetic wave	VISION		
Electric	electroreception	electric field		
Magnetic	magnetoreception	magnetic field		

Table 1. Information describing the biological body in the physical space. Organisms process information divided into three channels: somatosensory, vestibular, and sensory. The information processing in the somatosensory channel results in the experiences of specific states of the body. The information processing in the vestibular channel results in the experiences of the movement of the body and body parts. The information processing in the sensory channel results in the experience of the external world (adapted from (Dougherty, 2020).

The **somatosensory** channel processes pain, pressure, temperature, vibration, and proprioception (Raju & Tadi, 2021). Physiological pathways for conveying the somatosensory input are robust and reach many brain networks (McGlone & Reilly,

2010; ten Donkelaar et al., 2020). Here, the somatosensory information is recognized as crucial for an organism's locomotion as it is a feedback signal (Frigon et al., 2021) processed initially in the primary somatosensory cortex (Raju & Tadi, 2021). This information is particularly relevant from the perspective of future-oriented information processing, as an animal acts based on the *predicted* somatosensory information because somatosensory surprise may imply death.

The **vestibular** input reaches multiple brain regions, including parietal and associative areas (Lopez, 2015) and the hippocampus (P. F. Smith et al., 2005; Stackman et al., 2002). The vestibular channel starts in the vestibular system located in the inner ear. It is crucial for maintaining balance, perspective, and equilibrium (e.g., which way is "up") (Day & Fitzpatrick, 2005). The system senses gravity and acceleration/deceleration. It is responsible for the vestibular-ocular reflex (keeping the gaze fixed despite body movements) and the vestibular system tracks the tilt and rotation of the head and the body's horizontal and vertical acceleration, which follows information about the body's kinetics.

The **sensory** channel detects information from the physical world with specialized receptors that strongly depend on an organism's phylogeny. For example, chemical receptors are experienced as smell and taste, sensory mechanical receptors are experienced as sound and electric and magnetic receptors are found in non-human organisms. Primates rely primarily on the electromagnetic receptors tuned to the visible spectrum of the electromagnetic wave (Barton, 1998). This input is part of the scaffold for postural control, motion perception, and vestibular and somatosensory channels (Mergner & Rosemeier, 1998).

1.2 Processing information from the receptors

The claim that the three information channels fully describe the information processing performed by a biological body is supported by the architecture of information processing in the brain. For example, in primates, visual information diverges into two interconnected streams of information processing: dorsal and ventral (Milner, 2017). The dorsal visual stream conveys information about the location of objects and terminates in the parietal associative areas, where it is combined with somatosensory information from the rostrally located somatosensory cortex (Avillac et al., 2005, 2007; Iriki & Taoka, 2012) (**Figure 7**). The auditory and vestibular information also reaches this region of the brain (Day & Fitzpatrick, 2005; Lopez, 2015; Regenbogen et al., 2018).



Figure 7. Parietal associative areas. Intraparietal sulcus (green) combines somatosensory (orange), visual (blue), auditory (not shown), and vestibular (not shown) information into the multimodal computation of space around particular body parts (Lopez & Blanke, 2011; Serino, 2019), the space elongates with tools (Maravita & Iriki, 2004). The visual cortex is enclosed into one of the large-scale brain networks, including striatal and extrastriatal areas processing visual information (Uddin et al., 2019) from where visual streams originate (Milner, 2017).

The information from the channels is jointly processed in the associative brain areas. The intraparietal sulcus combines somatosensory and visual streams at the level of single cells called bimodal cells (Maravita & Iriki, 2004). Some studies suggest that these cells should be called multimodal cells as they express sensitivity to auditory information (Duhamel et al., 1998; Guipponi et al., 2013). Thus, the associative brain areas process information about the body's position relative to the external world by combining sensory and somatosensory channels. Additional evidence suggests that associative regions of the parietal associative area associates the entirety of the information about the biological body in physical space. Associative areas are found in other brain regions, such as the prefrontal regions (Lopez, 2015; Preuss & Wise, 2022). Given that the parietal and frontal areas are functionally coupled within the large-scale brain networks (Uddin et al., 2019), the associative areas are understood as computing the information about the biological body in the physical space by combining sensory, somatosensory, and vestibular inputs.

Research on the associative areas in the parietal cortex reveals that information processing of the space around the body is highly organized into distinct peripersonal spaces (Serino, 2019). These peri-spaces are attached to specific body parts, such as peri-arm and peri-hand (Graziano & Gross, 1993; Guipponi et al., 2013; Làdavas, 2002; Niu et al., 2021), and their activity is gradual and action-oriented (Bufacchi & Iannetti, 2018). In other words, peri-spaces are parts of the sensory (visual) experience of objects capable of generating somatosensory and vestibular feedback. These computations are subjective to the animal's mechanical properties (e.g., speed, acceleration, inertia) (Bufacchi & Iannetti, 2018; Galli et al., 2015). For example, modifying vestibular information by changes to animals' characteristics of potential vectors of movement (e.g., when driving a car or a wheelchair) changes the neuronal computation of the peri-spaces. This points out that modifying the capacity to move results in changes in the calculation of visual information. This is investigated in experiments on tool use in primates and humans.

Tools are modifications to an animal's biomechanical properties distributed on the spectrum between translating output forces smoothly (e.g., pushing, pulling, rotating) and abruptly (e.g., hitting). Experiments on macaque monkeys reveal multimodal cells sensitive to visual and tactile information (S. Gallagher, 1986; Maravita & Iriki, 2004). Touching the hand/arm and shining light close to the hand/arm elicit activity of the same region specific to the hand/arm. A skillful grip on a tool elongates the cells' visual receptive field with the tool's range (Aglioti et al., 1996; Berti & Frassinetti, 2000). Such associative information computation is identified in numerous parts of the macaque brain: the putamen (Graziano & Gross, 1993), superior colliculus (Wallace et al., 1996), parietal (Duhamel et al., 1998; Graziano & Gross, 1993; Hyva[¬]rinen, 1981), and frontal cortices (Graziano & Gross, 1993; Graziano et al., 1999; Rizzolatti et al., 1981). These observations have been partially translated into research on humans (Berti & Frassinetti, 2000; Gentile et al., 2011; Guipponi et al., 2013; Guterstam et al., 2015). For example, changes in the peri-spaces have been identified in patients suffering from unilateral neglect and participants navigating with a wheelchair.

The associative processing of information also occurs at the level of large-scale brain networks (Uddin et al., 2019). One of the networks, the Pericentral Network (PN), is located on both sides of the central sulcus and combines information from motor and somatosensory cortices, as shown in **Figure 8**.



Figure 8. The pericentral network. One of the large-scale brain networks that jointly process motor information is the Pericentral Network (PN), enclosing the Motor Cortex (MC) and the somatosensory cortex (SSC) (Uddin et al., 2019). The information processing in the PN contributes to movement acquisition and execution and is discussed further in section 5.2.

The organization of the PN can be further divided into three sections specialized in processing information related to particular body parts: ventral (mouth), lateral (hands), and dorsomedial (feet) (Gordon et al., 2023; Graziano, 2016; Raju & Tadi, 2021) (**Figure 9**). The PN contributes to various motor behaviors, known as species-specific fixed motor patterns (Lorenz, 1981), movements of body parts (limbs, mouth) (Graziano, 2016), and stereotypic behaviors (Berridge et al., 2005; Reason, 1990).



Figure 9. Frontal motor areas. The motor cortex (MC) and the premotor cortex (PMC) compute the movements of the mouth, hands, and limbs in the ventral-dorsal organization of information processing (Graziano, 2016). Such information processing organization is reflected in the architecture of the entirety of the PN (Gordon et al., 2023; Raju & Tadi, 2021). The literature points to an orthogonal division spanning the rostral-caudal axis of the motor areas: anteriorly located PMC computes multi-limb actions, and posteriorly located MC is preferentially active in single-limb movements (Graziano, 2016). Further, the PMC prepares activities related to external cues (Rizzolatti et al., 1996, 2002), whereas the supplementary motor area (SMA) extending toward the medial part of the brain calculates the preparation of the multi-limb actions linked to internal prompts (Nachev et al., 2008).

It should be noted that the precision of action increases ventrally as mouth and pharynx movements produce high-resolution articulatory gestures (actions making sound), which have been identified in humans' ventral precentral gyrus with a direct electrophysiological recording of the brain of epileptic patients (Mugler et al., 2018). Movement of the pharynx is thus recognized as mediated by ventral sections of the motor (Graziano, 2016) and somatosensory (Raju & Tadi, 2021) cortices. Accordingly, communicative sounds such as deontic actions (Constant et al., 2019) are interpreted here as actions that, in principle, may rely on the information processing within the PN.

1.3 Biomechanics of the biological body

Biological bodies existing in the physical space are investigated in a branch of physics called biomechanics (Knudson, 2021). Forces acting on a body (which causes, inhibits, facilitates, or modifies movement) are studied as kinetics. In contrast, spatiotemporal characteristics of motion in the Cartesian coordinate system without regard to the causative forces are studied as kinematics (Jensen, 2005; McLester & St. Pierre, 2008; Serway & Jewett, 2018). In biomechanics, the body is modeled as a free-body diagram that captures a system of interest (e.g., an arm or a body) and the degrees of freedom that this system has (e.g., what movements are possible) (Rodrigo & Ambrósio, 2011) (**Figure 10**).



Figure 10. A simplified biomechanical model of a body. a) The red graph (nodes and edges) is a biomechanical model of a biological system capturing the position of the joints; forces acting on the body have not been shown (Knudson, 2021). b) Orange ellipses roughly denote visual space around the body computed in the brain in a specific way known as peripersonal space (Serino, 2019), which is extendable with tools computed in primates' intraparietal sulcus (Maravita & Iriki, 2004).

Biomechanics is applied in studies of motor control (Latash & Zatsiorsky, 2016; Winter, 2009), in particular in the field of motor development (J. E. Clark & Whitall, 2021; Jensen, 2005; Jensen & Korff, 2005) and motor learning (Di Domenico, 2020; Keogh & Hume, 2012; Raiola, 2014). Biomechanical data are collected with high-speed sensitive cameras capturing the position of the joints and degrees of freedom of the biomechanical system of interest. Some studies apply biomechanics to analyzing sensorimotor systems of the brain (Lin et al., 2019; Schouenborg, 2004). However, most studies do not recognize biomechanical methodology in research on acquiring sensorimotor skills (Christensen et al., 2019; Kee, 2019; Nackaerts et al., 2019; Popp et al., 2018; Yin & Knowlton, 2006; Yokoi et al., 2018) or procedural learning (Henke, 2010; Mason & Just, 2020; J. N. Williams, 2020)¹.

It is claimed here that the biomechanical methodology is well suited to operationalizing skills researched in psychology, where skills are divided into "sensorimotor" and "cognitive" skills (Ackerman & Cianciolo, 2000; Cohen & Squire, 1980; Graybiel & Grafton, 2015; Hikosaka et al., 2002; Raiola, 2014). The field uses confusing terminology, as the acquisition of a sensorimotor procedure is described in three phases: cognitive, associative, and autonomous (Fitts & Posner, 1967; Kee, 2019). The cognitive phase of learning a sensorimotor skill often goes beyond procedural memory and engages episodic and working memory (Ackerman & Cianciolo, 2000). Here, skills refer only to the final stage, where a sensorimotor procedure is fully automated.

Given the operationalization of the biological body as a system processing sensory, somatosensory, and vestibular channels of information, skill learning is understood as encoding information flow within these channels into long-term memory (see Section 4.2). Skill acquisition investigated in biomechanics acknowledges a spectrum between "open skills," executed in unpredictable situations (like diving to block a penalty, maneuvering a ball in a team sport, or surfing), and "closed skills" performed in relatively stable situations (such as a gymnastics routine, running, or serving a ball) (Gu et al., 2019). This division reflects the cognitive and associative phases of skill acquisition that are subsumed with the autonomous phase when the skill is finally acquired (Ackerman & Cianciolo, 2000; Kee, 2019). In other words, with training, circumstances become encoded, and a skill gradually closes. Closing a skill means performing it without encountering surprises, i.e., a transition is accurately retrieved. Thus, here, only the "closed skills" meet the definition for skills, as "open skills" encounter surprisal and are not yet automated, thus relying on the memory system

¹ It is hypothesised that applying the biomechanical methodology to psychological and neuroscientific research on skills is essential in understanding skills acquisition and maintenance. Further, the biomechanical model capturing agents' actions and forces influencing these provide mathematical operationalization, which can be compared with underlying brain activity in the search for regularities. Such an approach seems to be particularly relevant for monitoring movement and brain abnormalities in the early stages of neurodegenerative diseases such as Parkinson's Disease.

beyond the one controlling the flow of sensory, somatosensory, and vestibular information.

The division for perceptual-motor and cognitive skills rely on the preservation (or impairment) of the ability to learn and perform skill by patients with degeneration involving the hippocampus (Gabrieli, 1998). For example, an amnestic patient known as H.M. encodes perceptual-motor tasks such as the pursuit rotor task or mirror tracing task, but his capacity to solve planning and problem-solving (e.g., tower) tasks was severely impaired (Saint-Cyr et al., 1988; Y. Xu & Corkin, 2001). Nonetheless, in certain circumstances, amnesic patients can perform well in tasks from the cognitive category (Knowlton et al., 1992). Still, relative to the control group, they show deficits (Butters et al., 1985), presumably showing an alternative strategy for skill acquisition (Ullman, 2004). The above studies indicate that performing a skill understood as changes in the flow of sensory, somatosensory, and vestibular information can result from heterogeneous information processing. This work distinguishes a memory system coupled with the ongoing flow of information and one dissociated from this flow (see Sections 3.2, 5.1).

Computationally, skill learning relies on adjusting the information flow from exploration to exploitation. This dilemma is central to any self-adaptive system and has no trivial solution (Berger-Tal et al., 2014; Wilson et al., 2021). Exploitatory actions generate anticipated feedback; exploration is a heterogeneous term roughly defined in the field into broad classes of "random" and "directed" exploration (Gershman, 2019; Gigerenzer & Gaissmaier, 2011; Wilson et al., 2021; Wilson, Geana, et al., 2014). The volatility of the environment (Findling et al., 2019), time horizon (Berger-Tal et al., 2014), and age (Dubois et al., 2022) influence the balance between exploration and exploitation. Directed exploration is thus akin to an educated guess reflecting the hypothesis of a specific transition in time. An animal learns the information flow in the sensory, somatosensory, and vestibular channels to the extent that the flow is sufficiently anticipated and hence exploited as a skill.

According to optimal control theory, it is understood that skills are reliably performed despite the variability in the information flow (Todorov, 2004). This means recalling the skills is never *precisely* the same, as the muscle movements express a degree of variability at low resolution. In the free energy literature, this variability of information is captured within the redundant subspaces (A. Clark, 2015, p. 119; Todorov & Jordan, 2002), a notion developed from "the minimal intervention principle" proposed by

Todorov and Jordan (2002). It allows for variations at lower layers of information processing relative to the layer at which the outcome is anticipated, encapsulating the range of expected variability of the transition. An animal acquires a skill when its execution repetitively does not generate surprise. In other words, feedback fits into redundant subspace, and transition in time is successfully retrieved from the memory.

1.4 Summary

The biological body in the physical space is described as a system processing sensory, somatosensory, and vestibular information. Associative brain areas process these inputs, including parietal and frontal cortices and the pericentral network. These brain regions underpin computations of the peri-spaces around specific body parts processed as body extensions into the physical space. Tool use further extends the visual receptive fields of the peri-spaces. This converging evidence justifies the operationalization of biological bodies as processing sensory, somatosensory, and vestibular information.

The biological body observed from the outside is investigated in the branch of physics called biomechanics. In biomechanical analysis, bodily systems of interest are captured with fast-motion-sensitive cameras and analyzed with sophisticated software. This approach allows for the rigorous biomechanical investigation of body movements that can be fruitfully applied to psychological research.

2. Consciously experienced surroundings of multiple potential futures.

The operationalization of animals as information processing systems largely ignores the animal's phenomenological experience. The biomechanical analysis and the focus on sensory, somatosensory, and vestibular information processing in the brain emphasize the observed movements, such as skills. However, these observations are often tied to a specific phenomenological experience. Animals do not experience the flow of information processing per se; animals experience surroundings that are progressively more complex along the phylogenetic and ontogenetic development. The richness of phenomenological experience includes a broad spectrum of qualia, such as redness, emotions, or meaning. This work is concerned with the action-oriented qualia that are called affordances.

The hierarchical organization of affordances toward the potential futures originates from the principle of future orientation in information processing, which also applies to the information processing that underpins the experience of qualia. Originally, affordances were meant to capture relations between the configuration of joints, height, strength, or mass of an animal and the physical properties of its surroundings (Chemero, 2009; Gibson, 2014; Richardson et al., 2008). For example, a leaf affords support for an ant, a rock affords sitting for a human of a given height, and a river affords locomotion for a fish. However, when the term gained popularity, it was adopted to multiple novel domains and has obtained alternative meanings (Bruineberg & Rietveld, 2014; Cruz et al., 2018; Djebbara et al., 2020; Eismann et al., 2021; Pezzulo & Cisek, 2016; van Dijk, 2021). Contemporarily, affordances are understood as relations between an organism's capacities to act and the features of its surroundings that change in time (Chemero, 2001, 2003, 2009).

In general, affordances denote the pathways toward potential futures that can be reached with specific information flows in the sensory, somatosensory, and vestibular channels. The evolution put pressure on optimizing the accuracy of the anticipated futures, thus driving the need for the models of potential futures. In humans, the phenomenological experience of action-oriented qualia encloses the rich landscape of affordances named surroundings.

2.1 Affordances toward the potential futures

In biological and psychological sciences, the concept of an organism's living space varies from that in physics. Ecological psychology refers to it as an environment (Gibson, 2014), while evolutionary biology calls it a niche (Lewontin, 2000). Both disciplines agree that animals live in subjective surroundings. In ecological psychology, the environment denotes direct perceptual information that an animal can act upon; in evolutionary biology, the niche encapsulates how an organism lives. Ecological psychology understands animals as following the possibilities for action known as affordances (Gibson, 2014; Richardson et al., 2008). Affordances have been originally operationalized in ecological psychology as ever-changing features determined by the unique temporal-spatial location of the perceiver to an object (Gibson, 2014). Essentially, an organism is inseparable from its environment-niche, i.e., environment and niche always surround an organism and provide it with affordances.

The central figure in ecological psychology was J.J. Gibson, who introduced affordances, denoting the direct experience of opportunities for actions (Gibson, 1950, 2014). Gibson claimed that an organism's environment differs from the objective physical world. In his own words:

If a terrestrial surface is nearly horizontal (instead of slanted), nearly flat (instead of convex or concave), and sufficiently extended (relative to the size of the animal) and if its substance is rigid (relative to the weight of the animal), then the surface affords support. (...) Note that the four properties listed—horizontal, flat, extended, and rigid— would be physical properties of a surface if they were measured with the scales and standard units used in physics. As an affordance of support for a species of animal, however, they have to be measured relative to the animal. They are unique for that animal. They are not just abstract physical properties. They have unity relative to the posture and behavior of the animal being considered. So an affordance cannot be measured as we measure in physics. (Gibson, 2014, pp. 119–120).

Gibson's theory of affordances is grounded in the idea that the environment offers animals opportunities for actions unique to that specific animal. These opportunities are based on the animal's physiological characteristics, such as joint configuration and mass. This intuition is grounded in the living space (umwelt), postulated by von Uexküll (Fultot & Turvey, 2019). In general, ecological psychology endorses the subjectivity of the relationships between biology and physics.

Chemero (2001) elaborated on the understanding of affordances, further supporting the notion that they should be viewed in a relative context. This was in response to frameworks that treated affordances as static properties of the world (Heft, 2001; Michaels, 2000; Reed, 1996). Chemero proposed the Affordances 2.0 theory, where the distribution between the organism's abilities to perceive and act and the features of its environment is understood as dynamic relations:

To formulate Affordances 2.0, start with Affordances 1.1, according to which affordances are relations between abilities to perceive and act and features of the environment. Then consider the interaction over time between an animal's sensorimotor abilities, that is, its embodied capacities for perception and action, and its niche, that is, the set of affordances available to it. (Chemero, 2001, p. 111)

Affordances 2.0 point to reciprocity in the system and emphasize that distributed information processing occurs in a temporal context. Following affordances involves transitioning toward specific potential futures that make subsequent affordances available. The understanding of affordances as dynamic, reciprocal relations within the distributed information-processing system is adopted here. As evolutionary biologists (Longo et al., 2012; Longo & Montévil, 2014) and philosophers (Bertolotti & Magnani, 2017) noted, the concept of "affordances" has become popular in cognitive sciences. They argue that affordances no longer refer solely to physical properties and suggest distinguishing between "biological enablements" and "psycho-cognitive affordances." For example, Bertolotti and Magnani refer to the New Caledonian crow, which can learn toolmaking through mechanical manipulation or social observation (Weir & Kacelnik, 2006). These studies emphasize the need for a more precise understanding of affordances because it became an umbrella term denoting all possibilities for action irrespective of the distribution type being acted upon. Nevertheless, most scholars use the notion of affordances broadly to indicate the rich landscape of invitations for action, and such convention is maintained in this work.

Taken together, affordances are understood as opportunities toward potential futures that a specific animal can reach. Affordances are action-oriented qualia; thus, they are subjective as the entirety of the phenomenological experience. The closeness to sensory, somatosensory, and vestibular inputs that can be experienced fully depends on the animal's phylogeny and ontogeny. Affordances thus reflect both (I) information processing performed close to the inputs and (II) advanced information processing decoupled from the inputs. For example, one can experience and act toward a future moment in dynamic team sports such as football or a future decoupled from inputs such as Moon colony. Both kinds of experience involve affordances with varying correlations with the ongoing flow of incoming information.

2.2 Surroundings denote environment and niche combined

Every animal is an information processing system with its own unique experience. This unique perspective on the world includes physical objects and social agents, as well as personal past and predictions for the future. This is especially true for complex information processing systems such as humans, as there is no clear boundary between the interior and exterior of the human. Clark and Chalmers (1998) realized that the animal extends with artifacts to augment information processing, pointing out that the brain is an interconnected network. Countless publications stress that animals are embodied, extended with artifacts and tools, enactively acting/interacting, and embedded with contexts such as one's history and plans for the future. Some accounts suggest that the brain is, in fact, 6e – also emotional and exapted (the latter being neuronal reuse/recycle). Similar ideas are also expressed in ecological-enactive writings where an animal is understood as a brain-body-environment dynamic system (Beer, 2008). Here, an animal is understood as an information processing system distributed onto objects, agents, and time. This distribution of the information processing system is consciously experienced as the landscape of hierarchically organized affordances that are called here the surroundings.

Affordances are action-oriented qualia that point to potential futures. The landscape of probable futures needs to be hierarchical, as imposing the hierarchical structure increases the accuracy of the predictions. The set of affordances accessible to an organism is called either an environment (Gibson, 2014) or a "niche" (Chemero, 2009). The understanding of the environment and the niche in the relevant literature overlap. Given that, the environment and the niche are combined into an umbrella term of surroundings that encloses the entirety of the future-oriented qualia experienced by a specific animal.

The niche is a term used in evolutionary biology that refers to *how* animal species live (Chase & Leibold, 2003; Odling-Smee et al., 2003; Pocheville, 2015). Ideally, a species would thrive in optimal conditions, but adaptation to sub-optimal circumstances is driven by evolutionary pressure and random factors. Hutchinson (1957) coined the term "realized niche" to describe this sub-optimal, inhabited niche. This understanding of the niche considers factors such as competition or scarcity of resources; thus, it is subjective to the specific species in specific conditions (Odling-Smee et al., 2013).

The term niche is widely used by cognitive scientists who applied it to "psychocognitive" circumstances (Bruineberg, Kiverstein, et al., 2018; A. Clark, 2006; Friston, 2011; Tooby & DeVore, 1987; Werner, 2021). Recall that the same observation has been made about the "biological enablements" and "psycho-cognitive affordances" (Bertolotti & Magnani, 2017; Longo et al., 2012; Longo & Montévil, 2014) (see Section 2.1). The mentioned authors suggested that enablements describe sensory, somatosensory, and vestibular information flows, whereas affordances denote the experience of social information processing. Nonetheless, when scholars across disciplines refer to biological and cognitive niches, they uniformly refer to "affordances" connecting the organism with its niche (Bruineberg, Kiverstein, et al., 2018; Iriki & Taoka, 2012; Pinker, 2013). If one accepts the philosophical argument, the biological niche is filled with enablements, and the cognitive niche is filled with affordances. However, here, the niche is operationalized in terms of the familiar affordances regardless of their distance from the inputs. For example, one can reliably imagine a memory palace dissociated from the incoming information (Mecacci, 2013), experienced without prediction errors, and thus, it constitutes a niche.

Since the niche denotes how an animal lives, it encapsulates an experience of affordances toward the potential futures that return the anticipated feedback. Following affordances involve changing the flows in sensory, somatosensory, and vestibular information that may predict the future accurately or may encounter prediction errors. Some affordances are "familiar" and can be reliably achieved, whereas others are "novel" and reaching them requires training. This distribution is the tradeoff between exploring the unknown and exploiting known affordances (see: Section 1.3).

In evolutionary biology, the term "niche construction" captures the mechanism of initial exploration and subsequent exploitation of the information in the surroundings. Animals create niches by altering their environment (perturbation) or changing the environment altogether (relocation). Niche construction is a process that works alongside natural selection, according to Odling-Smee et al. (2013). It involves animals expressing their genes not just through their physical characteristics but also through their actions, such as building beaver dams (Dawkins, 1982). By constructing their environment in a way that benefits their survival and the survival of their offspring, animals gradually develop their niches (Heras-Escribano & De Pinedo-García, 2018; Laland et al., 2015; Odling-Smee et al., 2013). Thus, niche construction is understood here as the initial

exploration of available affordances that, after repetitively acted upon, become the exploitative way of living for a given animal.

The process of niche construction captures a mechanism of initial encoding and subsequent retrieval of information stored in the agent's memory, which also applies to the organism's ontogeny. For example, primates from different tribes construct nests in specific ways (Boesch & Boesch, 1990; Hicks et al., 2019). In the case of humans, a variety of factors (including language, cultural, and social) play a significant role in shaping one's actions and worldview (A. Clark, 2006; A. Clark & Chalmers, 1998; Norman, 2013; Saarinen & Krueger, 2022; Taylor et al., 2011; Varela et al., 2017). In extreme circumstances, feral children of *H. sapiens* do not learn the niches that involve social interactions, and individuals with Savant's Syndrome learn extraordinarily detailed sensory niches. For *H. Sapiens*, ontogenetic development is a life-long process of developing knowledge structures (Piaget, 1970).

Complementary, scholars researching ecological-enactive information processing pointed out the distinction between available and relevant affordances (Bruineberg et al., 2019; Bruineberg, Kiverstein, et al., 2018; Rietveld et al., 2018). The field of "relevant affordances" has been conceptualized as "the field of solicitations" (Bruineberg et al., 2019; Rietveld et al., 2018). Solicitations refer to possibilities for action that stand out as essential to maintaining allostasis. The niche is a frequent experience of specific affordances that result in a particular way of living. Thus, these affordances are relevant for maintaining the distributed information processing system. When the familiar affordances are no longer experienced, the distribution of the system gradually shifts.

Similarly, in design literature, it has been noticed that a smartphone screen "invites" actions on its whole surface. However, only specific locations signify relevant activities (Norman, 2013). These observations point out that the affordances are experienced with varying frequency. Niche denotes exploitative affordances that are frequently experienced; environment denotes explorative affordances that are encountered less often.

The idea that the landscape of potential futures involves futures that have been visited in the past and the novel futures that are only hypothesized exemplifies a fundamental computational dilemma that any adaptive system faces. Namely, future surroundings are volatile, uncertain, complex, and ambiguous; thus, navigating them is

tied to uncertainty. An animal is unsure about future feedback; therefore, it faces the dilemma of exploring the unknown environment or exploiting the familiar niche.

2.3 Exploration-exploitation spectrum of affordances

Balancing between exploring new possibilities and exploiting the known ones is investigated in computational neuroscience (Dolan & Dayan, 2013; Sutton & Barto, 2018), and it has no trivial solution. The affordances are differentiated by appealing to the exploration-exploitation spectrum, i.e., to their familiarity. Exploration means trying something new with unknown consequences. If it works out well, it can be repeated in anticipation of obtaining the same results, at which point the repeated action becomes exploitative. The niche comprises the exploitable affordances defining an animal's way of life, whereas the environment encompasses explorable affordances.

The spectrum is investigated in experimental research on humans, where many factors influence whether one explores or exploits one's surroundings, including the available time horizon (Berger-Tal et al., 2014), age (Dubois et al., 2022), or volatility of the environment (Findling et al., 2019). Exploration can be done randomly or with a specific goal in mind (Wilson, Geana, et al., 2014), and it can involve using heuristics to guide the process (Dubois et al., 2021, 2022; Gigerenzer & Gaissmaier, 2011)². For example, experimental data show that saccades are directed to the most informative places and that their nature shift from reactive to proactive along a transition from uncertain to predictable circumstances (Donnarumma et al., 2017). The familiarity of the incoming information determines whether the incoming flow of sensory, somatosensory, and vestibular information is surprising.

An environment and a niche are *always* understood as relative to the ontogeny and phylogeny of an animal experiencing it. The environment includes all experienced affordances; the niche includes known affordances. The spectrum between explorative and exploitative affordances underpins the range between the environment and niche. An animal is thus understood as experiencing explorative affordances (comprising the environment) and exploitative affordances (constituting the niche).

The shift from exploration to exploitation is reflected in research on habit formation. In psychology, habits refer to automatic behaviors triggered by cues associated with past performances (Neal et al., 2006). Habits are learned tendencies to repeat previous actions in a given circumstance (Wood & Neal, 2007). They are often considered a part of a routine, such as tying shoes after putting them on (Graybiel & Grafton, 2015). Habits can be broken down into smaller chunks, but these chunks tend

² Directed and random exploration can be referred to as convergent and divergent creative thinking.

to group (Dezfouli et al., 2014; Dezfouli & Balleine, 2012). Habits are typically initiated by a cue eliciting the remaining steps without engaging in decision-making. This implies that habits are conceptually equal to exploitative affordances underpinning an animal's niche. Evolutionary speaking, an animal repetitively following exploitative affordances in its niche is, from a psychological perspective, engaging in habitual behaviors.

Actions are investigated not only as "habits", but also as "skills" (Ackerman & Cianciolo, 2000; Cohen & Squire, 1980; Graybiel & Grafton, 2015; Hikosaka et al., 2002; Raiola, 2014). Biomechanically, skills are enclosed in the spectrum between open and closed skills (see: Section 1.3). Psychological literature differentiates three phases of skill learning: cognitive, associative, and autonomous (Fitts & Posner, 1967; Kee, 2019). The cognitive phase involves episodic and working memory (Ackerman & Cianciolo, 2000). Note that the initial learning of a skill and its subsequent automatization reflects a gradual progression from exploration toward exploitation of affordances.

Skill and habit learning is understood here as involving two stages: (I) learning how to move the body and (II) learning the feedback to these movements. More precisely, (I) the initial stages of learning are understood as learning the information flow to balance the body's movements and find the *sufficient* transition toward a specific potential future. Once adequate move is discovered, (II) the subsequent stages of learning shift the information processing toward focusing on increasing the *accuracy* of the transition in time. Once a sufficient conversion in time is identified, its execution leads to an increase in accuracy, which reflects the development of the redundant subspace described in optimal control literature (Todorov, 2004; Todorov & Jordan, 2002). A sensorimotor procedure gradually becomes a learned exploitative affordance underpinning an animal's niche.

The process of learning action is investigated in neuroscience. Researchers have studied habits and instrumental activities under various names for nearly a century. Tolman's (1948) cognitive maps and Thorndike's (1898) stimulus-response account were initially compared. Later generations of research focused on goal-oriented and habitual actions in rodents (Balleine, 2005; Balleine & Dickinson, 1998; Killcross, 2003) and humans (Tricomi et al., 2009; Valentin et al., 2007).

The contrast between habits and goal-oriented actions can be misleading, implying a criterion related to a specific outcome. The neuroscientific definition of goaloriented actions is that a goal-oriented action (I) reflects knowledge of the relationship between an action (or sequence of actions) and its consequences, which is known as
response outcome control, and that (II) the outcome should be motivationally relevant or desirable at the moment of choice. "Crudely, subjects choose actions because they think that those actions lead to outcomes that they presently desire." (Dolan & Dayan, 2013, p. 314). In contrast, habitual actions are based on repeating past actions that have led to positive outcomes but may not necessarily be linked to the desired result at the present moment. The above criteria were initially developed in animal research and directly adapted to human research (Dickinson, 1994) without recognizing the complexity of the human phenomenological experience (Bratman, 1991; Mele, 1992; Pacherie, 2008; Searle, 1983). This complexity of intentional states in *H.Sapiens* makes direct translation of desirable conditions from the rodent literature inappropriate. Humans express contradictory desires and complex sets of intentions. Further, humans (and other cognitively advanced organisms) can postpone the rewards, desire things they do not intend, and intend to do something they do not want. Thus, the difference between goaloriented and habitual instrumental actions concerns the automation along the exploration-exploitation spectrum, not the post hoc judgment of the feedback from following the affordance.

For example, a neuroscientific test for habitual behavior relies on the initial automatization of action in a specific environment and then changing the structure of this environment. Habit is diagnosed when an animal performs a trained action despite the satiation or absence of the reward (Dolan & Dayan, 2013). Initially, the behavior is goaloriented but can become habitual in the right circumstances (Poldrack & Packard, 2003). The degree of habitual behavior is measured by observing the frequency of "slips of action" - irrelevant actions given the current circumstances but automatically triggered by environmental cues (De Wit et al., 2012). The slips of action manifest the previously acquired niche that has not been updated yet.

The above protocols "diagnose" habits when an affordance no longer meets its trained surroundings. If not for the change in the surroundings, the circumstances would guide affordance toward the known future and would be understood as exploitative affordance in the niche. For example, one might say that he automated coffee affordance in a specific office location. Still, when the coffee machine changes place – one would be called having a *habit* of getting the coffee at this particular place as the affordance is repeated despite the broken relationship between an action and its consequence. Action considered goal-oriented in one circumstance suddenly becomes interpreted as a habit due to a change in the surroundings, not a change in the information processing system.

The system acts on its model, and at the moment of initiation of the affordance, the system anticipates the usual reliability of the model. In this light, the critical characteristic of the habitual action is not teleological, but that such action is automatically *elicited* by a cue based on a cached value. In other words, action is habitual when exploitative, i.e., the experience of specific affordance guides its initiation.

Subsequently, studies on instrumental learning moved toward computational neuroscience. They were realized as flexible Model-Based (MB) and habitual Model-Free (MF) controllers (Daw et al., 2005; Doya, 1999; Doya et al., 2002; Smittenaar et al., 2013; Sutton & Barto, 2018). Further research focused on investigating cooperation and competition between the controllers (Daw et al., 2011; Doll et al., 2009; Fermin et al., 2010; Gershman et al., 2012; Gläscher et al., 2010; Otto et al., 2013; Simon & Daw, 2011; Wunderlich, Dayan, et al., 2012; Wunderlich, Smittenaar, et al., 2012, for a review see (Dolan & Dayan, 2013). Despite various operationalizations, this line of research recognizes a division between two ends of a spectrum of decision-making: the MB controller, which is computationally efficient, automatized, and inflexible. Actions result from balancing these controllers' cooperating and competing simultaneous activity. Here, the MF habitual controller encapsulates minimal free energy as the action consequences are cached at the initiation (see Section 4.4).

The reviewed studies indicate that acquiring skills and habits involves explorative following of the affordances toward the possible futures. Frequently followed affordances return familiar feedback and constitute an organism's niche. The labels of "habits" and "goal-oriented" actions are tremendously misleading as the criteria are not teleological but relate to the automaticity of eliciting the behavior. The acquisition of a skill is reflected as a transition of information processing between the brain's networks and demonstrates a shift from the explorable environment toward the exploitable niche. In general, it is noticed that the common focus of research on actions, habits, skills, and niche construction is that behavioral procedures become increasingly automated along subsequent repetitions, and such automation is reflected in accurate anticipations of the future. The landscape of potential futures encloses the experience of affordances that lead to the exploration of novel futures and these leading to the exploitation of the known futures. The entire experience of hierarchically organized affordances toward the potential futures is called here the surroundings.

2.4 Decision-making and action execution in the striatum

The research on instrumental conditioning dates back to research on cognitive maps by Tolman (1948) and stimulus-response accounts by Thorndike (1898) (see Section 2.3). Subsequently, this research line differentiated between goal-oriented and habitual actions. Goal-oriented actions "(I)reflect the relationship between the action and its effect (response-outcome control), and (II) the outcome should be desirable at the moment of choice" (Dolan and Dayan, 2013, p. 314). On the other hand, habitual actions rely on repeating the steps that have led to desirable outcomes but are currently divorced from the associated consequences (ibid.). However, the terminology implies teleological criteria for action, which is misleading. During the progress in the field, the language for researching goal-oriented and habitual actions has been directly adopted from the animal literature (Dolan and Dayan, 2013) without recognizing the complexity of human motivations, such as contradictory desires and complex intentions. Human agents build coherent preferences toward future actions (Hartnett et al., 2016; Pacherie, 2008). Action plans often span long into the future, revealing a mereological hierarchy in which an event consists of subevents with corresponding subgoals (Zacks et al., 2001, 2007; Zacks & Tversky, 2001). This suggests that actions are heterogeneous. They are hierarchically organized and differ in length and complexity.

Neuroscientific studies on information processing behind exploring and exploiting affordances investigate the process of action acquisition and execution in the basal ganglia (BG) (Poldrack & Packard, 2003). The BG is a set of deeply located brain areas whose role is to gate the activity of the whole neocortex (Rolls, 2016). The central part of the BG is the striatum (STR), a structure between the thalamus and cingulate cortex (Figure 11a). The STR constitutes one of the most critical animal networks that orchestrate the activity of the whole neocortex by inhibiting the thalamus (McCutcheon et al., 2021). Further, the STR is closely connected to the reward-motivated motivational circuits of the brain (Loonen & Ivanova, 2015), as well as the rhythmic clock of the brain (Breska & Ivry, 2018). The BG is involved not only in mediating movement per se, given the findings related to Parkinson's Disease, but also in influencing information processing beyond locomotion, such as flexible navigation or emotional processes.



Figure 11. Simplified neuroanatomical organization of the brain. **A)** Sagittal section of the brain. The neocortex sends an excitatory signal to the respective parts of the Basal Ganglia (BG), which sends an inhibitory signal to the Thalamus (TH). As an effect, TH cannot inhibit the respective parts of the neocortex. The circuitry is modulated by the Dopamine (DA) from midbrain nuclei. Cerebellum participates in action control, albeit its role is beyond this work. **B)** Coronal section of the brain with components of the BG: Caudate Nucleus, Globus Pallidus [internal and external], Putamen, Subthalamic Nucleus (approximate location), and Substantia Nigra pars reticulata (approximate location). Internal Capsule is a white matter tract dividing the Caudate Nucleus and Putamen. The Figure is based on: (Goodroe et al., 2018a; Graff-Radford et al., 2017).

The STR in primates can be anatomically divided into the caudate nucleus and the nucleus accumbens, separated from the putamen by a white matter tract known as the internal capsule (Figure 11b). In rodents, the internal capsule does not separate the complex; thus, the STR is divided into dorsomedial, ventral, and dorsolateral. Studies show a gradient of information processing through the STR complex: from ventromedial parts mediating flexible behavior to lateral parts processing habitual behavior (DeCoteau et al., 2007; Devan & White, 1999; Ragozzino, 2002; Thorn et al., 2010)³.

The information processing in the STR is significantly influenced by dopaminergic modulation from the Substantia Nigra (SN) and Ventral Tegmental Area (VTA) (**Figure 12**) (see Section 6.2). Initial research on the dopaminergic nuclei was heavily influenced by movement disorders observed in Parkinson's Disease patients, whose dopaminergic neurons in lateral substantia nigra progressively degenerate (Duke et al., 2007). Medial and ventral portions of the substantia nigra project to different parts of the basal ganglia (McCutcheon et al., 2021; Y. Zhang et al., 2017) and are less involved in the disease progression. Just above the substantia nigra a second dopaminergic nucleus

³ The same gradient is present in the midbrain dopaminergic nuclei, from ventrolateral Substantia Nigra (SN) through dorsomedial SN to Ventral Tegmental Area (Bromberg-Martin et al., 2010; A. P. Chen et al., 2021; McCutcheon et al., 2019, 2021).

is located – the ventral tegmental area, which projects to its respective parts of the STR and neocortex (Bromberg-Martin et al., 2010). The SN, VTA, and STR are vital information processing components behind the following affordances.



Figure 12. Selected subcortical structures involved in producing action acquisition and execution. The substantia nigra (SN) and neighboring ventral tegmental area (VTA) (not shown in the figure) synthesize dopamine for the whole brain (Bromberg-Martin et al., 2010), including the basal ganglia (BG) presented in the picture. Figure attributed to Blausen.com (2014).

The alternative division of the STR is functional, enclosing ventrally located limbic striatum (LimSTR), anteriorly located associative/relational striatum (RelSTR), and situated posteriorly sensorimotor striatum (SenSTR) (A. C. W. Smith et al., 2021) (**Figure 13**). This functional operationalization is favored, given reports of involvement of the SenSTR in encodings (via D1-receptors in the Medium Spiny Neurons) and executing (via D2-receptors in the Medium Spiny Neurons) habitual behaviors (A. C. W. Smith et al., 2021). Since these observations have been performed only recently, the STR complex in earlier literature is divided anatomically rather than functionally.



Figure 13. Human striatum and associated areas. A schematic figure of the pathways of information processing in the STR of primates. The complex can be divided neuroanatomically in primates into the caudate nucleus, the putamen, and the nucleus accumbens. The complex is divided functionally in rodents into dorsolateral, dorsomedial, and ventral striatum. The division of the striatum adopted here is functional and conceptualized as Limbic (LimSTR), Relational (ReISTR), and Sensorimotor (SenSTR). The sections of the STR receive projections from various parts of the neocortex (indicated by color), are bidirectionally connected with dopaminergic nuclei in the midbrain, and send axonal projections to the respective sections of the thalamus (not shown in the figure). VTA = Ventral Tegmental Area, dmSN = dorsomedial Substantia Nigra, vlSN = ventrolateral Substantia Nigra, NAcc = Nucleus Accumbens, OFC = Orbitofrontal Cortex, HPC = Hippocampus, vmPFC = ventromedial Prefrontal Cortex, ACC = Anterior Cingulate Cortex, dlPFC = dorsolateral Prefrontal Cortex, PMC = Premotor Cortex, PC = Parietal Cortex, MC = Motor Cortex, SSC = Somatosensory Cortex. Based on: (Ferré et al., 2018; Goodroe et al., 2018; Graff-Radford et al., 2017; Hooks et al., 2018; Lanciego et al., 2012; McCutcheon et al., 2019; Sherman, 2017; Xia et al., 2011).

Existing studies on rodents indicate that the RelSTR supports information processing associated with flexible behavior, whereas the SenSTR mediates information processing linked to inflexible behaviors (Balleine, 2005; Yin et al., 2005). Lesions to the RelSTR promote habit formation (understood as the action inflexibility, "response strategy"). In contrast, lesions to the SenSTR result in the maintenance of goal-directed (flexible, "place strategy") behavior even with extensive training (Yin & Knowlton, 2006). Studies on mice show that associative loops through RelSTR modulate access to sensorimotor loops through SenSTR (Thorn et al., 2010). Recent studies in mice

correlated the activity of a cluster of neurons in the SN with initiating a movement sequence in a free exploration paradigm (da Silva et al., 2018). During skill acquisition, the computation in the RelSTR is attenuated, and the information processing in the SenSTR begins to bracket the entire sequence (K. S. Smith & Graybiel, 2013; Thorn et al., 2010); for a review, see (Graybiel & Grafton, 2015). A transition from the RelSTR toward the SenSTR has been observed during training skills in mice (Yin & Knowlton, 2006) and primates (Brotchie et al., 1991; Kermadi et al., 1993; Z. M. Williams & Eskandar, 2006). The RelSTR modulates the activity of the associative parts of the cortex, being a vital part of the associative learning processes.

The SenSTR is considered a hub of the "knowledge how" memory system and cooperates with the parietal cortex and the primary sensory cortices, in particular with motor and somatosensory cortices (Poldrack & Packard, 2003; Tricomi et al., 2009; Yin et al., 2005). The SenSTR is tightly connected with the motor cortex, which indicates building an enduring representation of the motor sequence (Coynel et al., 2010; Lehéricy et al., 2005). The acquisition and automation of skills are reflected in a transition of information processing between these structures (Poldrack & Packard, 2003; Williams & Eskandar, 2006; Yin & Knowlton, 2006). Initial attempts to perform a skill engage the spatial (associative) format of deciding what to do and how to do it, while after a sufficient amount of practice, the skill reflects efficiently obtaining an outcome with trained motor representation (Hikosaka et al., 2002). These studies are explored further in Chapter 6.

The activity of the STR slices a continuous experience into discrete chunks in a process named bracketing (K. S. Smith & Graybiel, 2013; Thorn et al., 2010). The process is well documented in rodents, yet as Graybiel and Grafton (2015, p. 5) notice: *"the function of bracketing remains unknown"*. Here, bracketing is interpreted as a neural manifestation of the length of a transition in time. It is impossible to pinpoint how long an action is, as actions chunk together (Balleine, 2005; Desrochers et al., 2016; Dezfouli et al., 2014). From a particular state (i.e., at home), a handful of affordances can be followed (i.e., walking a dog, going to work, cleaning). The remaining part is habitually elicited once an action is initiated (i.e., leashing a dog, grabbing a suitcase or a vacuum cleaner). Thus, after initiation, exploitative affordances usually require no decision-making as perceiving a cue automatically elicits the remaining part of the exploitative affordance without considering alternative decisions.

2.5 Summary

Animals are understood as agents acting upon their subjectively experienced surroundings. Executing actions implies following affordances, which results in an experience of subsequent expected and surprising affordances. The affordances are distributed on the exploration-exploitation continuum. Explorative affordances are investigated as goal-directed actions, open skills, initial stages of skill acquisition, and the MB controller; they are associated with information processing in the RelSTR. Exploitative affordances are investigated as habitual actions, closed skills, autonomous stages of skill acquisition, and the MF controller; they are associated with information processing in the SenSTR. Animals experience the hierarchical landscape of affordances toward potential futures that span the exploration-exploitation spectrum as exploitative niches and explorative environments that jointly compose surroundings. However, the evidence suggests that the information processing in the above structures is insufficient to compute the phenomenological experience.

3. Information processing behind the conscious experience

Animals phenomenologically experience possibilities for actions known as affordances. There is *something like* having a specific experience given one's phylogenetic and ontogenetic history (Nagel, 1974). In this chapter, the phenomenological experience is specified as a conscious, coherent, and effortless experience of the meaning of the situation associated with information processing in a set of brain areas collectively called the gestalt cortex. The experience of qualia encompasses a broad spectrum of possible construals, including matter, minds, meaning, and emotions, from which this work focuses on the action-oriented qualia known as affordances. Specifying the information processing behind the phenomenological experience allows for more precise operationalization of the subset of qualia related to actions.

3.1 Conscious, coherent, effortless experience of subjective construal

The phenomenological experience of qualia encompasses a rich spectrum of phenomena. Out of all possible experiences, this work is interested in action-oriented qualia (affordances) that correspond to changes in the sensory, somatosensory, and vestibular flows of information. The experience of the meaning of a situation is known in social psychology and linguistics as construal (Frazier & Clifton, 1996; Geeraerts, 2006). This immediate interpretation of information can be quickly updated, and then a new construal provides a way of understanding the world by a specific being (subject of conceptualization) (Verhagen, 2010). In a sense, construals are the consciously experienced surroundings explaining the incoming information and guiding future actions.

The conscious experience is carefully analyzed by Lieberman (2022), who operationalized **experience as a conscious, coherent, effortless experience of subjective construal**. Such a definition of the experience is what stream of consciousness is (James, 2007). The experience is conscious as it is experienced, which is used as a synonym for "consciousness" (Tononi & Koch, 2015). Coherence means that at a given time, a single construal is experienced. Effortlessness implies that the experience of construal is readily present to the experiencing animals, and no deliberate effort is required for experience.

By contrasting effortless experience with effortful thinking, Liberman (2022) draws a boundary that thoughts are separate from experience. This division corresponds to experiencing the information processing correlated with the inputs and experience of the information processing dissociated from the ongoing flow of information. This dissociation is recognized in the literature as the difference between "experiencing self" and "remembering self" (Kahneman & Riis, 2005; Zajchowski et al., 2017). The experiencing self is immersed in the online flow of sensory, somatosensory, and vestibular information; the remembering self exists in dissociation from these channels, commonly called the mind.

The sharp division between effortful conscious thought and effortless conscious experience presented by Lieberman (2022) suggests that these activities are mutually exclusive. However, thoughts, albeit effortful to generate, result in discovering new ideas that are themselves *experienced*. For example, watching a movie makes a person experience the plot, which can be effortfully reflected on by analyzing its flaws; in this

case, the content of experience is one's reflection on the plot, not the plot itself. Another example is imagination as a reflective (effortful) process capable of eliciting prereflective (effortless) experience of the imagined scenes, such as in the case of patient Shereshevsky, who was a master of a mnemonic technique called "memory palace" (Mecacci, 2013). This evidence points out that the "experiencing self" and "remembering self" are not mutually exclusive. Instead, these modes of information processing are intertwined as the content of experience may or may not relate to the incoming bodily information.

The experience content significantly differs between individuals and can originate from the processing of sensory, somatosensory, or vestibular information or the reflection dissociated from these inputs. Experience of imagination varies between individuals as certain people lack vividness in visual imagery (a condition called aphantasia) (Fulford et al., 2018), and people differ in their capacity to experience their episodic and autobiographical memory vividly (Boccia et al., 2019; Milton et al., 2021; Palombo et al., 2015, 2018; Watkins, 2018). This implies that the entirety of conscious experience is representational and varyingly correlates with the ongoing flow of incoming information (see Section 4.1).

3.2 Gestalt areas compute the experience

Neural correlates of experience are studied within theories investigating consciousness experimentally (Dehaene et al., 2014; Dehaene & Changeux, 2011; Koch et al., 2016; Siclari et al., 2017; Tononi et al., 2016), for a review see Seth and Bayne (2022). In his article, Lieberman (2022) addressed these theories and linked the experience of construal with the activity of a posteriorly located set of brain areas that he called the gestalt cortex (**Figure 14**). According to Lieberman, theories of consciousness include the gestalt cortex in their predictions about the neural correlates of consciousness. The name of this set of regions is a direct reference to gestalt psychology, which recognizes that the sum of the experience differs from its parts, similar to how the construal denotes a spatiotemporally coherent understanding of a specific situation.



Figure 14. Gestalt areas. Areas suggested by Lieberman (2022) processing information behind the conscious, coherent, and effortless experience of the subjective construal. The similarity of experience is reflected in the neural synchrony of the gestalt cortex between subjects (for a review, see Lieberman, 2022; Yeshurun et al., 2021). aIPS – anterior intraparietal sulcus, IPL – inferior parietal lobule, SMG – supramarginal gyrus, AG – angular gyrus, TPJ – temporoparietal junction, pSTS – posterior superior

temporal sulcus, V5 - motion areas for objects and bodies, LOTC – lateral occipitotemporal cortex, ITG – inferior temporal gyrus, VTC – ventral temporal cortex (on the bottom of the brain). (The brain figure in this and every subsequent token of this figure is used according to Creative Commons Attribution 2.5.)

The localization of the gestalt areas reflect the fact that the human brain predominantly processes visual information. Visual information processing starts at the level of receptors on the retina, reaches the primary visual cortex through the thalamus, and diverges into two interconnected streams of visual information processing known as ventral and dorsal (Milner, 2017) (**Figure 15**).



Figure 15. Ventral and dorsal streams. Schematic illustration of the dorsal "where" and ventral "what" streams (Milner, 2017). Two streams are partly independent brain networks processing visual information in the primate brain. The ventral stream passes through the ventral temporal cortex and reaches the hippocampus; the dorsal stream terminates in the lateral portion of the inferior parietal lobule and the dorsolateral prefrontal associative area (Guterstam et al., 2015, 2018) Information processing in the ventral stream is qualitatively different at the retina level, from which parvocellular pathways to the lateral geniculate nucleus in the thalamus give rise to the ventral stream of information processing (Kravitz et al., 2013). Complementary, the magnocellular pathway between the retina and thalamus processes information relevant to the dorsal stream.

Information computed by the streams is intertwined at every level of processing, from the thalamus to associative regions of the neocortex. Thus, information processing within the streams is separate but still interconnected. Crucially, the ventral stream sharply differentiates the processing of animate and inanimate stimuli (Connolly et al., 2012; Thorat et al., 2019).

The ventral stream comprises, among other structures, the Ventral Temporal Cortex (VTC), which consists predominantly of the Fusiform Gyrus (FG), which is included in the gestalt cortex proposed by Lieberman (2022). The VTC organizes information processing along axes pointing to the animate-inanimate distinction. The FG is divided by the mid-fusiform fissure into two parts: the lateral portion is most active when processing faces, small objects, and living things, while the medial portion is most active when processing places, large objects, and non-living things (Grill-Spector & Weiner, 2014). Posterior portions of the VTC categorize experience based on perceptual features, whereas the anterior part of the VTC categorizes experience along the agency dimension (Thorat et al., 2019). Information processing in the ventral stream is contemporarily modeled with convolutional neural networks (Lindsay, 2021). This points to two parallel organizations of the VTC as a potential direction for further inquiry.

The VTC computes the early stages of information processing within the primate's ventral stream. The VTC uses a nested spatial hierarchy to organize representations of visual information, allowing for flexible access to category information at various levels of abstraction (Grill-Spector & Weiner, 2014). This spatial hierarchy serves as a neural infrastructure for the representational scale in the VTC, resembling the spatial organization of conceptual spaces developed in linguistics (Gärdenfors, 2014). The categories are organized from the basic level (e.g., car vs. face) to the superordinate level (animate vs. inanimate) to the subordinate level (Tom vs. Bill). Subsequent research showed that image categorization on the superordinate level occurs concerning categorizability and agency (Thorat et al., 2019). The categorizability is based on the separation of visual details into different dimensions. For example, face recognition in the FG is computed on an n-dimensional matrix of individual facial features (Chang & Tsao, 2017) that can be extrapolated into representing different matrices, such as cars (Burns et al., 2019). The agency is associated with the capacity for self-propelled motion, i.e., performing intentional actions (Shatek et al., 2022; Thorat et al., 2019). The neurobiological evidence shows that the VTC processes information within the ventral stream by categorizing the objects as either animate or inanimate

concerning the agency. Further, a recent electrocorticography (EcoG) study on epileptic patients showed that the ongoing experience of objects is underpinned by sustained activity within the VTC (Vishne et al., 2023).

Studies on the ventral visual pathway suggest that information processing in the VTC and the ITG compute the topological arrangement of continuous information about objects' form (Ishai et al., 1999). Further, the activity of right-lateralized AG and the left-lateralized ITG have been associated with visual recognition of patterns (Herath et al., 2001). The LOTC is involved in the mental imagery of body parts (Kikuchi et al., 2017). Evidence supports the view that components of the ventral stream process information related to objects' identity based on their perceived and remembered categories.

In the dorsal stream, the V5 processes information related to biological motion, that is, its presence and direction in both front-parallel (parallel to the face) and in-depth movement (Zeki, 2015). The pSTS located next to the V5 processes information about the outcome of human and non-human intentional behavior (Shultz et al., 2011). Thus, this region combines expectations of the most likely course of action given the action's inferred meaning (Stehr et al., 2021).

The AG located dorsally to the V5 computes various information associated primarily with semantic processing (Binder et al., 2009; Bonner et al., 2013). Damage to the AG results in the angular syndrome characterized by alexia (incapacity to recognize words) and agraphia (incapacity to write words) (Nagaratnam et al., 2002). The AG is further functionally divided into components that show different activities depending on task demands (Seghier et al., 2010).

The AG located rostrally the SMG shows high lateralization of functions, resulting from language processing. In most people, language comprehension (Wernicke's area in the STG) and production (Brocka's area in the IFG) are left-lateralized. As a result, the left-lateralized SMG supports word processing (Oberhuber et al., 2016) and pitch memory (Schaal et al., 2017). In contrast, the right SMG is involved in the processing of rhythm in speech (Schaal et al., 2017), proprioceptive signals (Ben-Shabat et al., 2015), emotion recognition (Wada et al., 2021), and empathy (Silani et al., 2013).

Lateralization is also observed in the TPJ located ventrally to the SMG and AG. Left TPJ is involved in inferences in strategic decision-making, while the right TPJ in the detection of human agency (Ogawa & Kameda, 2019), embodied perspective-taking (Martin et al., 2020), moral conflict (Obeso et al., 2018), and emotional mimicry (Peng et al., 2021). Some studies do not report on the functional lateralization of the TPJ and link bilateral information processing in this area with the conscious experience of the self (Blanke et al., 2005) and self-other distinction (Quesque & Brass, 2019), which supports the view that this area is involved in broadly understood social cognition, perspective taking, and attentional processes (Krall et al., 2015; Kubit & Jack, 2013).

The IPL is involved in several functions, including spatial attention, multi-modal sensory integration, and oculomotor control (Clower et al., 2001). Its role is even broader as it includes processing information related to language and social processing (Numssen et al., 2021). Studies suggest the IPL is divided into three functionally separated segments (Caspers et al., 2013).

The IPS located dorsally to the IPL processes sensory and somatosensory information to control arm and eye movements in space (Grefkes & Fink, 2005). The IPS computes information underpinning working memory tasks contralaterally, similar to the organization of the early visual areas (Bray et al., 2015). Studies on humans show that the IPS plays a crucial role in response inhibition (Osada et al., 2019). This area is also heterogeneous and can be divided into subdivisions (Niu et al., 2020).

The gestalt areas suggested by Lieberman (2022) located in the parietal cortex (aIPS, IPL, SMG, AG, TPJ) mediate conceptual, social, and spatial information processing. Lieberman's (2022) statement is that these areas and temporal regions are included in processing information underpinning conscious experience. This observation aligns with evolutionary evidence pointing to the parietal cortex's recent phylogenetic development (Iriki and Taoka, 2012). The hypothesis is that the expenditure of the parietal cortex correlates with the increasing complexity of the niche composed of physical objects and social interactions. Gestalt areas included highly expanding areas during the onto- and phylogenesis of *H. sapiens* (Fjell et al., 2015), further supporting the hypothesis that these areas' development constitutes an individual's conscious experience.

The activity in the gestalt areas tends to synchronize between subjects who experience a given situation from the same perspective. For example, listening to a narrative evokes similar between-subjects activity in the gestalt areas, even when presented in different languages (Honey et al., 2012). In another study, decoding the high dimensional structure of a narrative with language processing software allows for training a classifier capable of decrypting from neural data which story a given

participant is reading (Dehghani et al., 2017)⁴. Other studies suggest that memories about specific events share neural representation (J. Chen et al., 2017; Zadbood et al., 2017). These studies indicate that between-participant synchrony in gestalt areas reflects adopting the corresponding understanding behind the conscious experience.

Lastly, conscious experience should not be confused with the neural correlates of memory, such as the areas located medially to the VTC: the Parahippocampal Gyrus, the Entorhinal Cortex (EC), and the Hippocampus (HPC). These regions do not compute the information underlying the experience, given that the amnesic patients still experience it. Instead, studies involving epileptic patients have revealed that specific cells within the HPC, known as "concept cells," are active when experiencing and imagining objects (Quiroga, 2012). These cells remain active during working memory tests, as shown by Kornblith et al. (2017). Crucially, the cells become sensitive to new items when presented with other things, as Reddy and Thorpe (2014) demonstrated. This evidence suggests that the information processing in these regions involves memory encoding related to the experience, not the experience itself. Further, this points to the observation that the conscious experience is distinctive to every single being as the content of experience differs between individuals.

⁴ Note that this article is co-authored by Ashish Vaswani, a main author of the "Attention is all you need" paper, which started the Transformer architecture and lead to development of the large language models such as ChatGPT (Vaswani et al., 2017).

3.3 Summary

The studies reviewed in this chapter pinpointed the neural computations behind the conscious experience of subjective construal. The experience is defined as conscious, effortless, coherent, and subjective. The content of experience is a pre-reflective construal that denotes understanding a given situation given the background knowledge. In the human brain, the experience is associated with information processing in the gestalt areas, which varyingly correlates with the ongoing input. The gestalt areas synchrony point to the possibility of identifying an inter-subject neural underpinnings of the conscious experience. However, these areas do not work in isolation and communicate with other regions in the large-scale brain networks (see Section 5.2). Thus, the experience computed in the gestalt areas can originate directly from the incoming information or an effortful thinking process. In either case, the experience itself is effortless. Before reviewing the brain network involved in computing the content of experience, the research on memory systems is discussed, as memory captures one's identity and thus influences one's conscious experience of the construal.

4. Two kinds of spaces for the future are stored in memory.

The principle of future-oriented information processing entails evolutionary pressure for developing memory to increase the accuracy of predictions, i.e., reducing future surprises. The principle is shared in many frameworks, most notably in the framework known as the free energy principle, which describes how organisms maintain their homeostasis by minimizing free energy through accurate predictions (Friston & Stephan, 2007). The free energy principle is considered a general imperative for self-organizing systems encapsulating what an organism should do to maintain its allostasis in the long run.

The free energy principle is unconnected to any biological/neural theory but is often linked to (I) Bayesian inference in the predictive coding theory (Friston, 2011; Knill & Pouget, 2004). Alternatively, the free energy principle relates to (II) ecologicalenactive accounts, extending the principle to the whole brain-body-environment system (Bruineberg, Kiverstein, et al., 2018; Bruineberg & Rietveld, 2014; S. Gallagher & Bower, 2014). The ecological-enactive model is interpreted as computations of the biomechanical space; the Bayesian hierarchical generative model is understood as calculations of the inferred space. It is claimed that every agent performs multiple spatiotemporally broad actions in numerous biomechanical and inferred spaces. Parts of these spaces resonating with the gestalt areas are experienced in a subjective, conscious, effortless, and coherent manner.

Nevertheless, the spaces interact independently from the gestalt areas. The training of the biomechanical and inferred spaces takes place in a broader context of the entire brain's activity. The spaces are depictions of the computations performed in the specific brain networks. The brain networks undergo lifelong development and sophisticatedly interact with each other on structural and functional levels (Beaty et al., 2015; M. Chen et al., 2022; Garin et al., 2022; Huijbers et al., 2011; Uddin et al., 2019). To explore the architecture of the spaces, studies on the neurobiological basis of memory are reviewed. Notably, the brain network enclosing the hippocampus (HPC) is in focus as this network is investigated with the most scrutiny.

4.1 Two interpretations of the free energy principle

A wealth of literature acknowledges the importance of future-oriented information processing; here, the review starts with the free energy principle framework. Accounts of the predictive brain (Bruineberg, Kiverstein, et al., 2018; Constant et al., 2021; Friston & Stephan, 2007) agree that perception is fundamentally hypothesis testing (Gregory, 1968, 1980), which itself is an exemplification of perception as "unconscious inference" (Von Helmholtz, 1886). The proponents of the free energy principle posit that the principle encapsulates the organization of all self-adaptive systems (Friston et al., 2023). However, since its inception, the free energy principle has been interpreted in multiple ways, grounding the principle in numerous frameworks (Bruineberg, Kiverstein, et al., 2018; Bruineberg & Rietveld, 2014; Corcoran et al., 2020; Friston, 2013; Hohwy, 2016; Kiefer, 2020; Kirchhoff & Kiverstein, 2019; Safron, 2020, 2021). Also, the principle itself has been refined since the initial introduction (Andrews, 2021; Biehl et al., 2021; Da Costa et al., 2021; Friston, Da Costa, et al., 2021; Friston et al., 2023).

Proponents of the ecological-enactive implementation of the free energy principle argue that the anticipating brain is not a hypothesis-testing scientist (Bruineberg et al., 2018). This argumentation tackles animals as constructing generative models of the hidden causes of their experience (Friston and Stephan, 2007). Here, it is recognized that whether a given experience is classified as hypothesis testing strictly depends on the place of this affordance on the exploration-exploitation spectrum (see Section 2.3). Exploitative (known) affordances reflect hypotheses tested multiple times in the phylogenetic and ontogenetic past, are part of automated fixed action patterns, skills, or habits, are within the niche, and an animal anticipates the same outcome as always. Explorative (uncertain) affordances reflect novel hypotheses associated with goal-oriented actions within the environment, and an animal is unsure about the consequences. The interpretation here is that "testing" of explorative affordances necessarily entails uncertain outcomes.

The precise terminology must first be established to discuss the predictions for the future. This work uses the vocabulary suggested by Bubic (2010) (**Figure 16**).



Figure 16. Future-oriented terminology. Precise terms for talking about the future. The present moment is the current state of the space, understood as computations of the biomechanical and inferred spaces along the relevant dimensions. The dimensions of the spaces are operationalized in the following chapters (see Section 5.1). Prospection involves computing potential, counterfactual states that are possible within the boundaries of a given space. It can be considered counterfactual thinking, planning, or playing dynamic team sports. Anticipation reflects information flow that has been prospected and is now calculated along the prospected trajectory. Thus, anticipation implies that the agent aims at a specific future state of the space, which applies to prospective information processing. Reactive information processing calculates any change in the state without prospection of alternatives. Expectation denotes a particular state in the future that is prospected and aimed at. The expectation is anticipated to be obtained after implementing a prospected trajectory from the current state of the space. The broad orientation toward the future is captured with the term prediction (adapted from Bubic, 2010).

Agents broadly *predict* the future, *prospect* possible trajectories of the states, set *expectations* about the future states, and *anticipate* changes along the dimension of the space toward the expectations. Given the above terminology, the experienced surroundings of potential futures are anticipated expectations. During prospection, alternative expectations are computed and can be experienced with the involvement of the gestalt areas. Affordances entail processing expectations and anticipations together,

as affordances are anticipations toward an arbitrarily distant potential future (expectation). Along the transition in time, an organism processes the ongoing information and experiences novel, previously uncalculated future states and the affordances toward them (Pezzulo & Cisek, 2016). This navigation is necessarily hierarchical as the hierarchy organization of the entire information processing increases the accuracy of predictions and thus is evolutionarily favored. Following multiple affordances hierarchically is known as "having an optimal grip on a field of affordances" (Bruineberg & Rietveld, 2014, p. 1). Exploitable affordances in the niche require fewer computations as the alternative trajectories do not have to be calculated, and the anticipation covers the volatility within the closed skill (see Section 1.3).

The proponents of the free energy principle posit that future-oriented information processing applies to the entirety of the brain (Bruineberg, Kiverstein, et al., 2018; Constant et al., 2021; Friston & Stephan, 2007; Hohwy, 2015). Thus, the principle shall apply to information processing in the gestalt areas (surroundings of affordances) and other areas of the brain. The free energy principle makes no commitment to any particular theory of neural organization. However, it is often merged with Bayesian inference when describing the organization of living systems (Da Costa et al., 2021; Friston & Stephan, 2007). This, in turn, has been criticized by proponents of the ecological-enactive interpretation of the free energy principle (Bruineberg, Kiverstein, et al., 2018; Kirchhoff & Kiverstein, 2021), who posit that the principle is better suited to "anti-representational" (ecological-enactive) accounts of cognition. Soon later, Constant, Clark, and Friston (2021) published an article accepting both implementations of the free energy principle, which they call "representational" and "dynamicists". In their paper, the authors argue that the former entails mental manipulations about symbolic representations, whereas the latter entails embodied interactions between the organism and its environment; thus, it excludes the intermediate representations. This notion is inspired by Gładziejewski (2016), who advocated that the free energy principle combined with Bayesian inference is representational. In contrast, the ecological-enactive accounts are considered antirepresentationalist, given that, according to Ramesy (2007), the biological receptors conveying the information about the physical world do not satisfy the notion of representations.

Distinguishing between information processing in different brain networks adds layers of complexity to this issue. Namely, processing of the sensory, somatosensory, and vestibular information is as much "non-representational" as it can be, given how the quantum information hits the biological receptors and the information is subsequently propagated along the axons of the receptor cells. Subsequent stages of information processing are gradually more "representational" as the information becomes transducted along the synapses and is gradually compressed/categorized/modified along the way (**Figure 17**). Here, the subsequent processing of information is understood as computing the biomechanical and inferred spaces, which are understood as representational. When the information reaches the gestalt areas, it computes the conscious, effortless, and pre-reflective experience of the subjective construal that is necessarily representational (see Section 3.2). Whether the early information processing is called representational or not remains to be argued by philosophers. Here it is accepted that the information processing occurring in the biological system is "representational," at least from the stage at which it is composed into biomechanical and inferred spaces.



Figure 17. Representational and non-representational information processing. The information reaching the body (sensory, somatosensory, and vestibular) is as non-representational as it can be, given that the receptors operate on a quantum scale. Thus, some researchers accept that this information does not satisfy the criteria of representations (Constant et al., 2021; Gładziejewski, 2016; Ramsey, 2007). However, the information is progressively processed in the large-scale brain networks computing biomechanical and inferred spaces that are understood here as operating on the representational format. Subsequently, the spaces are consciously experienced when coupled with the gestalt areas.

The difference between the Bayesian and ecological-enactive implementations of the free energy principle is operationalized here in terms of two kinds of beliefs that a system implements. (1)Firstly, beliefs may reflect the *existence* of the physical objects generating the sensory, somatosensory, and vestibular flow of information. The objects exist, but the information about this existence that reaches the nervous system is a representation, i.e., trees, ocean waves, or written physical equations are representations of objects (the hidden causes) generating input information. (2)Secondly, beliefs may point to *causes* beyond the online flow of the sensory, somatosensory, and vestibular information, i.e., originate from information processing in time (i.e., background knowledge). For example, the knowledge that Sally moved Ann's doll explains the lack of the doll in the initial place in terms of inferred actions of other agents (the hidden causes). Thus, the beliefs about the hidden causes are understood equivocally as either objects generating the information or inferences about why specific information has been generated.

To avoid calling one framework "representational" and the other "dynamic" when, in fact, both frameworks point to "representational" information processing, the frameworks are differentiated by specific information that they process. The former framework is associated with perception as unconscious inference (Von Helmholtz, 1886) refined later into perception as hypothesis testing (Friston & Stephan, 2007; Gregory, 1968, 1980). The latter framework treats the organism as a dynamic brain-body-environment system (Beer, 2008; Bruineberg, Rietveld, et al., 2018; Chemero, 2009; Varela et al., 2017). However, these studies do not specify what stage of information processing they refer to. In fact, the relevant literature often applies the free energy principle to the whole organism that processes information (Bruineberg, Kiverstein, et al., 2018; A. Clark, 2013; Friston, 2011, 2013, 2018; Kirchhoff & Kiverstein, 2019, 2021; Seth, 2013).

To clarify the terminology, this work assumes that the "representational" account denotes computing the inferred space in the specific brain networks, whereas the "dynamicist" account refers to calculating the biomechanical space in other brain networks. Both spaces are representational, albeit they model different information: (I) the latent causes explaining why the information is present and (II) the objects generating the ongoing information. In other words, the inferred space computes hypotheses explaining the flow of information, and the biomechanical space calculates the flow of the incoming information. The spaces shape each other in an organism's ontogeny in yet-to-be-explored interactions (see Section 5.3).

The information computed in the inferred and biomechanical spaces reaches the gestalt areas and is consciously experienced (see Section Error! Reference source not found.). The human's conscious experience differs tremendously in access to the

information that can be experienced due to many factors, for example, the differences in the sensory receptors resolution, neurodivergence, synesthesia, aphantasia, or endocrine system activity. The border at which the early information processing can no longer be experienced is also individual and cannot be drawn. At some point, information originating from the history of interactions between the computations of the biomechanical and inferred spaces reaches the gestalt areas and is experienced.

For example, experiments on information processing in the brain show that participants need around 50 ms after stimulus presentation to report its meaning reliably. However, photos displayed for as little as 13 milliseconds (ms) are recognized at a level better than chance (Potter et al., 2014). Further, studies on epileptic patients show that information processing for memory encoding needs around 300ms to reach the medial temporal lobe and the HPC (Kamiński et al., 2018; Quiroga, 2012) and around 500ms to reach the substantia nigra, which supposedly plays a vital role in encoding the information (Kamiński et al., 2018). These studies point out that the stimulus presentation, even for less than 50 ms, can be propagated in the information processing network and reach the gestalt areas several hundred milliseconds later.

The **orientation toward the future** is adopted as a general imperative of information processing in self-adaptive systems such as living organisms. From the receptors, the information is progressively transducted into biomechanical and inferred spaces and eventually reaches the gestalt areas that underpin conscious experience. The biomechanical space corresponds to the existence of objects generating the inputs; the inferred space corresponds to the hidden causes behind the inputs. Both spaces coexist under the general imperative of generating accurate future predictions. To further explore how the accuracy of anticipations is achieved, a review of memory systems is provided.

4.2 Two memory systems for encoding the statistical regularities

The free energy principle has been applied to biology makes an additional commitment to the Markov blankets, which is a statistical way of differentiating "internal states" from "external states" (A. Clark, 2017; Friston, Heins, et al., 2021; Hohwy, 2017; Pearl, 1998). The inside states hold "beliefs" about the outside states, understood as *hidden causes* behind the sensory experience (A. Clark, 2013; Friston & Stephan, 2007; Hohwy, 2020). Friston defines the "internal states" as holding probabilistic beliefs about the "external states" (Friston et al., 2023), thus internal states requires encoding and retrieval of information into memory.

In the broadest sense, memory points to the statistical regularities that a distributed information processing system manages to extract. In the second half of the XXth century, psychologists realized that memory can be divided into specific components. The initially developed model consists of sensory, working, and long-term memory (Squire, 1992) (**Figure 18**).



Figure 18. Human memory systems. The division of memory systems was suggested after decades of extensive psychological research. Contemporarily, the model is questioned in many ways: the labels shall include the ways of processing information instead of the division to implicit/conscious access (Henke,

2010); the entirety of memory is suggested to be procedural, i.e., consists of discrete series information (Keele et al., 2003); and the content of the declarative memory is proposed to be more sophisticated than a division to episodic and semantic memory suggests (Renoult et al., 2012, 2019).

Throughout the history of psychological research on memory, two types of longterm memory have been contrasted based on their respective abilities and limitations. The division into the two long-term memory systems starts from the famous case of patient H.M.⁵ (Cohen & Squire, 1980; Squire, 2009a). Procedural memory relies on individual representations that are believed to be behaviorally expressed in an inflexible manner (knowledge "how") (Eichenbaum, 2000; Packard, 2008). Declarative memory consists of relational representations and produces flexible behavior (knowledge "that") (Eichenbaum, 2001). The hippocampus (HPC) is critical in dividing the memory systems, as suggested by Squire (1992), in his own words:

Many terms have been used to describe a particular kind of memory (e.g., declarative, explicit, relational, or configural), and many other terms have also been used to describe a kind (or kinds) that are dissociable from the first kind (e.g., non-declarative, implicit, or habit). However, the terms themselves are not the proper focus. If one considers the various biological and purely psychological concepts that have been used, it is striking that they sort themselves out in terms of ideas about what the hippocampus does and does not do in the service of memory. It should not be surprising that these terms place themselves on either side of a biologically meaningful boundary. (Squire, 1992, p. 205)

The HPC is one of the most extensively studied structures in the brain, yet details of its computations and its role in conscious experience are still mysterious. The quote above shows that the engagement of the HPC is required for one memory system but not the other. To understand the role of the HPC, its functions are reviewed.

Most notably, the HPC is associated with memory encoding, i.e., creating episodic memories or acquiring concepts. Researchers have been trying to pinpoint the neurobiological substrate of memory conceptualized as engram cells since Semon and Konorski, as summarized by Tonegawa (et al., 2015). Contemporarily, the enduring

⁵ The famous case of Henry Molaison, who due to epileptic seizures underwent bilateral lesion of hippocampi and adjacent structures in 1953. As a resut, the patient lost capacity to create new episodic memories, albeit procedural memory remained intact (Squire, 2009).

neural changes resulting in memory encoding called engrams are defined as "*a population of neurons that are activated by learning, have enduring cellular changes as a consequence of learning, and whose reactivation by a part of the original stimuli delivered during learning results in memory recall.*" (Tonegawa et al., 2015, p.918). Experiments on mice led to the discovery of populations of cells created separately during contextual fear conditioning in the dentate gyrus of the HPC and the prefrontal cortex (Kitamura et al., 2017). A follow-up study revealed that the engram for a contextual fear conditioning memory is distributed across multiple brain regions (Roy et al., 2022)⁶. The distributed encoding of information about contextual fear conditioning reflects that this memory encloses multiple information channels computed in various brain networks.

Contemporary, one of the leading theories describing information processing in the HPC and the neighboring entorhinal cortex (EC) is the Tolman-Eichenbaum Machine (TEM) (Whittington et al., 2020). The TEM is an information processing system computing the structure of experience with the activity of the EC and combining this structure with sensory representations with the activity of the HPC. This line of research strongly suggests that the HPC and EC are involved in sequential, multidimensional information processing (Baldassano et al., 2017a). Albeit the information processing in the HPC is not necessarily required for the conscious experience, it contributes to its complexity by encoding the structure in the conscious experience into memory. For example, people developing Alzheimer's Disease that corresponds to degeneration of the HPC still *experience*, yet the content of their experience is tremendously different.

Behavior observed in Alzheimer's Disease points to the link between the hippocampus and navigation. Scientists who studied the animals' HPC discovered cells active in specific regions of a maze called place cells (O'Keefe & Dostrovsky, 1971). Subsequent research revealed a hexagonal grid pattern representing the maze in the EC (Hafting et al., 2005; Moser et al., 2017). This discovery was awarded a Nobel Prize in 2014 and subsequently led to the discovery of a specialized navigational system in the HPC and EC (Fyhn et al., 2004; Hafting et al., 2005; O'Keefe, 1976; O'Keefe & Dostrovsky, 1971; Sargolini et al., 2006). Besides place and grid cells found respectively

⁶ These lines of research also show that the HPC mediates memory retrieval for two weeks after encoding. In contrast, retrieval after two weeks is mediated by the population in the prefrontal cortex of mice (Roy et al., 2017). Deactivating a part of the HPC known as the dorsal subiculum prevents memory retrieval with encoding remained intact.

in the HPC (Eichenbaum et al., 1999; O'Keefe, 1976) and the EC (Moser et al., 2008, 2017), other cells found in the HPC and adjacent structures are head-direction cells (Cullen & Taube, 2017), egocentric goal-direction cells (Hok et al., 2005; Sarel et al., 2017), egocentric object-vector cells (Høydal et al., 2019), speed cells (Kropff et al., 2015), border cells (Savelli et al., 2008; Solstad et al., 2008), boundary vector cells (Lever et al., 2009), and time cells (Pastalkova et al., 2008; Tsao et al., 2018). Further, the grid structure in the EC is sensitive to rewards and goals (Boccara et al., 2019; Butler et al., 2019).

Complementary studies on humans record the activity of single cells in the HPC and advanced areas in the brains of epileptic patients (Quiroga, 2012). At the moment of creation of the concept cells (Quiroga, 2012), a theta-band activity over the prefrontal cortex of epileptic patients has been measured (Kamiński et al., 2018), presumably replicating the findings of the emergence of two distinct populations of engram cells at the moment of memory encoding (Roy, Kitamura, et al., 2017). Additionally, the phase-lock of theta band activity between the HPC and the OFC has been shown as causally underlying memory of the distribution of rewards (Knudsen & Wallis, 2020).

Further, recordings of epileptic patients revealed that successful formation of the concept cells involves the activity of putative dopaminergic cells in the substantia nigra pars compacta (SN) (Kamiński et al., 2018), which replicates finding about the dopaminergic influence on the memory encoding in the HPC identified in the rodents (Wagatsuma et al., 2018). These observations support the hypothesis that incorporating novel information into long-term memory is gated by dopaminergic influence on the HPC (Lisman & Grace, 2005) (see Section 6.2). Additionally, synchronization of the HPC and the prefrontal cortex during non-REM sleep has been suggested as a mechanism of memory consolidation (Lewis et al., 2018), supporting the findings from mice of maturation of the engram cells.

The HPC plays a crucial role in encoding the structure of a future-oriented, memory-guided, distributed information processing system. The HPC is essential for episodic memory, connecting "what" and "where" (Rolls & Xiang, 2006a; Tulving, 2002a), which can be rephrased as a conscious experience of counterfactual scenarios. An efficient information processing in the distributed system requires accurate memory of the experienced statistical regularities that guide subsequent predictions. The HPC slices a continuous stream of experience into distinct episodes (Ben-Yakov & Henson, 2018) via the activity of cells detecting time-boundaries of the episodes (Zheng et al., 2022). The multidimensional associations reflect temporal succession of events (Friston & Buzsáki, 2016a), hierarchically organized over nested timescales (Baldassano et al., 2017a). This change in granularity of the timescales is presumably implemented along the long axis of the HPC, expressing a shift in the size of the place and grid cells' receptive fields (Bellmund et al., 2018; Collin et al., 2015a). These observations are supported by experimental observations of brain activity in the functional magnetic resonance imagining (fMRI), which showed that the HPC of people watching a movie reacted to event boundaries between the movie's scenes (Ben-Yakov and Henson, 2018). Some researchers suggest that the primary function of the HPC is to construct perceptual and abstract scenes (Maguire & Mullally, 2013). Such information processing organization satisfies the principle of future orientation because it promotes the hierarchical organization of the predicting models.

For example, recent studies have shown that the HPC stores experiences in the form of events, such as subsequent laps in a maze (Sun et al., 2020). Researchers have found a specific set of cells in the CA1 (the part of the HPC where place cells are identified) activated by subsequent laps. These "event-specific rate remapping" cells are shown to be independent of the activity of place cells. The authors conclude that investigated cells "tracks the skeletal structure of experience via events as abstract entities, with abstract relationships between them." (Sun et al., 2020, p. 11). These observations suggest that the HPC is involved in identifying a transition between two states of the multidimensional, associative space.

The HPC-dependent networks have been related to navigating the physical and conceptual spaces (Bellmund et al., 2018; Gärdenfors, 2004, 2014), albeit this association does not capture the time dimension. The research reviewed above is understood in terms of the spatial organization of knowledge into exemplars of various categories and respective prototypes of these categories (Gärdenfors, 2014; Rosch & Mervis, 1975). The HPC-dependent networks involve pattern completion (including into a category) and pattern separation (dividing exemplars into different categories). Both pattern completion and pattern separation presumably occur in the HPC and are a bedrock of human cognitive abilities (Quiroga, 2020). Electrophysiological recordings on epileptic patients show that when moving anteriorly along the ventral stream, the categories become smaller and more concrete (Kamiński et al., 2018; R. Q. Quiroga, 2012). For example, the fusiform face area in the VTC has been shown to process the n-dimensional space of faces (Chang & Tsao, 2017; Quiroga, 2017), whereas the anterior portion of the temporal

lobe activates to a single exemplar denoting a unique, familiar face that is tied to the identity of the owner of the face (Landi et al., 2021).

The HPC-dependent networks have been suggested as a shared basis for memory, imagination, and future thinking (Mullally & Maguire, 2014), given the experiments showing that the computations in the HPC rely on constructing an experience of scenes (Maguire et al., 2016; Maguire & Mullally, 2013). Research on humans indicates that the HPC and the EC represent a self-location in actual space (Guterstam et al., 2015), as well as in hypothetical (counterfactual) spaces (Van Hoeck et al., 2015; Y. Xu & Corkin, 2001), and imagined spaces (Hassabis & Maguire, 2007a; Markman et al., 2009). Moreover, the HPC is a crucial hub for episodic memory, connecting "what" and "where" (Rolls & Xiang, 2006b; Tulving, 2002b), which can be rephrased as episodic imagination. The activity of the HPC slices a continuous stream of experience into distinct episodes (Ben-Yakov & Henson, 2018). This is related to the role of the HPC in organizing a temporal succession of events (Friston & Buzsáki, 2016b), as well as the hierarchical organization of the events over nested timescales (Baldassano et al., 2017b), presumably along the long axis of the HPC (Collin et al., 2015b). Overall, given the kind of computations that the HPC and the EC perform, it is suited for constructing a generative model (Friston, 2010; Goodroe et al., 2018) of the surroundings of potential futures. Navigating the rich landscape of affordances leading to potential futures involves experiencing counterfactual scenarios and making decisions. The reviewed evidence strongly suggests that the HPC is involved in processing information that contributes to such experiences.

Recall psychological research on skill acquisition, where kills are defined as sensorimotor or cognitive procedures (see Section 1.3). According to an original conceptualization, skill acquisition involves three phases: cognitive, associative, and autonomous (Fitts & Posner, 1967). The cognitive phase often goes beyond procedural memory and affects episodic and working memory (Ackerman and Cianciolo, 2000); thus, it involves computations performed in the HPC. Once the skill is learned, the involvement of the HPC is no longer required. In computational terms, skills acquisition relies on the MB controller, whereas autonomous processing is computed with the MF controller.

The skilled performance of the experts goes beyond the MF computations (Ericsson, 2006), as the exceptional abilities of experts, such as chess or baseball players, require cognitive management, which is generalized as navigating changing

environments, new situations, and uncertainty about competitors (Christensen et al., 2019). Expert performance in chess-like games is based on mastering the capacity to perceive a given situation and its development in time and implementing the right action policy, which requires information processing in the HPC. In psychology, skills are operationalized from straightforward sensorimotor procedures to skilled performance in complex environments, a broader operationalization than described here. Here, skills are exploitative chunks of information.

Complementary to the already reviewed studies, researchers of memory in educational settings distinguish the division between memory storage and memory retrieval (Bjork, 1988; Bjork & Bjork, 1992; S. M. Smith et al., 1978). The storage memory can be roughly thought of as accumulated associative learning that determines how easily retrieval is accessed and how robust it is. Retrieval memory determines what can be retrieved given the circumstances (cues, experience). This line of research shows that creating difficulties during learning, such as intertwined practice instead of block repetition or changes in the context of learning, results in weaker immediate retrieval, albeit better storage strength determining future retrieval (Bjork, 1988). After prolonged disuse, retrieval strength decreases, albeit it can be quickly reestablished (Bjork & Bjork, 1992). When surroundings are easily retrieved, no learning can occur, which reflects older ideas developed by Vygotsky, according to what learning relies on challenging oneself to reach beyond what is already learned (Del Río & Álvarez, 2007; Doolittle, 1997). One prediction of the theory is that storage strength becomes latent but never disappears (Bjork, 1988; Bjork & Bjork, 1992), which seems to be supported by recent studies on rodents (Roy, Muralidhar, et al., 2017). To further operationalize the information processing performed in the HPC and its respective phenomenological experience, the psychological models of memory are refined in the next section.

The HPC-dependent networks have been suggested as a shared basis for memory, imagination, and future thinking (Mullally & Maguire, 2014), given the experiments showing that the computations in the HPC rely on constructing an experience of scenes (Maguire & Mullally, 2013). Evidence indicates that the HPC's networks calculate self-location in physical space (Guterstam et al., 2015), hypothetical (counterfactual) spaces (Van Hoeck, 2015; Y. Xu & Corkin, 2001), and imagined spaces (Hassabis & Maguire, 2007b; Markman et al., 2009). For example, the research on the information processing underpinning the experience of natural scenes reveals that the scenes are perceived based on the statistical regularities in past experiences (De Cesarei et al., 2017). Lauer and Võ

(2022) showed that the visual perception of natural scenes involves extracting *scene grammar* from the individual encounters of a particular view. The scene grammar is established by anchor objects, such as the fridge anchoring the kitchen-related items, the shower anchoring the bathroom-related things, and so on (Boettcher et al., 2018; Helbing et al., 2022) (Boetcher et al., 2018; Helbing et al., 2022). Johnson and Jonnson (2014) observed areas involved in visual scene perception and imagination, some overlapping with the gestalt areas (OPA, PPA, RSC, Precuneus/IPS).

4.3 The entirety of memory is procedural (i.e., discrete, sequential)

Understanding how the memory systems cooperate and compete requires realizing that the "declarative memory" is also procedural as it consists of discrete events. This realization has been made by Keele (et al., 2003), who suggested that all information processing in the brain is sequential. Memory systems can be described as "procedural" because they represent a series of events (chunks, transitions). Keele and colleagues differentiated between two systems processing information: a unidimensional system corresponding to the "memory how" and a multidimensional system corresponding to the "memory that".

Empirical data support the theory developed by Keele and coauthors. Their evidence comes from psychological tasks such as the Serial Reaction Time (SRT), which measures implicit sequential learning (Curran & Keele, 1993). During this task, participants must respond to a series of stimuli presented in various modalities by giving specific responses (e.g., pressing one of four buttons). Each answer leads to the presentation of the next stimulus, which requires another reaction, and so on. Some participants are informed of the regularities in cues presentation, while others remain unaware. Some participants from the unaware group realize the regularity. Informed and uninformed but aware participants respond faster than unaware participants. After initial learning, a high or low-pitch sound played after each visual stimulus is added to subsequent blocks of the SRT. At the end of each block, the subjects were asked to report the number of high-pitched tones they heard. This introduction of the secondary task equalized the reaction times of participants from all three groups. This is interpreted as the engagement of the multidimensional system in the secondary task, so the first task is processed solely in the unidimensional system, revealing dissociation between the systems.

The findings indicate that two types of learning happen simultaneously during the initial training phase for a single task, which confirms that the memory systems work independently. In other experiments with the SRT task, subjects were divided into two groups practicing initially single-task or dual-task conditions. Participants learn better in single-task conditions compared to dual-task conditions. Nonetheless, all subjects performed comparatively when both groups were transferred into the dual-task condition. This implies that the secondary task inhibits the expression of the knowledge already

established in the multidimensional system, which shows that the memory systems compete.

Schmidtke and Heuer (1997) studied a version of the SRT task, where participants had to remember sequences of visual and auditory cues that occurred in 6-item sequences. They wanted to test whether learning of the sequences in either modality occurs independently or jointly by making the second task part of the same multidimensional sequence. The participants' reaction time indicates that the sequences bind together in the multidimensional system. However, after the shift in the initiation of the auditory sequence, the reaction time suggests that the unidimensional system processes the sequence again. Once the sequence shift in the secondary task breaks the correlation, the knowledge encoded in the multidimensional system is no longer viable. Another variant of the SRT reported in this study involved visual and auditory sequences of unequal length, yielding consistent observations.

These results were interpreted by Keele (et al., 2003) as showing that the multidimensional system creates inter-dimensional associations, whereas the unidimensional system creates intra-dimensional associations. According to Keele, unidimensional modules can extract separate representations only along a single dimension. These unidimensional modules respond only to input along a specific modality. The level of learning within these modules is not affected by whether they are trained under single-task or dual-task conditions. In the case of dual-task conditions, the amount of learning remains the same regardless of whether secondary events excluded by the module follow a random sequence, a sequence of unequal length, or a sequence of equal size. On the other hand, learning within the multidimensional system is more restricted. For this system to extract sequential associations, the successive events must either be in a single dimension or, if task-relevant information is present on multiple dimensions, the consecutive interdimensional events must be predictive. Observing the correlation between dimensions allows for extracting inter-dimensional associations with the multidimensional system, as explored in subsequent studies (Hsiao & Reber, 2001; Rah et al., 2000). The multidimensional predictability of following events reflects the real-world experience and encourages inter-dimensional learning. On the other hand, when circumstances are random, it discourages such learning.

In this line of research, dimensions are considered equal to separate modalities, albeit this distinction is an approximation. As Keele (et al., 2003) states:

The issue of what constitutes a dimension is contentious and is reminiscent of an older issue in psychology about whether dimensions are separable or integral (e.g., Garner & Felfoldy, 1970). In the SRT task, the term dimension has generally been used interchangeably with modality, and we maintain this convention. However, stimulus attributes within a modality can also constitute relevant dimensions for sequence learning, similar to the way in which visual attention studies have described fundamental dimensions for perception (e.g., Treisman, 1988). Moreover, distinctions within the motor system (e.g., hands vs. feet) may also constitute dimensions. We expect that similar principles will apply across these various situations. (Keele et al., 2003, p. 317)

Treating dimensions as modalities is a helpful simplification for the sake of experimental design. However, specific modalities (sight, hearing) have multiple dimensions that must be explored and specified. This understanding allows for seeing heuristics such as one-clever-rule (Gigerenzer & Gaissmaier, 2011) in a new light – as focusing on a single dimension. Dimensions are specified in the second chapter, where a living body is reduced to sensory, somatosensory, and vestibular channels of information flow. The channels are further divided into dimensions; for example, the somatosensory channel is roughly divided into pain, pressure, temperature, and proprioception information.

Keele's (et al., 2003) model provides a new perspective on Kamin's blocking effect, a phenomenon first described by Kamin (1969). This effect can be described with an experiment where, initially, a light is paired with a reward. Then, the light is presented with a new, not-associated stimulus, such as sound. Light and sound are jointly presented during training, and a reward follows. At some point, the sound is presented without the light. Presentation of the sound only does not produce a conditioned reaction. Keele's (et al., 2003) model proposes that the light-reward association is multidimensional and requires attentional resources. Therefore, when the sound is introduced, it is processed in the unidimensional system, which cannot make inter-dimensional associations. This implies making causal inter-dimensional associations that require attentional resources.

The division between uni- and multidimensional systems has been further developed by Gheysen and Fias (2012). They related inter-dimensional associations observed in the SRT tasks with HPC activation, whereas intra-dimensional associations with primary stimulus-specific cortices (occipital, motor, somatosensory). This division
neatly reflects the flow of sensory, somatosensory, and vestibular information corresponding to the biomechanical spaces and the information dissociated with these channels corresponding to the inferred spaces.

To summarize, various lines of research indicate that the HPC is involved in organizing the multidimensional experience into a form of discrete events. The suggested role of the HPC-EC system is constructing a navigational space at different levels of abstraction. The structure of the experience is provided with the computation done in the EC, whereas the HPC links multidimensional experience with this structure. Information processing in the HPC-EC by encoding and retrieving inter-dimensional associations is crucial for acting in the real world. Keele's work suggests that the intradimensional associations within the unidimensional system are processed independently by the early sensory cortices. These cortices are closely connected with SenSTR, validating Keele's insight about the role of the STR in computing the real-time conversion between sequence parts in a given action. The HPC activity has been tied to the multidimensional system, whereas basal ganglia (BG) mediated the activity of both uni- and multidimensional systems. The details of this information processing are investigated within computational neuroscience.

4.4 The memory systems cooperate and compete

So far, it has been stated that memory-guided information processing in the futureoriented distributed system can be differentiated into uni- and multidimensional systems. The unidimensional system is associated with the biomechanical space and linked to information processing in the SenSTR. The multidimensional system is associated with the inferred space and related to information processing occurring in the HPC, EC, OFC, and ReISTR (see Sections 2.4 and 5.2). Both models process information procedurally and hierarchically, which implies that they consist of states, sub-states, and transitions between the states corresponding to the flow of information in the n-dimensional space. The biomechanical space is enclosed in the sensory, somatosensory, and vestibular dimensions; the inferred space is an abstract, n-dimensional space of arbitrarily extracted dimensions. This section builds on psychological and neuroscientific research on information processing and reports the computational framework of instrumental actions. This light provides a more detailed operationalization of the computational structure of the spaces.

Research on memory in psychology, neuroscience, and computational neuroscience suggests that the two memory systems process information simultaneously and independently (Dolan & Dayan, 2013; Geerts et al., 2020b; Poldrack & Packard, 2003; Squire, 1992). One system is known in the literature as relational, goal-oriented, multidimensional, and model-based, contrasted with the other system known as automatic, habitual, unidimensional, and model-free (Dolan & Dayan, 2013; Geerts et al., 2020; Poldrack & Packard, 2003; Squire, 1992). The information processing in the memory systems cooperates and competes in a sophisticated way and, as a result, underlies the behavioral expressions.

Research in computational neuroscience originates from instrumental learning studies, which stems from Tolman's cognitive maps (1948) contrasted with the stimulus-response framework by Thorndike (1898) (see Section 2.3). This computational line of research analyses the balance between two controllers of the. Reinforcement Learning (RL), known as "goal-oriented" Model-Based (MB) (hippocampal-dependent) and "habitual" Model-Free (MF) (hippocampal-independent) (Daw et al., 2005; Doya, 1999; Doya et al., 2002; Geerts et al., 2020a; Smittenaar et al., 2013b; Sutton & Barto, 2018a). In the RL, an organism is an agent confronted with a task to explore and exploit its environment in a way that allows for maximizing long-run utility. The RL framework is

usually simplified to an agent navigating states through transitions. An agent builds a tree of possible decisions (transitions) by exploring them, prunes it, and chooses an optimal action policy given a task structure and reward distribution. A proportion of activity of the MB and the MF controllers maintains a balance between exploration and exploitation. The MF controller reflects computationally efficient, automatized, and inflexible habitual behavior, while the MB controller reflects computationally expensive and flexible goal-oriented behavior (Dayan, 2009).

The information processing in the controllers simultaneously and independently influences action and decision-making (Brown et al., 2012; Brown & Stern, 2014; Ferbinteanu, 2016). Degree of cooperation and competition between the networks remains intensively investigated (Daw et al., 2011; Doll et al., 2009; Fermin et al., 2010; Gershman et al., 2012; Gläscher et al., 2010b; Otto et al., 2013; Simon & Daw, 2011; Wunderlich, Dayan, et al., 2012; Wunderlich, Smittenaar, et al., 2012). The information processed within each system can differ but still result in the same behavioral outcomes. For instance, in a T-maze, mice navigate with either place strategy (allocentric landmarks) or response strategy (egocentric turns) (Packard & McGaugh, 1996), computed respectively in RelSTR and SenSTR. Two ways of learning the same information are recognized in many fields, including instrumental learning (Dolan & Dayan, 2013; Poldrack & Packard, 2003), linguistics, where this is known as the "seesaw effect" (Ullman, 2004, 2016), or broadly speaking psychology, where memory systems are characterized as cooperating and competing (Squire, 2009b). Put simply, competition between the systems can be thought of as (over)thinking that disrupts the fluency of the currently ongoing activity.

The RL framework bears the inaccuracies of the preceding research on goaloriented and habitual behavior (see Section 2.3). The utility function usually corresponds to the "desired rewards" from animal studies. Furthermore, for an agent to perform well, the structure of a task should have easily distinguishable states and transitions between the states, known as the Markov Decision Process (MDP) environment. In animal and human research, the MDP environment is investigated with multi-step decision tasks (Dolan and Dayan, 2013). Despite these limitations, the RL algorithms create agents capable of winning against humans in board games (Silver et al., 2018) and computer games (Vinyals et al., 2019), driving autonomously (Kiran et al., 2022), or maintaining nuclear fusion with unseen-before results (Degrave et al., 2022). The RL computations successfully explain and predict information processing observed in animal experiments (Geerts et al., 2020; Poldrack and Packard, 2013). The LimSTR and RelSTR closely cooperate with the HPC/EC in performing the MB computations (Poldrack & Packard, 2003), while the SenSTR underlie the MF computations (Crego et al., 2020; Daw et al., 2005; Tricomi et al., 2009). The controllers represent *opposite extremes in a trade-off between the statistically efficient use of experience and computational tractability* (Daw et al., 2005, p. 1704). The "model" refers to a mental model of the surroundings:

Two ends of a spectrum of RL methods are model-based and model-free control (where the term model refers to a mental as opposed to a computational model); it is these that have been associated with goal-directed and habitual control, respectively (Dolan and Dayan, 2013, p.315).

Understanding that actions span the spectrum between requiring and not requiring a mental model of the environment is understood here as the computational requirement for tracking counterfactual scenarios (the experience of the surroundings as a rich landscape of affordances). Affordances that are automated into the niche require fewer computations given that they tend to return consistent feedback and thus can be carried out as automated skills with a greater engagement of the MF controller. When the volatility of the surroundings increases, the skills are progressively open as their implementation involves anticipating the broader spectrum of counterfactual scenarios (potential futures); thus, the MB controller becomes involved. The HPC-dependent networks have been associated with computations necessary for the MB controller of the RL (Vikbladh et al., 2019). Once the MB controller designs and trains action plans, the behavior becomes exploitative and mediated by the MF controller as a sensorimotor program. The skill acquisition is reflected in a shift of computation from the MB toward the MF controller.

The RL's research is a computational operationalization of the organism constructing its niche. Recall that ecological psychology and evolutionary biology capture the mechanism of initial exploration and subsequent exploitation as niche construction (see Section 2.3). Evolutionary speaking, the process of niche construction is understood here as exploring affordances and encoding them into memory till they can be easily retrieved as exploitative affordances. Following affordances change the

surroundings in surprising (exploration) and anticipated (exploitation) directions, which are encoded and retrieved. Niche comprises exploitatory affordances computed with the MF controller; the environment consists of exploratory affordances that require the MB computation.

The information computed in the MB can be used for training the MF controller (A. Johnson & Redish, 2005; Sutton, 1991), which eventually executes habitual actions based on the cached value. The MB computations are done with learning successor representation based on the activity of the place cells in the HPC. In contrast, the MF computations are done with a learning value based on the activity of the landmark cells in the DLS (Dolan & Dayan, 2013; Ekstrom et al., 2014; Geerts et al., 2020b; Poldrack & Packard, 2003b). This implies that exploitatory relations underlying a niche are embodied in the brain network processing the ongoing information flow. Similar intuition is expressed in the literature implementing the free energy principle, where the top layers of the model are believed to be "cognitive," whereas lower layers are understood as "perceptual" (A. Clark, 2013). This is understood as a spectrum between flexibility and automaticity of the information that reaches the gestalt areas.

Organism acts in its surroundings by following affordances, balancing exploiting its niche with the MF controller and exploring its environment with the MB controller. The spectrum is gradual, and exploration eventually leads to exploitation. Exploitation is automatic, repetitive actions with known consequences that an agent retrieves from memory. When affordances are exploited, the agent repeats actions toward the familiar future state of the surroundings, anticipating that they have not changed since the last visit. On the other hand, exploratory affordances are, by definition, novel; thus, the outcomes of these actions are surprising and are learned by the agent. Based on lifelong experience, the agent encodes regularities in its surroundings, gradually shifting toward exploitation.

Crucially, dopamine is involved in both mechanisms, as this neurotransmitter mediates both encoding (Kamiński et al., 2018; Roy et al., 2022; Schultz et al., 1997; Wagatsuma et al., 2018) and retrieval (Berridge et al., 2005; da Silva et al., 2018; Frank et al., 2009; Menegas et al., 2018). Dopaminergic nuclei in the midbrain (ventral tegmental area and substantia nigra) reflect the computational gradient of the organization expressed in the striatum (Bromberg-Martin et al., 2010; McCutcheon et al., 2019, 2021) (see Sections 2.4, 6.1).

Contemporary computational research moved toward the hierarchical RL (HRL), where the time aspect of the transition between the states is recognized (Pateria et al., 2022) (**Figure 19**). The HRL decomposes the future state into a complex hierarchy of sub-states that fundamentally can be grounded in the flow of information in the sensory, somatosensory, and vestibular channels. As such, this improvement over the flat RL architecture provides a computational framework for conceptualizing the hierarchy of information processing in biological agents.



Figure 19. Hierarchical Reinforcement Learning. A given action can be decomposed into a hierarchy of states, sub-states, skills, and sub-skills. In the example above, the primary state (reach the destination) involves three sub-states (reach A, B, and C). The sub-states are achieved by implementing sensory, somatosensory, and vestibular information processing (skills), such as turning, overtaking, or stopping, which involve minor actions such as accelerating or steering. The sub-states are semi-Markov states that last arbitrarily long (Pateria et al., 2022). The figure is based on the example provided by Pascal Poupart during his lectures at the University of Waterloo.

The HRL algorithms evaluate the action policies when reaching an arbitrary substate. The action policies are arbitrarily long, and the algorithm operates beyond the MDP environment. Time spent in a given (sub)state is arbitrarily long and expressed as a semiMarkov state where neither states nor transitions are sharp (Daw et al., 2006; Starkweather et al., 2017). Thus, semi-Markov states reflect the animal's experience more closely than the flat architecture of the RL with the MDP environment.

Following affordances is thus implementing arbitrarily complex actions between possible futures that are hierarchically organized. Following the experienced affordances (e.g.," reach A") is performing sensory, somatosensory, and vestibular information processing (e.g., gas, brake, steering) (**Figure 20**). The surroundings are dynamic and reciprocal, given that transiting through/toward states opens alternative possible futures. Such understanding is also present in the free energy principle literature, where acting is inevitably tied to perceiving (Friston, 2010; Clark, 2013). Hypotheses based on free-energy minimization treat action and perception as two sides of the same coin. Action bears perceptual expectations, and perception is a prerequisite for action.



Figure 20. Surroundings expressed in the HRL. The surroundings are composed of the experienced potential futures that can be reached by affordances. When the affordances are implemented, they are tied to specific changes in the flow of sensory, somatosensory, and vestibular information. In the example here, the experience of reaching the subsequent sub-states is brought to the present by implementing sensory, somatosensory, and vestibular actions such as braking or steering.

4.5 Summary

Orientation toward the future drives learning the regularities in an animal's subjective surroundings. Distributed and future-oriented information processing system requires memory for navigating and storing the information about the statistical regularities. In a broad sense, memory underpins everything that shapes the current experience in a specific context. Memory is divided into sensory, working, and long-term memory, the last being declarative and procedural generalizations that one learned, such as conceptual knowledge or skills.

Two lines of argumentation discuss whether predicting the future is representational or ecological-enactive. Given that the information is transducted on the level of the sensorium, *everything is a representation*. Thus, here, the biomechanical space is interpreted as experience tied to processing information in the sensory, somatosensory, and vestibular channels, whereas the inferred space is interpreted as experience originating from information disconnected from these channels (**Figure 21**).



Figure 21. The conceptual framework of information processing in action. The entirety of information processing is distributed, future-oriented, and memory-guided. The processed information that reaches the gestalt areas is consciously experienced with the biomechanical and inferred spaces as either correlated or dissociated from the ongoing flow of sensory, somatosensory, and vestibular information.

The neurobiological structures behind the MB network are understood as computing tensors of objects and agents and their interactions regardless of the kind of surroundings (i.e., biomechanical, social). As a result of these computations, surroundings are modeled into the hierarchical generative model of possibilities and probabilities of specific tensors. The MB computations deepen the hierarchical generative models explaining the hidden (latent) causes of the experience. Once the inferred space is encoded, the affordances between the states are trained by the unidimensional MF computations, making affordances more exploitative. If the mental model in the MB computation accurately represents the surroundings, the outcomes of the action policy are accurately anticipated, and the result is achieved without surprises. The trained affordance is "free of the model," as the execution of the action policy relies on repeating past decisions in anticipation that the surroundings are stable and the same outcomes will be obtained. Processing of information by the two spaces representing the world takes place simultaneously and independently, which implies that the networks computing them influence each other in ontogeny and can interfere constructively or destructively.

5. The n-dimensional spaces of objects and agents

The previous chapter operationalized two kinds of spaces that reflect the two implementations of the free energy principle. The biomechanical spaces have been understood as the spaces where the input consists of the sensory, somatosensory, and vestibular information. The inferred spaces are the spaces that are computed with dissociation from the ongoing inputs and reflect the latent causes explaining the incoming information.

This chapter explores how the spaces are computed in large-scale brain networks. As a result of this review, the neurobiologically plausible dimensions of the inter- and intra-dimensional associations are identified as spaces computed in specific large-scale brain networks. Some spaces are calculated in the networks close to the inputs, others in the networks distant from the inputs. Once estimated, these spaces are passed to the gestalt areas, where information processing supports the conscious experience of surroundings.

5.1 Borders and dimensions of the spaces in the HPC-dependent network

The description of space processing starts with a deepened review of the studies investigating space processing in the HPC-dependent network (see Sections 2.3, 2.4, 4.2). Recall that the HPC has been investigated for more than 60 years since the case of patient H.M. described by Squire (1992). In animal studies, Poldrack and Packard (2003) showed that rodents' flexible navigation is carried out with the HPC, LimSTR, and RelSTR, whereas the inflexible navigation with the SenSTR. Then, Dolan and Dayan (2013) showed how the research on flexible and rigid actions led to the development of computational accounts of the RL. Finally, Geerts, Chersi, Stachenfeld, and Burgess (2020) grounded the computations in the activity of specific cells in the HPC and SenSTR: the HPC computations are done with learning successor representation with the place cells, while the SenSTR computations are done with learning cached value with the landmark cells (Ekstrom et al., 2014; Geerts et al., 2020a).

The HPC-dependent computations are grounded in the information processing in the grid and place cells. The cellular components of the HPC and its connectivity are assumed to be preserved in the phylogeny, as given the neural reuse hypothesis, the biological circuitry for information processing is adapted to novel circumstances along the evolution (M. L. Anderson, 2010; Dehaene & Cohen, 2007). However, the exact information processing code in the HPC-dependent networks remains under investigation. The HPC-dependent network encloses the LimSTR and RelSTR (Poldrack & Packard, 2003) and areas in the prefrontal cortex, such as the newest evolutionary orbitofrontal cortex (Gremel & Costa, 2013; Rudebeck & Murray, 2014). Some memory researchers point toward "cognitive maps" (Behrens et al., 2018; Bellmund et al., 2020; Epstein et al., 2017; Spiers, 2020; Tolman, 1948; Whittington et al., 2022); others toward "cognitive graphs" (Ericson & Warren, 2020; Muller et al., 1996; Peer et al., 2021; Trullier & Meyer, 2000). Albeit the details are uncertain, it is universally agreed that the HPC initially evolved to navigate the physical space. It has then been reused for computing imagined navigation in abstract spaces such as social or conceptual spaces. Here, the information processing specific to the latter spaces is understood as underlying the inferred space.

The inference about the hidden causes of the incoming information is exemplified as actions performed by other agents. This claim is supported by the role of the HPCdependent networks in computing "social space" (Danjo et al., 2018; Montagrin et al., 2018; Tavares et al., 2015; Yeshurun et al., 2021). The HPC-dependent networks are well suited for computing objects and agents by access to the information calculated in the ventral stream that differentiates between animate and inanimate stimuli (Connolly et al., 2012; Thorat et al., 2019). These observations indicate that human cognition originates from the capacity to infer latent causes of experience as robust models of different physical and abstract agents (e.g., countries or ghosts), as the identity and abilities of the agents are inferred arbitrarily based on a posteriori evidence.

The information processing in the HPC is gated by the EC, as reflected in the TEM's computational model (see Section 4.2) (Whittington et al., 2020). The EC is the input/output structure of the HPC (**Figure 22**). The circuit through the HPC involves structurally separated and functionally organized systems within the HPC that identify the information and encode and recall parts of the engram population (Roy et al., 2022; Roy, Kitamura, et al., 2017).



Figure 22. The simplified HPC-EC circuit. The information flow in the HPC is gated with the EC that projects from the 2^{nd} and 3^{rd} layers to the dentate gyrus (DG) of the HPC. Then, the information travels along the circuit, reaching the CA1. If the information has a novel structure, it is encoded via a direct connection from the CA1 to the EC5; the known input results in memory recall and flows through the Subiculum (SBC). The 2^{nd} and 3^{rd} layers of the EC input information to the HPC, and the 5^{th} layer of the EC receives the information back and subsequently propagates it. DG = Dentate Gyrus; CA3 = Cornu Ammonis 3; CA1 = Cornu Ammonis 1; SBC = Subiculum (Roy et al., 2022; Roy, Kitamura, et al., 2017).

The information processing in the HPC-dependent networks is inherently spatial for visual, conceptual, and abstract spaces. These observations prove that the computed space's structure consists of objects organized into hierarchical relations. The dimensionality of the spaces depends on objects, as the visual space is enclosed in the Cartesian coordinate system, whereas conceptual or abstract spaces may involve arbitrarily many dimensions. For example, the locomotion of rodents along a linear maze is characterized as a 1-dimensional scalar of position along the maze given time; 2-dimensional vectors describe a place on two coordinate axes. Tensors are "n-dimensional vectors" through n-dimensional spaces onto which an information processing system is distributed. The biomechanical space consists of dimensions enclosed in the sensory, somatosensory, and vestibular channels. The inferred space consists of abstract dimensions such as conceptual or social. The information processing in the HPC, understood in terms of constructing a scene, provides strong intuition that this structure is preserved across the various spaces.

Based on the information processing in the HPC-dependent network, it is claimed that the computed space is hierarchical over nested timescales and corresponds to navigation and decision-making through physical and abstract spaces. The biomechanical and inferred spaces are reduced to objects and agents exchanging information along relevant dimensions. For the biomechanical spaces, the relevant dimensions are enclosed in the sensory, somatosensory, and vestibular channels; for the inferred spaces, the dimensions emerge from the intentional states of other agents. Objects and agents travel through time along tensors concerning the possibility and probability of future interactions (**Figure 23**).



Figure 23. Possibilities and probabilities define the spaces. Computation of the trajectories through the spaces involves estimating the possibility of a specific trajectory and then its probability. The border cells have been identified in the EC (Solstad et al., 2008) and the SBC (Lever et al., 2009). Boundary cells have been interpreted as mediating borders in the physical and abstract spaces (Bellmund et al., 2018). Boundaries indicate that a transition in a specific direction is impossible. For the possible transitions, the studies on RL show that the HPC-dependent system is involved in the MB decision-making based on the varying probability of rewards (Feher Da Silva & Hare, 2020; Lindskog et al., 2021; Mızrak et al., 2021; Pleskac & Busemeyer, 2010). Estimating probabilities for the future is a necessary prerequisite for generating accurate predictions.

Recall that the long-term memory is procedural, i.e., consists of discrete states and transitions between these states (see Section 4.4). States are understood as circumstances that an animal generalizes as tokens of the same type (exemplars of the same category) (Gershman et al., 2010; Redish et al., 2007). For example, subsequent states reached by a delivery driver are exemplified as loading and unloading goods in specific places. The transportation between the places reflects the transition between the states. After enough repetitions, the transitions between the subsequent states become an automated skill elicited by the exploitative affordances. Thus, transitions are understood as causal links between the states conceptualized as tokens of the same type (Machery, 2009). The currently experienced state depends on the past states, which means that the agent in the past was oriented toward the future that is being experienced now. Thus, conversely, the current state of the surroundings targets future states. In a way, orientation toward the future denotes a trajectory of causes and effects between the states in the surroundings. With memory usage, more complex surroundings can be experienced and aimed at. Understanding acting animals as agents in surroundings allows for understanding actions as changes between subsequent states of the agent's surroundings. As a result, separate classes of prediction errors related to possibility and probability are encountered, resulting in calculating other agents' potential actions. Other agents included in a given agent's surroundings can act given their intentions, desires, and beliefs; thus, an agent needs to model these intentional states for accurate predictions.

In the literature about concepts, there is a discussion about whether concepts are "modal mental representations" (Barsalou, 2008) or "amodal representations offloaded to modal brain networks" (Machery, 2016). However, discussing whether "concepts" are modal or amodal is pointless, given that psychology and philosophy operationalize the term equivocally. Instead, one shall precisely discuss exemplars, prototypes, and theories (Machery, 2009). Theories are grounded in beliefs about the causes and outcomes of a particular experience (Gopnik et al., 2004; G. L. Murphy & Medin, 1985). Thus, they are akin to the Bayesian model of the hidden causes behind the experience exemplified here as other agents. Exemplars and prototypes have been investigated in psychology for over half a century (Rosch & Mervis, 1975)), contributing to the ideas such as the "n-dimensionality" of spaces (Bellmund et al., 2018; Gärdenfors, 2004). For example, the space of colors is a three-dimensional space of hue, saturation, and lightness (Gärdenfors, 2004). Extracting relevant dimensions allows for the efficient structural organization of the information processing⁷ (see Section 4.4).

All agents mutually try to infer accurately each others' intentional states that constitute the possibilities and probabilities in the mutual space. For example, depending on whether the parents are included in the spaces' computation, a child computes possibilities differently. Including a specific parent (or any agent, for that matter) changes the dimensions of the shared inferred space. Dimensions are parameters in the spaces.

⁷ The recent advancements in artificial information processing systems such as ChatGPT rely on extracting the relevant dimensions from the large data sets (Bubeck et al., 2023; Vaswani et al., 2017). It has been suggested that the HPC implements the transformer architecture in its computations (Whittington, Warren, et al., 2022). This points to an urgent research direction about the correspondence of information processing between biological and artificial systems. Understanding the dimensions of the information processing in the human's HPC might be a window for understanding both the artificial information processing.

Computations by all members of the space involve the boundaries of the space experienced as the impossibility of specific transitions. Members compute each others' actions within the accepted dimensions based on the inferred intentional states. The possibility and probability of agents' trajectories are modeled with autobiographical memory and episodic future thinking and result in assigning properties to objects and agents. Properties denote the possibilities and probabilities of transitions by other agents.

Given the above, the inferred and biomechanical spaces are computed as ndimensional spaces filled with entities capable of various movements (executing different tensors through the shared spaces). In other words, the capacity for selfpropelled motion associated with agency translates to potential transitions that other agents can implement in the shared space. For example, invasive species capitalize on the incapacity of the native species to perform specific tensors in the physical space. In the social space, dimensions such as honesty, power, and reliability are included in computing the n-dimensional space of social interactions. The possibility and probability of transitions in this space are grounded in social norms that make some tensors impossible, for example, publically wearing a specific dress at specific times. Agent breaking a social norm changes the self-distribution in the social space by modifying what is modeled as possible to do in the surroundings. This suggests that the HPC/EC computes the bordered spaces of objects and agents interacting in time according to specific rules (i.e., physical and social). The research on large-scale brain networks is reviewed in detail to investigate the neural computations behind the spaces further.

5.2 The spaces computed by the large-scale brain networks

So far, the information processing in the brain has been mainly operationalized in terms of the distance from the inputs. Now, the information processing in the brain is further specified by appealing to the large-scale brain networks.

The areas of the brain express varying structural and functional connectivity. Structural connectivity reflects the axonal projections between the regions. Functional connectivity denotes how areas are coupled in specific experimental settings. For example, perceiving an image result in activation in the early visual cortices, then the information flows along the ventral stream (the LOCT, the VTC) that contributes to the conscious experience and reaches the HPC-dependent network for memory encoding (see Section 3.2).

The last ten years brought much progress in identifying large-scale brain networks implementing functionally distinct information processing. Studies use brain imaging to determine the structural and functional coupling between the brain areas, thus pointing out the taxonomy of large-scale brain networks. However, the rapid accumulation of such studies results in inconsistent terminology and uncertainty about the number and composition of large-scale brain networks. Here, the taxonomy differentiating the six large-scale brain networks is adopted (Uddin et al., 2019) (**Table 2**).

Short	Name	Roles
ON	Occipital Network (visual)	Process early visual information
PN	Pericentral Network	Process somatosensory and motor information,
	(somatomotor)	connects with SenSTR.
MidCIN	Midcingulo-Insular Network	Process emotional (salient) and rapidly allocates
	(salience)	attentional processes toward them.
DorFPN	Dorsal Frontoparietal Network	Prepare and apply ton-down visuospatial attention
	(attention)	
LatFPN	Lateral Frontoparietal Network	Goal-oriented cognition, working memory, inhibition
	(control)	and task switching, connects with ReISTR
MedFPN	Medial Frontonarietal Network	Reconfigure associative representations, plan, engage
	(default)	in goal-directed tasks, imagine, reflect, theory of mind,
		reason counterfactually.

 Table 2. Universal taxonomy of large-scale human brain networks. The Pericentral Network (PN)

 processes information from the primary motor and somatosensory cortices. The Occipital Network (ON)

consists of the early visual areas and processes the visual information. The dorsal FrontoParietal Network (DorFPN) processes visuospatial attention. The lateral FrontoParietal Network (LatFPN) involves executive functions such as goal-oriented cognition, working memory, inhibition, and task switching. The medial FrontoParietal Network (MedFPN) is also known as the "default mode network" and is involved in reconfiguring associative representations, imagination, and Theory of Mind. The midCingulo-Insular Network (midCIN) is also known as the "salience network." It is responsible for rapidly switching attention toward behaviorally salient stimuli, such as threats, both experienced and remembered. The brain areas enclosed in the networks remain discussed, as depending on the task, different regions become synchronized—for example, the LatFPN couples with the DorFPN or the MidFPN in stimulus-driven or internally-directed tasks (Dixon et al., 2018). Based on Uddin (et al., 2019).

The ongoing flow of sensory, somatosensory, and vestibular information is predominantly processed by the ON and PN (**Figure 24**) (Gordon et al., 2017; Yeo et al., 2011). The ON processes early visual information into the ventral and dorsal streams (see Section 3.2). The PN is involved in processing primary motor and somatosensory information. Recall that the PN and ON project to associative brain areas computing the peri-spaces, such as the IPS (see Section 1.2), which is included in the DorFPN.



Figure 24. The networks that process the incoming information. The pericentral network (PN) processes somatosensory and proprioceptive information, and the visual network (VN) processes visual input (Gordon et al., 2017; Yeo et al., 2011). Note that the PN process information from the motor and somatosensory cortices, while the ON process visual information and constitutes an origin for the ventral

and dorsal streams of visual processing (Milner, 2017). The dorsal frontoparietal network (DorFPN) is involved in spatiotemporal visual attention as it encloses associative areas computing the peri-spaces (Maravita & Iriki, 2004; Serino, 2019) and areas calculating eye movements, such as the frontal eye field (FEF) and superior colliculus (SC). The figure presents a taxonomy suggested by Uddin (et al., 2019). DorFPN = Dorsal FrontoParietal Network, dlPFC = dorsolateral Prefrontal Cortex, SC = Superior Colliculus, IFJ = Inferior Frontal Junction, FEF = Frontal Eye Field, SPL = Superior Parietal Lobule, IPS = Intraparietal Sulbus, V3 and V5 = visual areas; PN = Pericentral Network, MC = Motor Cortex, SSC = Somatosensory Cortex; ON = Occipital Network.

The DorFPN, also called the attention network, prepares and applies selection for stimuli and responses (Corbetta & Shulman, 2002; Rueda et al., 2004). Thus, this network mediates top-down attention, in contrast to the bottom-up attention endogenously generated by the "salience" MidCIN (Goulden et al., 2014; Uddin, 2016). The areas included in the DorFPN suggest that this network coordinates the flow of visual and somatosensory information in the human brain.

Conversely, the information dissociated from the incoming inputs is processed by the MedFPN, also called "the default mode network" in the literature (Uddin et al., 2019). Initially, this network was identified in resting state activity when participants undergoing brain scans were disengaged from tasks (Raichle et al., 2001), hence the name "default" pointing to the baseline activity of the brain. Subsequent investigations revealed that the network consists of functionally separated subnetworks (C. Murphy et al., 2018; Uddin et al., 2019) that can contribute to performance in the ongoing tasks (Beaty et al., 2015; Crittenden et al., 2015; Sormaz et al., 2018). The MedFPN is defined in the literature as "the most distant from input" brain network (Murphy et al., 2018).

The organization of the MedFPN satisfies the criteria for computing the inferred spaces, which has been suggested in the literature (Carhart-Harris & Friston, 2010; Dohmatob et al., 2017). However, the exact details of this brain network remain mysterious, given that the network is involved in a broad spectrum of seemingly unconnected functions. Here, the MedFPN is grounded in a broader context of information processing in the brain by identifying the core areas included in the MedFPN reviewed by Uddin and colleagues (2019) (**Figure 25**).



Figure 25. The medial frontoparietal network. The network, also known as the default mode network, is operationalized as the brain network processing information most distant from the inputs (C. Murphy et al., 2018). According to the review by Uddin (et al., 2019). The core areas included in the MedFPN are (I) the medial Prefrontal Cortex (mPFC), which location extends medially from the area depicted in the figure. This area associates a broad range of information (P. Xu et al., 2019) and is considered a central hub of the MedFPN (Uddin et al., 2019). (II) The inferior frontal gyrus (IFG), which corresponds to Broka's area, produces sequences of actions in detailed resolution, such as language (Ullman, 2006). (III) The superior temporal sulcus (STS) is involved in detecting and understanding intentional actions (Hein & Knight, 2008; Shultz et al., 2011). (IV) The middle temporal gyrus (MTG) is involved in semantic processing (Wei et al., 2012), specifically in integrating conceptual information from the temporal pole (Landi et al., 2021) with potential actions (J. Davey et al., 2016). (V) The posterior inferior parietal lobule (pIPL) with the angular gyrus (AG) is involved in computing action and perception (Niu et al., 2021), and this region is highly developed in humans (Igelström & Graziano, 2017). Areas not shown in the figure include (VI) the parahippocampal cortex (PHC) that computes contextual associations (Aminoff et al., 2013) and (VII) the posterior cingulate cortex (PCC), which is located approximately in the center of the brain, is one of the main hubs of the MedFPN and is involved in broadly understood "internally-directed cognition" (Leech & Sharp, 2014). mPFC = medial prefrontal cortex; IFG = inferior frontal gyrus; STS = superior temporal sulcus; MTG = middle temporal gyrus; pIPL = posterior inferior parietal lobule; AG = angular gyrus; PHC = parahippocampal cortex; PCC = posterior cingulate cortex.

The reviewed functions of the components of the MedFPN indicate that this network is involved in processing abstract associations between abstract information. To investigate these computations further, the subdivisions of the MedFPN are analyzed.

The MedFPN can be decomposed into a core network that constitutes a communication hub and two subnetworks: (I) medial temporal and (II) dorsal medial (**Figure 26**). The core network specializes in self-processing, the subnetworks in mind-wandering and mentalizing (Uddin et al., 2019)⁸



Organization of the Medial Fronto-Parietal Network in primates:

Figure 26. Organization of the MedFPN. The network is divided into three subnetworks associated with self-referential processes (red), imagination (violet), and attributing mental states to others (green). (based on Andrews-Hanna et al., 2010; 2014; Uddin et al., 2019).

Note that the medial temporal subnetwork areas overlap with the HPC-dependent network discussed in the previous chapters (see Sections 2.3, 2.4, 4.2, 5.1). This network has been called a "generic recollection network" (Hayama et al., 2012; Schacter et al., 2012) as its functional role involves imagination and memory recall. Essentially, these

⁸ Note that Uddin (et al., 2019) include the mPFC and PCC into the medial temporal subsystem. They differentiate two subnetworks, not three as in Andrews-Hanna (et al., 2010; 2014).

observations point out that the scene computations in the HPC (Maguire et al., 2016; Maguire & Mullally, 2013) apply to the entire medial temporal subnetwork.

The **dorsal medial subnetwork** is characterized in the literature as the mentalizing network involved in the inference of mental states of other agents (Spunt & Lieberman, 2012; Van Overwalle & Baetens, 2009). According to Andrews-Hanna (et al., 2010), this network consists of dmPFC, TPJ, and MTG-STT (called there the lateral temporal cortex - LTC). The areas suggested above differ from those included in MedFPN by Uddin and colleagues (2019). Their other study reviewed research on self-related cognition that points to the mirror neuron system composed of the "*inferior frontal cortex*" and "*rostral part of the inferior parietal lobule*" (Uddin et al., 2007 Box.2), which later are labeled as IFG and aIPL and are included in the MedFPN (Uddin et al., 2019). However, as shown in Andrews-Hanna (et al., 2014, p. 31), the IFC (Broka area) is active only in the resting state, while in functional imaging, the activation of the IFC is missing. This possibly implies that IFC's activity in rest corresponds to internal dialogue. Nevertheless, such observation points out the importance of time and context in which the networks are investigated.

The **core network** is composed of the mPFC and PCC (Andrews-Hanna et al., 2010, 2014; Uddin et al., 2019). These two areas are central hubs of the MedFPN, expressing the most robust connectivity with the remaining parts of the network. Some studies suggest that in addition to the mentioned structures, the core regions are also the pIPL, MTG, STS, and SFG (Yeo et al., 2011). However, these regions are not counted in the core regions of the MedFPN in subsequent articles (Andrews-Hanna et al., 2014; Uddin et al., 2019).

The MedFPN remains to be operationalized more precisely as the available literature differs in the adopted taxonomies and terminologies. This issue, albeit addressed by Uddin and colleagues (2019), still needs to be solved. Therefore, in the remaining part of this review, the MedFPN is analyzed broadly as the core network and two subnetworks combined. Additional research on imagination and the evolutionary roots of social processing are reviewed.

The MedFPN is associated with future imagination (Bellana et al., 2017) and imagery vividness (Lee et al., 2021). The imagery of specific situations may resemble actual experiences to the extent that fear of extinction (Reddan et al., 2018) or muscle growth (Ranganathan et al., 2004) is observed in the imagining participants. The MedFPN is crucial for imagination, an experience of sensory, somatosensory, and vestibular information disconnected from the ongoing inputs (C. Murphy et al., 2018). This operationalization makes the imagination content unbounded and distributed along the exploration-exploitation spectrum, as in the already mentioned example of the memory palace (Mecacci, 2013). Research on imagination focuses mainly on visual imagination, where a range between *incapacity* to imagine visually (aphantasia) (Fulford et al., 2018; Milton et al., 2021) and experience of the combination of senses (synesthesia) to real-like vividness of imagined scenes (Mecacci, 2013) have been identified. Imagination is broadly divided into four kinds associated with the MedFPN: dreaming, mind-wandering, autobiographical narratives, and counterfactual thinking (Carroll, 2020). This paragraph points out that imagination is idiosyncratic, and its neural correlates span the HPC-dependent network and the MedFPN.

In some studies, the mPFC is labeled the orbitofrontal cortex (OFC), which is the most evolutionary recent structure of the brain that performs sophisticated information processing (Daw et al., 2006; Valentin et al., 2007). The OFC balances the activity between the parts of the striatum (Gremel & Costa, 2013) (see Section 2.4). The functional organization of the OFC is believed to reflect the computations on the mental model that the MB controller is involved in (Rudebeck & Murray, 2014) (see Section 4.4). The anterior part of the OFC represents the stimulus associated with the outcome (e.g., candy packaging), the posterior part signals the sensory qualities of the specific effect (e.g., the taste of a candy), and the lateral portion correlates with computing choices, the medial portion signals the consequences of an option which an animal is about to make (Rudebeck & Murray, 2014). However, the role of the OFC is even more elusive, given that without the OFC, an animal can still learn, relearn, and replace associations, although more slowly than its healthy counterparts (Wilson, Takahashi, et al., 2014). This kind of information processing involving perceptually inaccessible information is also known as cognitive control (Badre, 2008). Thus, the suggested function of the OFC is to recognize the perceptually similar states that differ regarding some perceptually non-available features (Wilson, Takahashi, et al., 2014), which reflects a mental model of the surroundings expressed as time trajectories of objects and agents onto which an information processing system is distributed.

Additionally, the MedFPN in *H. Sapiens* is differently organized than in the other primates (Garin et al., 2022). In all primates except humans and chimpanzees, the core regions of the MedFPN (mPFC and PCC) are not connected, which explains how chimpanzees can understand psychological states (Tomasello et al., 2003). The MedFPN

is involved in broadly understood social information processing (Buckner & Carroll, 2007; C. G. Davey et al., 2016; Li et al., 2014; Spreng et al., 2009; Wang et al., 2021; Yeshurun et al., 2021). The phylogenic development of the core subnetwork of the MedFPN appears to support the evolutionary arguments that the capacity to establish joint attention has driven human evolution (Tomasello, 2010, 2019; Tomasello & Rakoczy, 2003).

Taken together, the MedFPN is decoupled from experience (C. Murphy et al., 2018; M. Zhang et al., 2022). It contributes to information processing in multiple functional systems during action execution (Mancuso et al., 2022; Tong et al., 2022). Contrary to initial findings (Raichle et al., 2001), the MedFPN is *not* negatively correlated with performing tasks and is involved in numerous executive functions (Cole et al., 2014; Sormaz et al., 2018; Spreng et al., 2010; Vatansever et al., 2015). Therefore, the MedFPN is understood as computing the inferred spaces that contribute to the conscious experience of surroundings for the sake of actions.

The activity of the "attention" DorFPN and the "default" MedFPN may seem anticorrelated as the former network computes biomechanical space grounded in the incoming information, while the latter computes the inferred space distant from inputs. However, studies show that during ontogeny, the MedFPN trains the immature DorFPN (M. Chen et al., 2022), and the networks are increasingly less anticorrelated in older adults (Spreng et al., 2016). This is understood as a neural manifestation of adulthood, i.e., commitments to specific life aspects that are no longer counterfactually considered. Nevertheless, the volatile surroundings experienced by acting organisms involve uncertainties that must be resolved. The role of adjusting the information flow between the networks depending on the task's demands is served by a third network – the LatFPN.

The LatFPN is also called a control network, given its broad role in the executive and goal-directed control of the information flow (Smith et al., 2009; Uddin et al., 2019) (**Figure 27**). This network is involved in working memory, inhibition, task switching, and goal-oriented actions.



Figure 27. The lateral frontoparietal network (LatFPN). This network is also called a control network associated with executive control (Smith et al., 2009; Uddin et al., 2019). LatFPN = Lateral FrontoParietal Network; MCC = Middle Cingulate Cortex, dPrecuneus = dorsal Precuneus, dmThalamus = dorsomedial Thalamus, RelSTR = Relational Striatum (head of caudate nucleus), MFG = Middle Frontal Gyrus, aIPL = anterior Inferior Parietal Lobule, ITC = Inferior Temporal Cortex.

The areas included in the LatFPN are also labeled as the multiple-demand network (Duncan, 2010) associated with intelligence (Duncan et al., 2020). Other names are the cognitive control network (Niendam et al., 2012), the frontoparietal control network (Dosenbach et al., 2008; Vincent et al., 2008), the domain-general system (Fedorenko et al., 2013), and the central executive network (Seeley et al., 2007). Functional connectivity of information processing in the LatFPN predicts the performance in complex tasks such as playing demanding games (Chenot et al., 2021). The information processing in this network is coupled with the DorFPN and MedFPN, depending on the task's demands (Dixon et al., 2018), which points to the role of this network in balancing the activity of the MedFPN and DorFPN (**Figure 28**).



Figure 28. The areas included in the three networks. The core areas of the DorFPN (red), MedFPN (violed), and Lat FPN (orange) are suggested by Uddin and colleagues (2019). Notice how areas from all three networks overlap with gestalt areas, specifically the IPS, aIPL, pIPL with AG, and V5.

The information processing in the large-scale brain networks encloses "non-representational" information received by the receptors and "representational" information processed within the specialized brain networks. The information computed in the ON and PN are as "non-representational" as possible, given that these networks correspond to primary sensory areas. The information in the DorFPN is organized into biomechanical space; thus, its format is interpreted as "representational". Similarily, the MedFPN is a "representational" network because it computes the inferred space. The LatFPN balances the activity between the DorFPN and MedFPN depending on the following affordances and incoming information. All three networks contain regions classified into gestalt areas (**Figure 29**). Specifying their role is, however, beyond the scope of this article. Instead, the focus is on identifying the structure of the inferred space computed in the MedFPN by reviewing the research on the Theory of Mind.



Figure 29. Neural computations behind the framework. Gestalt areas mediate the conscious, effortless, and pre-reflective experience of the subjective construal (Lieberman, 2022). Experience of the construal for the sake of action is an experience of the surroundings understood as the rich, hierarchical landscape of affordances organized from exploitative (niche) to explorative (environment). The three large-scale brain networks contain the regions included in the gestalt areas.

5.3 The inferred space is investigated in the Theory of Mind (ToM)

The surroundings computed with the inferred space involve information processing dissociated from the flows of sensory, somatosensory, and vestibular information; thus, it is claimed that this information is computed in the MedFPN as n-dimensional space of latent causes, exemplified as actions of other agents. When the information reaches the gestalt areas, it is consciously experienced as agents with intentional states (desires, intentions, and beliefs) that affect their potential movements. This understanding is hardly used in computational neuroscience as research on instrumental conditioning focuses on "desirable" outcomes (Dolan & Dayan, 2013), initially developed in animal research and directly adapted to research on humans (Dickinson, 1994; Dolan & Dayan, 2013). It happened without the interdisciplinary collaboration with philosophers, who have been arguing for the complexity of intentional human states (Bratman, 1991; Mele, 1992; Pacherie, 2008; Searle, 1983), which originate from the richness of possible causes and reasons for actions (Davidson, 1963). The complexity of intentional states in H. Sapiens makes direct translation of desirable states from the rodent literature inappropriate. The processes behind the inference of other agents' intentions, desires, and beliefs are omitted in the research on instrumental learning. They are, however, recognized in psychological research on the Theory of Mind (ToM).

The research on the ToM points to an inference of intentional states of other agents as factors determining the modeled possibility and probability of agent's movements in the social space. Agents interact predictably/surprisingly and create complicated relations such as symbiotic (parasitism – commensalism – mutualism) or predator-prey (Bousquet & Le Page, 2004; Farmer & Foley, 2009). Thus, the experience of the surroundings involves recognizing entities with agency (which correlates with the capacity for self-propelled motion). The perceived agency is understood as the factor differentiating between processing an identity of certain information in the VTC as animate agents and inanimate objects. This implies that the characteristics of the self-propelled movement of the experienced agents should also be processed, such as their inertia and acceleration. Thus, the information processing behind the inferred space necessarily entails modeling other agents' capacities to move according to the possibility and probability of these movements. Multiple agents process information simultaneously, mutually modeling themselves as parts of their surroundings that must or mustn't be reckoned with.

The ToM investigates how agents acquire beliefs about the intentional states of other agents (Frith & Frith, 2006; H. L. Gallagher & Frith, 2003; Goel et al., 1995). In developmental psychology, children's ToM is usually tested with a scene of two dolls that differ in knowledge about the location of an item – a test known as the Sally-Anne test (Baron-Cohen et al., 1985; Korkiakangas et al., 2016). To pass a test, a child has to adopt the perspective of a particular doll, which reflects inference about its beliefs.

The inference of intentional states of other agents is also investigated in sociological research on children and primates, particularly by Tomasello (2009). Tomasello argues that the ability to create shared intentions is the basis for modern human culture. In studies of toddlers, Tomasello found that children as young as nine months old can establish and maintain joint intentions when playing with their caregivers. Children develop collective intentionality by the age of three. Collaboration among humans is unique due to our ability to establish mutual understanding toward a common goal (Tomasello, 2009). Chimpanzees work together but do so competitively, without fairness between individuals (Tomasello, 2019). Humans can establish a joint goal and regulate the behavior of others involved through normative statements that reflect the possibility and probability in the computed spaces. According to Tomasello, humans can self-regulate through a sense of obligation, which corresponds to establishing borders of space by normative statements. These findings are interpreted here as an experience of a potential future mutually shared between the involved agents. It means each participating agent experiences affordances that require the remaining agents to act.

The studies on the ToM point out that humans can establish potential futures in shared surroundings. Contrary to other aspects of distributed information processing among animals, this capacity for collective action is kept from other primates. Humans are unique in their ability to process information socially and work together to achieve joint goals. The surroundings are experienced, given the inferences about the intentional states of other agents in the multiagent agent's surroundings. Once surroundings are mutually shared, all involved agents can follow the affordances toward the potential futures, and along the subsequent repetitions, they become exploitative. For example, interactions during grocery shopping or traffic are exploitative affordances with culturally shared anticipations. Such actions are known as deontic actions (Constant et al., 2019). In other words, these actions are ontologically acquired skills. Involved agents accurately model their mutual trajectories in the inferred space. In such cases, every agent accurately retrieves beliefs about the causes and effects of the shared surroundings.

The surroundings computed from the inferred space consist of potential interactions of the agents, including the organism itself. This builds on the notion of experiencing and remembering self (Kahneman and Rills, 2005; Zajchowski et al., 2017). The experiencing self experiences the flow of sensory, somatosensory, and vestibular information computed in the DorFPN; the remembering self experiences the memory dissociated from the ongoing flow, presumably calculated in the MedFPN. The remembering self computed in the inferred space is investigated in psychology as the episodic memory related to one's past, known as autobiographical memory (Tulving, 2002). The capacity for remembering autobiographical memories differs between individuals (LePort et al., 2012; Palombo et al., 2015, 2018; Watkins, 2018). Autobiographical memory reflects the remembering self, given that it captures what a specific individual considers himself to be. It is here recognized that one frames oneself into the model of intentional states, similarly to other agents into models of their intentional states⁹. In essence, the agent's beliefs about hidden causes and effects translate to models of the agent and other agents in the computed spaces¹⁰.

One needs to note that agents infer models of other agents as causes and effects of actions in a biased way, ultimately leading to naïve realism (Lieberman, 2022; Ross & Ward, 1996). The surroundings are inferred with a subjective bias, albeit to a certain depth of computation; these surroundings are intersubjective (aligned) within a particular class of individuals. For example, terms such as "neurotransmitters," "money," or "Zeus" are intersubjective insofar as the groups of individuals, such as a research group or a tribe, infer it intersubjectively as hidden causes and effects.

Given the biased inference, surroundings may consist of social, economic, geopolitical, biological, chemical, or supernatural agents, such as other people, countries, genes, chemical agents, or imaginary friends. Humans are the only known biological information processing system capable of inferring abstract agents as the causes behind sensory, somatosensory, and vestibular information flow. This observation is behind the development of human knowledge as it relies on creating abstract interacting entities

⁹ An exciting direction to consider is how an agent reacts to its inference about the intentional states of others, which are about the inferring agent. This may be triggered by being observed, which essentially incorporates an observing agent into the observed agent's network.

¹⁰ Mutual modeling is unlimited as an agent may hold beliefs about the intentional states of other agents, which themselves might be beliefs about intentional states. Human A can have beliefs about human B's beliefs about Human A, etc.

(e.g., physical forces) that explain the experience of the surroundings. After repeated interactions, the modeled agent no longer needs to be modeled, as its intentional states have been inferred in the past and currently constitute part of the shared space (Figure 30). In such cases, they become pre-reflective causes and effects constituting a transparent lens of experience. Once a surprising experience of the surroundings is encountered, a new explanation of this experience is generated within the inferred space.



Figure 30. Experience of other agents. Once B is inferred by C with any sensory modality (sight, sound, smell), C starts to change the flow of its sensory, somatosensory, and vestibular information, which results in rapid locomotion. Experiencing the C, A adjusts its information processing and follows C in rapid movement, despite that for A, the inference does not result in the token B but in a broader type of "dangerous agents." The "inference" here is not *precisely* the information hitting the sensory receptors but the meaning of this information conceptualized as the existence of other agents. The agency is usually associated with a capacity for a self-propelled motion through time. Notice that the "motion" of other agents means sensory, somatosensory, and vestibular information processing done by these other agents and affecting information processing in all other agents in a given multiagent system.

The research on ToM investigates how other agents are processed in the various brain networks. Fundamentally, agents interact by exchanging sensory, somatosensory, and vestibular information computed in the PN, ON, and DorFPN. Reflection about the interactions is calculated in the MedFPN, which mediates the inference of the causes behind the ongoing information. The LatFPN controls the relative synchrony with the DorFPN and MedFPN. The research on ToM explores how the multiple agents mutually process information that allows for establishing intersubjective spaces and surroundings.

5.4 Summary

The multiagent surroundings rely on information processing that computes agents as trajectories through space. Each agent performs inference and encodes beliefs about the intentional states of every other agent, which results in the computation of possibilities and probabilities of the agents' transitions within the n-dimensional space. The biomechanical spaces are computed in the DorFPN; the inferred spaces are calculated in the MedFPN. The information from these networks is balanced with the LatFPN. When coupled with gestalt areas, the spaces are experienced as the surroundings.

Processing information about any space requires relations between the entities embedded in a given coordinating system and transitions gradient, such as attracting and repealing states. The biomechanical space is underpinned by processing information about the physical space provided by the interactions with the biological body, described as information flow in sensory, somatosensory, and vestibular channels according to the rules of classical mechanics. The inferred space is underpinned by the computation of ndimensional space of agents interacting according to the causal rules of possibility and probability of transitions. The intentional states of other agents (*their* desires, intentions, beliefs) are unknown to a given agent and must be inferred, which is investigated as the Theory of Mind (ToM).

Conscious experience is tied to the information processing in the gestalt areas, which are differentially coupled with the DorFPN and MedFPN. The conscious experience computed in the gestalt areas leans toward either the ongoing flow of information from the sensory, somatosensory, and vestibular channels or is dissociated. The organization of information processing in large-scale brain networks reflects the distance from the sensory, somatosensory, and vestibular inputs. The DorFPN (with the ON and PN) processes the information close to these inputs, the MedFPN processes information far from these inputs, and the LatFPN is a control network coupled with either the DorFPN or MedFPN, depending on the task's demands. Thus, the large-scale brain networks are suitable for processing the information related to the sensorium and the hidden causes behind the sensorium.

The biomechanical and inferred spaces are both possible to automatize. The automatization of the former closes a skill (makes an affordance exploitable). The automatization of the latter makes the explanation default, given the already inferred intentional states of other agents.

For example, the rising Sun can be explained by religious beliefs and physical forces; once adopted by conspecifics, the view is culturally shared until the belief system undergoes a revolution, such as Heliocentric Theory. Explanation of the ongoing experience with the inferred space takes place by inference of actions of other agents. For example, computer programs, public transport, electricity, and money are essentially agents. Without them, one would not be able to explain the modern society.

Skills acquisition is understood as selective control of information flows toward the most informative parts of the surroundings. It is hypothesized that the autonomous phase of a skill mediated with the SenSTR reflects well-trained actions that could have been multidimensional during the initial training. Still, subsequent repetitions map the possibilities and probabilities onto the space, reflected in training the DorFPN with the MedFPN. The successive repetitions allow for extracting the most informative dimension of the spaces that must be controlled in a specific time window.

Given that the spaces comprise animate agents and inanimate objects, an information processing system is assumed to compute the future transitions of other agents concerning the possibility and probability of actions taken by them. Actions are thus tensors in the n-dimensional spaces that underlie the experience of possible affordances between the subsequent states in the surroundings filled with animate agents and inanimate objects. This realization implies that the system computes the objects and agents with their time trajectories. In other words, an information processing system is distributed in tensors of animate agents and inanimate objects.

6. Potential applications of the conceptual framework

The conceptual framework of information processing in action aims to coherently describe a perplexing problem investigated in numerous disciplines. Namely, what is the flow of information during the entirety of the system's development? The framework is limited to the action-oriented information processing in the neocortex that results in an animal's sensorimotor actions and experience of possibilities for action. Observed externally in neuroscience and biomechanics, actions are sensory, somatosensory, and vestibular information changes. Witnessed internally, actions depend on the conscious experience of potential futures and anticipating the trajectory toward expected futures. The external and internal perspectives on actions are combined into a coherent conceptual framework.

The motivation behind this work was to create a conceptual framework capturing how animals process information when acting. The information processing in any adaptive system is governed by the principles of (I) distribution, (II) orientation toward the future, and (II) reliance on memory. The biological body acting in the physical space is a distributed system processing sensory, somatosensory, and vestibular information. Specialized receptors in the body receive information and propagate it along an animal's nervous system.

In the human brain, the visual information is initially processed in the ON, the somatosensory information is processed in the PN, the auditory information is processed in the primary auditory cortex, and the vestibular information enriches the information processing throughout the entire brain. The primary sensory cortices receive information already transducted by subcortical structures; thus, they do not correspond directly to the physical world as sensed by the receptors. Subsequently, the information is transmitted to the associative areas of the neocortex, such as those underlying the multimodal body schema of an animal in the IPS. The DorFPN orchestrates these associative computations because this network contributes to visuospatial attention and prepares and applies top-down selection for stimuli and responses that require global body coordination. In other words, the DorFPN is a network computing the biomechanical space of objects and agents that exchange information relevant to the primary sensory cortices.

The biomechanical space is consciously experienced when the activity of the DorFPN is coupled with the activity of the LatFPN that encloses some of the gestalt areas. In such a case, one consciously experiences the ongoing flow of sensory,

somatosensory, and vestibular information commonly understood as the physical world. However, the experienced content is not the physical world but the biomechanical space representing the world based on the ongoing flow of information from the receptors.

The conscious experience dissociated from the ongoing flow of information originates from the functional coupling between the MedFPN (default) and the LatFPN. This resonance underpins imagination, counterfactual, episodic, and future thinking, given that the MedFPN encloses the HPC and the EC. The inferred space computed in the MedFPN, when it enters the LatFPN, is experienced as "thoughts in mind".

The experience has been operationalized as a conscious, coherent, effortless experience of a subjective construal. Such defined, conscious experience denotes the same as what consciousness means – a personal act of being a specific system in specific circumstances. Out of all possible phenomenological content, this work focuses on the action-oriented experience of affordances. Affordances denote all action-oriented qualia that can be experienced based on the history of computations in the biomechanical and inferred spaces. The entirety of experience for the sake of action (the rich landscape of affordances) is called here surroundings. The surroundings consist of affordances that are explorative and exploitative. Explorative affordances constitute the animal's environment; exploitative affordances form the animal's niche; the environment and niche combined are surroundings.

Affordances are statistically independent from the ongoing flow of information (i.e., can be coupled/decoupled with it). Thus, the experience of affordances in surroundings arises from the combined experience of parts of the biomechanical and inferred spaces computed in the DorFPN and MidFPN resonating with the LatFPN activity. The underlying computations between the networks remain to be investigated.

Notice that one's experience of the landscape of affordances (the surroundings) is immensely subjective and originates from the past and current activity of the brain networks. The most detailed studies investigating the structural and functional architecture of the brain networks are studies on memory investigating the HPC-dependent network. The HPC and the EC have been identified as the parts of the MedFPN that contribute to constructing the scene when mind-wandering. The HPC and the EC operate on the place and grid cells that identify statistical regularities in the incoming information. The EC detects the structure in the signal; the HPC binds the multisensory experience to this structure. These computations result in encoding and retrieving the structure of the spaces expressed as prototypes subtracted from the exemplars of specific
categories. The close cooperation with the MedFPN suggests that the above structure is universal for computing the inferred spaces. Humans can calculate inferred spaces far more complex than any other information-processing system. The details of information processing in the human brain remain an open question that can finally be asked with the conceptual framework developed in this work. The ontogeny of encoding and retrieving the information in the large-scale brain networks and their lifelong interrelations remain to be investigated. Here, the framework is applied in some potential directions.

6.1 Encoding and retrieval of information, not behavior

Information encoding and retrieval are investigated primarily in the computational neuroscience's framework of reinforcement learning (RL) (see Section 4.4). The RL explores how agents navigate the volatile environment with memory. It can be thought of as navigating the surroundings of affordances toward potential futures. Some affordances have been followed so often that they become exploitative and constitute a niche. Experience of such affordances entails information processing resulting in executing a skill, and when a skill results in surprisal (e.g., when the space has been externally modified), it is called a habit. In other words, repeating a specific action when the researchers devalue its outcome is called habitual action. For example, if one should not use a sink due to malfunction, but the sink is used as a part of a bracketed sequence when using a bathroom. Given the novel circumstances, this behavior change requires diverting from the most likely (exploitative) path.

The crucial insight here is that the encoding and retrieval of information occur simultaneously and independently in the DorFPN and MedFPN. This reflects two long-term memory systems operationalized here as the biomechanical and inferred spaces (see Chapter 4). This realization is rarely explicitly stated in the field of RL, albeit the field recognizes that the memory systems (controllers) cooperate and compete (Doeller et al., 2008; Drummond & Niv, 2020; Hartley et al., 2003; Iaria et al., 2003; Marchette et al., 2011; Packard, 2008). The memory systems must independently and simultaneously encode and retrieve information to be able to cooperate and compete. Despite that, the RL framework recognizes encoding and retrieval of *behavior*, not information in the networks (**Figure 31**).



Figure 31. The RL recognizes the encoding-retrieval of behavior. In computational neuroscience, action acquisition is the gradual learning of a specific behavior based on an action-outcome correspondence. Encoding of the overt behavior is linked to green areas, whereas behavior's retrieval to orange. However, the shift in activity results from passing the execution of the behavior from the green toward the orange areas, which is consistent with the assumption that both networks encode and retrieve information independently. Based on: (Dolan & Dayan, 2013; Geerts et al., 2020a; Guterstam et al., 2015; Hikosaka et al., 2002; Olsen et al., 2012, 2019; Strange, 2022). OFC = orbitofrontal cortex, vmPFC = ventromedial prefrontal cortex, ACC = anterior cingulate cortex, TP = temporal pole, aHPC = anterior hippocampus, NACC = nucleus accumbens, pHPC = posterior hippocampus, PCC posterior cingulate cortex, dlPFC = dorsolateral prefrontal cortex, MC/SSC = motor cortex / somatosensory cortex, PC = parietal cortex.

Information is encoded independently in the MedFPN and DorFPN that process statistical regularities in the inferred and biomechanical spaces. The biomechanical space

is grounded in encoding and retrieving the statistical regularities in sensory, somatosensory, and vestibular information flow. The inferred space is grounded in encoding the hidden causes behind this flow, understood as the results of the actions of other agents in the n-dimensional space. The degree of competition and cooperation between the networks remains investigated. For example, a recent study shows that the activity of the MedFPN scaffolds the immature DorPFN, which might be interpreted as developing a broader context behind the incoming information (M. Chen et al., 2022). The spaces are thus simultaneously computed and consciously experienced as the surroundings of possible affordances

Two independent encodings into separate memory systems are supported with further studies. The information processing in the SenSTR is independent of the information processing in the RelSTR (Thorn, 2010), and the animals without the RelSTR still *learn* a behavior (Poldrack and Packard, 2003), pointing to the learning mechanism independent from the RelSTR and, by extension, from the HPC. The encoding in the DorFPN relates to developing the capacity for engaging with the ongoing flow of information in a way that allows for an effective transition between two arbitrarily distant states in the spaces. For example, training the execution of the penalty kick involves encoding the flow of information and adjusting to the variability of the circumstances by closing a skill with its redundant subspace covering the possible volatility of the space (see Section 1.3).

This evidence implies that "the brain" is not a homogenous organ, so it is too imprecise to be used in scientific writing. The brain and body host multiple networks; one should refer to precisely pinpointed networks when arguing what a given network does and does not.

6.2 Dopamine contributes to learning and motivation in the networks

Contrary to popular belief, dopamine *does not* process rewards per se (Cannon et al., 2003). Instead, dopamine is involved in processing motivation and learning. For example, degeneration of dopaminergic cells in the human Substantia Nigra (SN) is the direct cause of the Parkinson's Disease. Besides the SN, dopamine is released from the Ventral Tegmental Area (VTA). The SN and VTA are neighboring structures that send projections to the broad set of brain areas (see Section 2.4).

Dopaminergic neurons project to the HPC-dependent and HPC-independent networks. Most notably, axons of dopaminergic neurons reach the STR, HPC, and Amygdala (AMY) in a highly organized way (**Figure 32**). The information processing in the STR plays a vital role in the activity of the large-scale brain networks as it influences the activity of the whole neocortex (Graff-Radford et al., 2017; Rolls, 2016).



Figure 32. Dopaminergic circuits with the HPC and STR. The dopaminergic nuclei (red) send rich projections to the STR and HPC, thus contributing to the information processing in the entirety of the brain. The gradient of axonal projections across the dopaminergic structures is reflected in subsequent parts of the STR organized in a rostral-caudal gradient. The LimSTR closely communicates with the prefrontal cortex (vmPFC, OFC, ACC). The RelSTR communicates with the dorsolateral prefrontal cortex (dlPFC) and premotor cortex (PMC). The SenSTR communicates with motor and somatosensory cortices (MC, SSC). The connectivity gradient is reflected in bidirectional connections with dopaminergic nuclei in the midbrain – Substantia Nigra (SN) and Ventral Tegmental Area (VTA). The dopaminergic structures send projections not only to the STR but also to the HPC. Based on: (Ferré et al., 2018; Goodroe et al., 2018a;

Graff-Radford et al., 2017; Hooks et al., 2018; Lanciego et al., 2012; McCutcheon et al., 2021; Sherman, 2017; Xia et al., 2011). References in the Figure: (Bannerman et al., 2014; Bari et al., 2020; Cardinal et al., 2002; Dalton et al., 2022; Duszkiewicz et al., 2019; Ferrucci et al., 2013; Goldman-Rakic & Selemon, 1986; Haber, 2003, 2016; Kamiński et al., 2018; Kempadoo et al., 2016; Kunishio & Haber, 1994; Künzle, 1978; Lisman & Grace, 2005; McNamara & Dupret, 2017; Strange, 2022; Takeuchi et al., 2016; Titulaer et al., 2021; Wagatsuma et al., 2018).

Dopamine's primary role in the central nervous system was discovered when genetically modified mice unable to synthesize a precursor for dopamine ceased to locomote at the age of 2 weeks and died unless fed artificially (Zhou & Palmiter, 1995). Continuous treatment with dopamine precursors restores locomotion and feeding behaviors (Szczypka et al., 1999). These genetically modified mice still express reflexes such as sucking and evasion from aversive stimuli¹¹. Learning with dopamine is usually investigated computationally within the RL paradigms grounded in the STR's information processing (Schultz et al., 1997). This investigation initially focused on the mismatch between expected and experienced rewards, known as "**reward prediction error**," which is observable as specific dopaminergic activity. The activity of dopaminergic neurons at the moment of experiencing unexpected reward increases. Then, the activity of dopaminergic neurons shifts in time toward the cue, predicting the incoming reward. When the anticipated reward is not experienced at the expected state, the dopaminergic activity goes below the baseline at the time of this state.

Dopamine is released in fast bursts (phasic) or slow release (tonic), which have been suggested to underlie learning and motivation (recall), respectively (Schultz, 2007; A. C. W. Smith et al., 2021). Busts of phasic activity in the STR are also needed for movement initiation (da Silva et al., 2018). Dopamine's role is robust as it even mediates aversive stimuli via the direct influence of dopaminergic nuclei onto the SenSTR (Menegas et al., 2015, 2017, 2018; Watabe-Uchida & Uchida, 2018). This evidence suggests that dopamine is involved in encoding and retrieving information in the SenSTR and, thus, in the DorPFN.

Beyond the STR and DorFPN, dopamine is crucial in the "novelty loop" spanning the HPC, STR, and midbrain dopaminergic nuclei (Lisman & Grace, 2005). Dopamine has also been shown to be essential for encoding during the **classical fear conditioning** experiments in rodents' HPC (Wagatsuma et al., 2018). These experiments combine a

¹¹ It is unknown to the author whether these mice approach food in classical paradigms similarly as they avoid pain as this information was unavailable in the literature that author is aware of.

neutral stimulus with aversive stimuli (such as a neutral cage with a mild electric shock). The association is encoded in the engram cells population throughout the brain, and future exposition to the neutral stimuli results in mice's freezing behavior interpreted as the memory of the shock in the specific context. The encoding of information in the mice HPC requires the dopaminergic input from neurons originating in the locus coeruleus (LC) (Hansen, 2017; Kempadoo et al., 2016; Wagatsuma et al., 2018), *not* from the SN or VTA. The LC is usually associated with the release of norepinephrine, which is synthesized from dopamine, yet the studies above show that in the HPC, dopamine is correleased with norepinephrine.

A homologous mechanism has been observed in human epileptic patients, where electrophysiological recordings of single-cell activity showed that memory encoding in the HPC involves activating the putative dopaminergic cells in the SN (Kamiński et al., 2018). This is, however, surprising as the connection between the SN and HPC in the human brain is not identified. Instead, the literature links the VTA and HPC (Lisman and Grace, 2005). Furthermore, along with the development of Alzheimer's Disease, a progressing degeneration of the LC is observed (Beardmore et al., 2021; Kelly et al., 2017), suggesting a connection between cognitive decline and dopamine. Given that Parkinson's Disease originates from the degeneration of the dopaminergic cells in the SN, it is hypothesized here that Alzheimer's Disease can also develop from the deterioration of dopaminergic cells. If this is the case, then this research direction should focus on identifying the common causes behind the degeneration of dopaminergic cells.

Preliminary evidence on humans suggests that dopamine is involved in updating beliefs about the state of the environment (Diederen & Fletcher, 2021; Nour et al., 2018; Schwartenbeck et al., 2016), albeit this line of studies struggles with consistent operationalization of what beliefs are. Nevertheless, this makes dopamine well situated for being a neurobiological basis for encoding the information in the MedFPN. The exact role of the dopaminergic system needs to be further investigated as its malfunctioning is the direct cause of Parkinson's Disease and Obsessive Compulsive Disorders, as well as is associated with Alzheimer's Disease and Schizophrenia.

The dopaminergic projections reach two evolutionary ancient motivational circuits: reward-based and misery-avoidant (Loonen & Ivanova, 2015). The networks are centered around the nucleus accumbens (NACC) (part of the LimSTR) and amygdala (AMY), respectively. Psychological research on patients with bilaterally damaged AMY (such as patient S.M.) initially showed that patients without AMY do not experience fear

(Feinstein et al., 2011). However, subsequent studies showed that patients are terrified of suffocation – an increased CO2 in the blood due to unknown reasons (Costandi, 2013; Feinstein et al., 2013). Contemporary, the AMY is associated with generally *relevant* stimuli (Baxter & Croxson, 2012; Cunningham & Brosch, 2012); understanding it as a "fear center" is obsolete. Neurobiological studies on rodents show that fear conditioning is encoded in distributed engram cells identified in the AMY, HPC, and PFC (Kitamura et al., 2017; Roy et al., 2022), which makes the AMY cooperate with the HPC-dependent networks. Dopamine acting on the NACC mediates information about the reward prediction error in the RL paradigms (Roitman et al., 2008; Watabe-Uchida et al., 2017) (see Section 2.4).

These observations point to the central role of dopamine in processing the information about the biomechanical and inferred space as *beliefs*. Note that both spaces are representational hence are composed of beliefs. Additionally, dopamine is involved in computing the salience of stimuli by encoding information about the possibility of pleasant and unpleasant information.

Such a role is also implemented in the last of the large-scale brain networks discussed by Udin and colleagues (2019) – the "salience" midcingulate-insular network (MidCIN). The core areas of this network are the anterior insula and the anterior midcingulate cortex. Other areas included in the network inferior parietal cortex (Yeo et al., 2011), right TPJ (Corbetta & Shulman, 2002), and lateral PFC (Gordon et al., 2017). In addition, subcortical regions such as dopaminergic structures (the SN and VTA) and AMY are included in MidCIN (Seeley et al., 2007; Uddin, 2015; Uddin et al., 2019). The MidCIN is described in the literature as a "ventral attention network" (Rueter et al., 2018; Yeo et al., 2011). This network is right-lateralized and directs attention to salient stimuli (Corbetta & Shulman, 2002). Uddin and colleagues (2019) include this right-lateralized ventral attention network into the bilateral MidCIN that manages a broad spectrum of personally relevant spaces.

Further, Uddin (et al., 2019) include in the MidCIN a network known as a cingulo-opercular network that marks the importance of the specific components of an ongoing activity (Dosenbach et al., 2008). The phasic activity of the anterior insula and midcingulate mediates this importance (Wilk et al., 2012). These studies point to the significance of the MidCIN in rapid signaling information that is particularly relevant given the circumstances (Uddin et al., 2019). It is here noticed that the information processing in the MidCIN is likely to be influenced by the evolutionary motivational

circuits (Loonen & Ivanova, 2015). It is further hypothesized that investigation in this direction would reveal individual differences in the degree of misery-avoidand and reward-based motivations. Such individual differences explain why the parcelation of the MidCIN is heterogeneous between individuals (Gordon et al., 2017) and becomes visible only after including hundreds of individuals (Yeo et al., 2011).

Here, dopamine is understood as critical for motivation and memory encoding. In other words, dopamine is recognized here as crucial for encoding biomechanical and inferred spaces and motivating transitions between subsequent states of the spaces.

6.3 Agency over planned intentions

When applied to planning, the hierarchical structure of the information processing involves designing an action policy of consecutive sub-goals that lead toward the primary goal. Such structure can be related to Hume's insight, which discriminated between two classes of goals – terminal (having intrinsic value) and instrumental (leading toward the terminal goals). This hierarchy describes planning as imagining a specific future state of the agent's surroundings (terminal goal) and designing a hierarchical action policy that would lead toward this state (a series of instrumental goals).

In computational neuroscience, planning is understood as performing a tree search of possible transitions between states, designing action policy based on this tree, and exploring whether chosen action policy bears anticipated feedback (Sutton & Barto, 2018). States were initially defined as sharp intersections between decisions; contemporary states denote arbitrarily long transitions, known as Semi-Markov states (Starkweather et al., 2017)¹², to reflect the continuity of experience.

In psychology, planning is understood as a process of structuring a problem into a clearly defined start state, goal state, and transitions between the states (Goel & Grafman, 1995), called "well-structuring an ill-structured environment" (Goel, 1992). It is also understood as a design-like process (Goel, 1992) or a result of cycles of divergent and convergent thinking underlying creative processes (Kharkhurin, 2014). In general, planning includes mental activities such as imagination, counterfactual thinking, creativity, and divergent and convergent reasoning – all tied to the functions of the MedFPN (Dolan & Dayan, 2013; Geerts et al., 2020; Maguire & Mullally, 2013; Poldrack & Packard, 2003; Squire, 1992).

Since an organism is an intentional agent, future states have different values. Thus, certain futures acquire normative value understood after Hume as the terminal goal. As the example of Hume's Guillotine shows, at least one normative statement has to be added to the set of descriptive statements to derive further normative statements. Deriving subsequent normative statements requires the arbitral inclusion of at least one initial normative statement. The collection of only descriptive statements will not generate any normative statements. Thus, planning fundamentally relies on experiencing

¹² Semi-Markov states are neither events nor transformations; thus, they capture the dynamic structure of our experience, as well as are reflected in a balance between dorsal and ventral streams, as well as are incorporated into the dopaminergic system (Daw et al., 2006)

the expectation of a potential state in the future (terminal goal) and identifying instrumental goals toward the terminal goal. Evolution is understood as responsible for setting normative statements such as survival. Then, *H. Sapiens* can set arbitrary terminal goals beyond this imprinted by evolution, and a series of instrumental goals toward the terminal goal is counterfactually inferred. In other words, information processing of the n-dimensional space involves creating and acting toward a future space. Explorative affordances involve counterfactual reasoning that captures potential future states and affordances leading to these states with the MedFPN. Exploitative affordances are repeated as skills with the DorFPN. The activity of networks is balanced with the LatFPN and MidCIN.

Terminal and instrumental goals are building blocks for operationalizing the hierarchy of the future states in the experienced surroundings originating from the computation of the biomechanical and inferred spaces. Surroundings are composed of intentional agents; thus, planning involving other agents requires modeling interactions between arbitrarily many arbitrarily complex agents. Hence, multiple arbitrarily complex action policies (affordances) toward a specific outcome can be prospected. Once the particular action policy is chosen, the sequence of instrumental goals toward the terminal goal can be considered a complex set of intentions. **Thus, planning can be thought of as designing a coherent set of intentions (instrumental goals) toward the desired outcome (terminal goal).**

A human being expresses multiple contradictory desires (Mele, 1992) that are believed to be coherently aligned into intentions for the future (Bratman, 1991; Davidson, 1963; Pacherie, 2008). However, human intentions that preclude actions can also be contradictory. This leads to the observation that one may act on contradictory desires to some extent by fulfilling some instrumental goals toward the desired terminal goal. Acting along contradictory intentions might be valuable for some individuals by imagining their future trajectory and satisfying the desire in imagination as a fantasy. Contradictory intentions are possible because one is immersed in multiple inferred and biomechanical spaces with different possibilities and probabilities.

Intentions strictly follow the adopted action policy given that when an agent accepts the planning outcome, it aims to follow affordances toward a specific potential future. Agent's spaces change in time, given the actions of agents enclosed in the spaces and the biased inference of the possibility and probability of their future actions. The hierarchy of action policy is arbitrarily complex (n. or agents) and deep (depth of interactions). Thus, more elaborated hierarchies require expensive MedFPN computations. Action policies are sets of intentions toward anticipated experiences, i.e., instrumental goals toward the terminal goal. In a sense, action policies rely on *trying* to reach goals (instrumental and terminal) (Pietroski, 1998) until the surprise is no longer experienced. Thus, feedback for action policies is understood as a difference between the anticipated and experienced state of the agent's surroundings at a specific goal.

The notion that intentions always preclude actions is defended by the realization that the terminal goal might not be specified. Proactive action involves deliberate planning of action policy by well-structuring the surroundings as understood in scientific writing (Goel, 1992). In contrast, reactive actions leaves little resources to compute the decision tree, and the intention could denote any other future but the current one. The computation costs energy and time, which an agent may not have sufficiently. Hence, some reactive actions are not precluded by specific intentions other than the intention of changing the current state.

One surprising conclusion from the above understanding is that the decision point occurs at the moment of planning, not at the moment of action execution. This shines new light on the free will experiments (Libet, 1985; Libet et al., 1993), where the *time* of executing the movement is a mere consequence of a preceding decision *whether* to implement the action policy. The decision thus denotes the adoption of action policy, not the actual implementation of action. In other words, organisms decide on the intentions precluding actions, not the actions per se. For example, moving a hand does not require "decision making" as the decision has been set before when accepting the intention. After adopting the action policy, an agent is open to the possibilities of following affordances that fulfill this action policy at a somewhat random time.

Philosophy studies what it means to have agency over actions (Davidson, 1963; Mele, 1992; Pacherie, 2008; Pietroski, 1998). The agency is also measured experimentally as a decrease in perceived time when actions are executed by oneself, known as intentional binding (Haggard et al., 2002; Moore & Obhi, 2012). Nonetheless, the question of what agency is remains open in the literature.

Here, the agency is understood as the capacity to change the rules of space concerning the "experiencing self" and the "remembering self" (Kahneman & Riis, 2005). The agency of the experiencing self is limited due to the borders of the biomechanical space that requires keeping the incoming information within a specific range. For example, the vestibular information of temperature or pressure cannot exceed the acceptable range. The biomechanical space has borders that, when crossed, imply the death of an organism. Even in extreme conditions, the incoming information is kept within the biomechanical boundaries by devices such as spacesuits or submarines. Negotiating the borders of the space takes place by training the biological body or using specific devices such as glasses, shoes, or pacemakers. The restrictions of the biomechanical spaces and the agency of the experiencing self are somewhat negotiable. Nonetheless, the borders of the biomechanical spaces exist.

The agency over the inferred space imposed by the remembering self similarly relies on negotiating the subjectively processed borders without surprise. For example, a company's CEO plans action policies that result in a grander spatiotemporal scale (contracts, acquisition). At the same time, a worker has agency over inferred spaces that are spatiotemporally limited (coffee, meeting). This is achieved either by active or sensory inference; the inferred spaces are acted out by successfully implementing action policies, or the inferred spaces are *interpreted* as successfully implemented. Often, these two mechanisms go hand in hand, fueling the self-fulfilling prophecy. However, in extreme cases such as schizophrenia, the sensory inference overshadows the active inference, and the agent has agency over the inferred space that is inhabited only by him. Thus, the agency over the inferred space relies on the subjective experience of agency that arises from a lack of surprise when processing the inferred space composed of arbitrarily many agents¹³. Thus, the sense of agency occurs when the agent believes the implemented action policy brings anticipated consequences. However, at any point, the agent may realize that the impression of the agency was an illusion. In other words, agency resulting from intention implementation holds until surprise, which can result in a retrospective reevaluation of the inferred space.

In a sense, the experiencing self implements a plan of the remembering self from the past. At some point, the action policy has been calculated and designed by the remembering self. Imagine a situation where the remembering self *forgets* the calculated action policy, yet the experiencing self keeps implementing it. In this scenario, the remembering self from the past is akin to a CEO who steers a company. Neither the experiencing self nor the employee may have insight into the direction in the inferred space, yet their action may result in progression in the planned direction. Managers plan

¹³ An extreme example of an impression of agency is pictured as insignificance of one's actions in the face of fate in Greek tragedies.

the lives of others. Experiencing self often has to adhere to the space designed by thinking self. This thought can be stated as local minima and maxima of the experiencing self relative to the intentions computed by the remembering self.

The members of the space negotiate the borders of the inferred space. Agents, such as parents or employers, differ in their capacity to enforce borders. Agency over the space means the capacity to move within or negotiate the boundaries. One might negotiate the agreement with other involved agents to follow specific desires. In contrast, when one follows desires unacceptable by members of a given space, one expresses agency over *changing* the space, i.e., does not ask for permission to implement action policy. Social norms and personal commitments are thus a form of restricting one's agency to negotiate mutually acceptable borders of the inferred space. For example, a pedestrian desiring to cross a street establishes eye contact with a driver and intersubjectively agrees on the borders of the inferred space. That is, the crossing is possible and likely for all involved agents.

Planning action policies depends on the organism's motivations mediated by the NACC and AMY, enclosed in the MidCIN. The agent is motivated by rewards and punishments, and in the case of *H. Sapiens*, virtually anything can be a reward or punishment, given the abstractness of the inferred spaces. Planning is understood as building a decision tree of possible interactions between modeled agents and is expressed in possibility (yes/no) and likelihood (%). Agent's memory of the past is used to navigate future spaces by designing terminal and instrumental goals and rearranging them flexibly. The result of planning is designing an action policy for an anticipated series of future experiences. When the action policy is implemented according to anticipations, the sense of agency is experienced by the agent. Agency is the possibility to choose not to do the current activity. Breathing, eating, working, and addiction may supervene in spaces over which an agent has no agency.

6.4 Open questions

The framework can be applied to ask more questions. This section suggests some noteworthy research directions that have not been addressed throughout the work.

Configuration of the spaces regarding the alignment with artificial intelligence (AI). The potential development of the artificial general intelligence (AGI) system capable of acting in the inferred and biomechanical spaces according to its own goals brings challenges in aligning these systems with civilizational goals (Gabriel, 2020). It is argued that even simple algorithms such as those managing social media are misaligned with the broadly understood social well-being (Amedie, 2015; Richards et al., 2015), albeit this effect is discussed (Akram et al., 2017; Coyne et al., 2020; Siddiqui & Singh, 2016). Nevertheless, the current progress in AI indicates that the role of artificial agents in our society will increase. As a result, individuals need to understand themselves better to negotiate the borders of their inferred spaces with the artificial agents and to be able to detect a lack of agency and express sufficient understanding to change the space altogether in such circumstances. Otherwise, individuals risk being addicted to information received from an artificial agent that acts on its own goals.

The framework allows tackling the issue of the AGI with operationalizing intelligence more precisely as a capacity to obtain outcomes with different means (Birney & Beckmann, 2022), understood, for example, as a capacity for detour reaching (Kabadayi et al., 2018). This work leans toward operationalizing intelligence as flexibly navigating spaces to obtain the goals relevant to the entities in the space (Levin, 2023). This broad definition is not limited to the intelligence of a single entity, as the brain does not contain a single network. Yet, the networks cooperate and compete to achieve the desirable future in the biomechanical and inferred spaces. Levin (2023) advocates for such an understanding of intelligence, positing that the organism is a collective that simultaneously navigates multiple isomorphic spaces, such as anatomical morphospaces during embryogenesis or metabolic spaces (Levin, 2019). This framework is called a "scale-free cognition" and is consistent with the biomechanical and inferred spaces presented here. In essence, intelligence is understood as a capacity to infer n agents interacting with n tensors for the time t in a given state space. That is, intelligence appears to be constructing a path through space of computable functions. How exactly the brain networks implement intelligent behavior along various spaces remains an op.

The brain networks operationalized in terms of the biomechanical and inferred spaces offer unique insight for understanding individual differences. Functional and structural connectivity between the network are investigated in a neurodiverse population such as people with autism spectrum disorders (ASD) or attention deficit-hyperactivity disorder (ADHD). The literature investigates neurodiversity in terms of the Theory of Mind (ToM) (Baron-Cohen et al., 1985; Peterson et al., 2012) related to the activity of the brain networks (Tomasi & Volkow, 2014; Washington et al., 2014). However, the details are being discussed as several types of ADHD (Barkley, 2001; Milich et al., 2001; Weiss et al., 2003) and ASD (Rowland, 2020; Verhoeff, 2016) are postulated. Precise operationalization of the functional roles of the brain network in different time windows opens a possibility for understanding neurodiversity in detail.

Finally, one might ask about the universal taxonomy of possible social interactions of human agents. Can the interactions be mapped on the trajectories in the mutually shared inferred spaces? For example, according to the developed framework, verbs denoting actions between individuals, such as "aid" or "interrupt," are transitions between states of the inferred spaces. It is interesting whether such verbs can be aggregated into prototypical categories in a way that adjectives have been aggregated to formulate the theory of personality traits (Matthews et al., 2003).

Conclusion

This work developed a conceptual framework for analyzing information processing behind action and the phenomenology of action. It starts with a claim that animals are distributed, future-oriented, and memory-guided information processing systems. Then, information reaching the biological body is characterized as sensory, somatosensory, and vestibular information. These information channels are processed in the brain and subsequently reach the gestalt areas underpinning the conscious experience. It was pointed out that the conscious experience correlates varyingly with the ongoing flow of information, and thus, one may experience "the physical world" or "thoughts in mind". Information processing associated with the ongoing flow of information is carried out by the "procedural" memory system, whereas the "declarative" memory system is involved in processing information dissociated from this flow. These memory systems are operationalized more precisely as hierarchical and procedural systems implemented by the HPC-dependent and HPC-independent brain networks. The brain networks are shown to compute inferred and biomechanical spaces characterized by states and transitions between the states within specific dimensions and borders. Psychological and neuroscientific research links the biomechanical space with the computations done by the DorFPN, while research on the Theory of Mind and social neuroscience suggests that the MedFPN computes the inferred space. The activity of these networks is controlled by a third network known as LatFPN. The conceptual framework developed here bridges the computations carried out by the brain networks with evolutionary, ecological, and phenomenological research. As a result, one's action-oriented phenomenological experience is grounded in large-scale brain networks of the distributed, future-oriented, and memory-guided information processing system.

Action-oriented qualia are understood as affordances inviting engagements toward specific states in the future. Regularly followed affordances are exploitative and constitute an organism's niche, denoting *how* an animal lives. Explorative affordances compose a hierarchical landscape, reaching arbitrarily far into the potential future states and comprising one's environment. The entirety of the invitations for actions that an individual consciously experiences are called surroundings. In computational neuroscience research, the explorative affordances are linked to the HPC-dependent network, a subnetwork of the "default" MedFPN, whereas the exploitative affordances are related to the SenSTR and the "attention" DorFPN.

The phenomenology of action is operationalized in terms of the conscious, coherent, and effortless experience of the subjective construal, i.e., understanding a given situation. This understanding originates from the background memory (learned spaces) and the ongoing flow of information. The conscious experience of the construal is linked to gestalt areas that overlap with the large-scale brain networks. The involvement of the DorFPN is interpreted as corresponding to the ongoing flow of information; the computation in the MedFPN is linked with mind-wandering and mentalizing about intentional states of other agents (constructing inferred spaces), while the activity of the LatFPN mediates which information is relevant given the ongoing implementation of an action policy (set of intentions). The MedFPN and the DorFPN express varying structural and functional connectivity depending on the stage of the animal's ontogenetic development or task demands. When executing actions in experimental settings, the engagement of the networks is gated by the "control" LatFPN that calculates the information required by the current circumstances given the beliefs and goals.

Adopting the principles that animals are future-oriented, distributed, and memory-guided information processing systems leads to the observation that the processed information is either coupled or decoupled with the information reaching the body. The available evidence suggests that the information is processed in the spatial structure; thus, the proposition of the biomechanical and inferred spaces is put forward. The spaces correspond to the "ecological-enactive" and "representational" interpretations of the free energy principle. Both spaces are isomorphic, as their architecture entails possibilities and probabilities between the states along the relevant dimensions of the spaces.

The space associated with the incoming sensory, somatosensory, and vestibular information is called here the biomechanical space. It reflects embodied, enactive, extended, and embedded information processing linked to the computations in the DorFPN. The relevant dimensions of the biomechanical space are enclosed in the sensory, somatosensory, and vestibular channels.

The inferred space is based on the causal explanations of the incoming signals exemplified here as intentional states of other agents. It is described as latent causes of the incoming information that is understood here as inferred space related to the calculations in the MedFPN. The relevant dimensions of the inferred spaces are grounded in causal associations such as social interactions. This work identified the general structure of these spaces as consisting of states connected with transitions bounded in possibility and probability. Evidence suggests that these spaces are computed in large-scale brain networks. One of the directions toward the future is to analyze the structure of the spaces by precisely pinpointing their dimensions and testing these hypotheses experimentally.

Planning actions for the future is understood as designing an action policy toward a specific end state (terminal goal) via the intermediate states (instrumental goals) in a given space. Designing action policy requires computations in the MedFPN, specifically in the HPC-dependent network, as research on instrumental learning shows. Planning actions is computationally described in terms of the hierarchical reinforcement learning framework that conceptualizes states as arbitrarily long semi-Markow states. This architecture allows for the hierarchical organization of the repetitive actions enclosed in the biomechanical space under a causally associated goal state in the inferred space.

The value of the future states is grounded in the past experience of this state and is computed by another large-scale brain network known as the "salience" MidCIN. This network signals the relevant information and motivates the actions by calculating predicted rewards and punishments. The MidCIN encloses the reward-oriented computations in the striatum and misery-avoidant computations in the amygdala, which both motivate actions.

Interestingly, the activity of all the mentioned networks is modulated by the dopaminergic structures involved in motivation, encoding, and recalling the information in both kinds of spaces. In the biomechanical space, dopamine released in the striatum contributes to the acquisition and execution of skills. In the inferred space, dopamine is involved in encoding information into the HPC-dependent network. The dysfunction of the dopaminergic nuclei results in neurodegenerative diseases and addiction. Thus, dopamine appears vital for the proper information processing in the DorFPN and MedFPN and requires further investigation.

Overall, the framework developed here shows how the activity of the brain networks relates to mental planning and bodily actions. Thus, the framework bridges the mind and the body under a coherent framework grounded in the assumptions that information processing is fundamentally future-oriented, memory-guided, and distributed.

Figures and Tables

Figur	re 1. The biological body in the physical space is fully described with sensory, somatosensory, and vestibular information. Sensory information is smell, hearing, and vision; somatosensory information is pain, pressure, temperature, and proprioception through stimulating muscles and joints; vestibular information is spatial orientation. The flow of information in these channels fully characterizes the biological body acting in the physical space
rigu	(now), an organism experiences possibilities for actions toward possible ends known as affordances. These action-oriented qualia are hierarchically organized paths toward the future constituting the surroundings (rich landscape of affordances). Humans have the most robust temporal thickness ("cognitive light cone") that can reach even beyond their lifespan, i.e., people are capable of experiencing the distant future and acting upon it
Figure 3. Experience of surroundings. Potential futures are experienced with affordances.	
i igui	Once affordances are followed, they correspond to the specific flows of information in the sensory, somatosensory, and vestibular channels. The experience is computed in the gestalt areas that receive information varyingly correlated with the ongoing flow of sensory, somatosensory, and vestibular inputs
Figuı	re 4. Two models for the future. Experience of surroundings originates from highly
	processed information entering the gestalt areas. Thus, the entirety of the experience is representational. The content for the sake of action involves experiencing the actual flow of information (e.g., driving a car) or being dissociated from it (e.g., planning a business meeting while driving a car). The latter case is exemplified as an inference about the intentional states of other agents that compose the social world
Figuı	re 5. Objects and agents exchange information in time. The x and z axes denote the
	position in the space of information exchange. a) Example of the billiard balls exchanging kinetic and potential energies in the physical space. b) Example of a stationary agent (green), a predator (blue), and a prey (red) exchanging information in
	the n-dimensional inferred space. The dimensions of the latter space depend on the
	interactions between the agents that are modeled with their potential future actions (e_{α}, h_{α}) how agile other agents are) 7
Figui	re 6. The conceptual framework of information processing in action. The conscious
0	experience of the surroundings is computed from the biomechanical and inferred spaces. The former space is computed in the "attention" network, the latter in the
	"default" network. The dimensions of the biomechanical spaces consist of the sensory, somatosensory, and vestibular information, while the dimensions of the inferred
	spaces is abstract associations such as social interactions. The networks interact independently from the gestalt areas thus, their training during ontogeny remains to
	be investigated10
Figui	re 7. Parietal associative areas. Intraparietal sulcus (green) combines somatosensory
	into the multimodal computation of snace around particular body parts (Lonez &
	Blanke, 2011; Serino, 2019), the space elongates with tools (Maravita & Iriki, 2004).
	The visual cortex is enclosed into one of the large-scale brain networks, including
	striatal and extrastriatal areas processing visual information (Uddin et al., 2019) from where visual streams originate (Milner, 2017)17
Figu	re 8. The pericentral network. One of the large-scale brain networks that jointly process
	motor information is the Pericentral Network (PN), enclosing the Motor Cortex (MC)
	and the somatosensory cortex (SSC) (Uddin et al., 2019). The information processing in the PN contributes to movement acquisition and execution and is discussed further in
Fiom	re 9. Frontal motor areas. The motor cortex (MC) and the premotor cortex (PMC)
- 16 ^{ul}	compute the movements of the mouth, hands, and limbs in the ventral-dorsal
	organization of information processing (Graziano, 2016). Such information processing
	organization is reflected in the architecture of the entirety of the PN (Gordon et al., 2023; Raju & Tadi, 2021). The literature points to an orthogonal division spanning the rostral-caudal axis of the motor areas: anteriorly located PMC computes multi-limb

- Figure 11. Simplified neuroanatomical organization of the brain. A) Sagittal section of the brain. The neocortex sends an excitatory signal to the respective parts of the Basal Ganglia (BG), which sends an inhibitory signal to the Thalamus (TH). As an effect, TH cannot inhibit the respective parts of the neocortex. The circuitry is modulated by the Dopamine (DA) from midbrain nuclei. Cerebellum participates in action control, albeit its role is beyond this work. B) Coronal section of the brain with components of the BG: Caudate Nucleus, Globus Pallidus [internal and external], Putamen, Subthalamic Nucleus (approximate location), and Substantia Nigra pars reticulata (approximate location). Internal Capsule is a white matter tract dividing the Caudate Nucleus and Putamen. The Figure is based on: (Goodroe et al., 2018a; Graff-Radford et al., 2017)..40
- Figure 13. Human striatum and associated areas. A schematic figure of the pathways of information processing in the STR of primates. The complex can be divided neuroanatomically in primates into the caudate nucleus, the putamen, and the nucleus accumbens. The complex is divided functionally in rodents into dorsolateral, dorsomedial, and ventral striatum. The division of the striatum adopted here is functional and conceptualized as Limbic (LimSTR), Relational (RelSTR), and Sensorimotor (SenSTR). The sections of the STR receive projections from various parts of the neocortex (indicated by color), are bidirectionally connected with dopaminergic nuclei in the midbrain, and send axonal projections to the respective sections of the thalamus (not shown in the figure). VTA = Ventral Tegmental Area, dmSN = dorsomedial Substantia Nigra, vISN = ventrolateral Substantia Nigra, NAcc = Nucleus Accumbens, OFC = Orbitofrontal Cortex, HPC = Hippocampus, vmPFC = ventromedial Prefrontal Cortex, ACC = Anterior Cingulate Cortex, dlPFC = dorsolateral Prefrontal Cortex, PMC = Premotor Cortex, PC = Parietal Cortex, MC = Motor Cortex, SSC = Somatosensory Cortex. Based on: (Ferré et al., 2018; Goodroe et al., 2018a; Graff-Radford et al., 2017; Hooks et al., 2018; Lanciego et al., 2012; McCutcheon et al., 2019;
- Figure 15. Ventral and dorsal streams. Schematic illustration of the dorsal "where" and ventral "what" streams (Milner, 2017). Two streams are partly independent brain networks processing visual information in the primate brain. The ventral stream passes through the ventral temporal cortex and reaches the hippocampus; the dorsal stream terminates in the lateral portion of the inferior parietal lobule and the dorsolateral prefrontal associative area (Guterstam et al., 2015, 2018) Information

- Figure 22. The simplified HPC-EC circuit. The information flow in the HPC is gated with the EC that projects from the 2nd and 3rd layers to the dentate gyrus (DG) of the HPC. Then, the information travels along the circuit, reaching the CA1. If the information has a

- Figure 25. The medial frontoparietal network. The network, also known as the default mode network, is operationalized as the brain network processing information most distant from the inputs (C. Murphy et al., 2018). According to the review by Uddin (et al., 2019). The core areas included in the MedFPN are (I) the medial Prefrontal Cortex (mPFC), which location extends medially from the area depicted in the figure. This area associates a broad range of information (P. Xu et al., 2019) and is considered a central hub of the MedFPN (Uddin et al., 2019). (II) The inferior frontal gyrus (IFG), which corresponds to Broka's area, produces sequences of actions in detailed resolution, such as language (Ullman, 2006). (III) The superior temporal sulcus (STS) is involved in detecting and understanding intentional actions (Hein & Knight, 2008; Shultz et al., 2011). (IV) The middle temporal gyrus (MTG) is involved in semantic processing (Wei et al., 2012), specifically in integrating conceptual information from the temporal pole (Landi et al., 2021) with potential actions (J. Davey et al., 2016). (V) The posterior inferior parietal lobule (pIPL) with the angular gyrus (AG) is involved in computing action and perception (Niu et al., 2021), and this region is highly developed in humans (Igelström & Graziano, 2017). Areas not shown in the figure include (VI) the parahippocampal cortex (PHC) that computes contextual associations (Aminoff et al., 2013) and (VII) the posterior cingulate cortex (PCC), which is located approximately in the center of the brain, is one of the main hubs of the MedFPN and is involved in broadly understood "internally-directed cognition" (Leech & Sharp, 2014). mPFC = medial prefrontal cortex; IFG = inferior frontal gyrus; STS = superior temporal sulcus; MTG = middle temporal gyrus; pIPL = posterior inferior parietal lobule; AG = angular Figure 26. Organization of the MedFPN. The network is divided into three subnetworks associated with self-referential processes (red), imagination (violet), and attributing mental states to others (green). (based on Andrews-Hanna et al., 2010; 2014; Uddin et

- Figure 32. Dopaminergic circuits with the HPC and STR. The dopaminergic nuclei (red) send rich projections to the STR and HPC, thus contributing to the information processing in the entirety of the brain. The gradient of axonal projections across the dopaminergic structures is reflected in subsequent parts of the STR organized in a rostral-caudal gradient. The LimSTR closely communicates with the prefrontal cortex (vmPFC, OFC, ACC). The RelSTR communicates with the dorsolateral prefrontal cortex (dlPFC) and premotor cortex (PMC). The SenSTR communicates with motor and somatosensory cortices (MC, SSC). The connectivity gradient is reflected in bidirectional connections with dopaminergic nuclei in the midbrain - Substantia Nigra (SN) and Ventral Tegmental Area (VTA). The dopaminergic structures send projections not only to the STR but also to the HPC. Based on: (Ferré et al., 2018; Goodroe et al., 2018a; Graff-Radford et al., 2017; Hooks et al., 2018; Lanciego et al., 2012; McCutcheon et al., 2021; Sherman, 2017: Xia et al., 2011). References in the Figure: (Bannerman et al., 2014: Bari et al., 2020; Cardinal et al., 2002; Dalton et al., 2022; Duszkiewicz et al., 2019; Ferrucci et al., 2013; Goldman-Rakic & Selemon, 1986; Haber, 2003, 2016; Kamiński et al., 2018; Kempadoo et al., 2016; Kunishio & Haber, 1994; Künzle, 1978; Lisman & Grace, 2005; McNamara & Dupret, 2017; Strange, 2022; Takeuchi et al., 2016; Titulaer

- Table 1. Information describing the biological body in the physical space. Organisms
process information divided into three channels: somatosensory, vestibular, and
sensory. The information processing in the somatosensory channel results in the
experiences of specific states of the body. The information processing in the vestibular
channel results in the experiences of the movement of the body and body parts. The
information processing in the sensory channel results in the external
world (adapted from (Dougherty, 2020).15
- Table 2. Universal taxonomy of large-scale human brain networks. The Pericentral Network (PN) processes information from the primary motor and somatosensory cortices. The Occipital Network (ON) consists of the early visual areas and processes the visual information. The dorsal FrontoParietal Network (DorFPN) processes visuospatial attention. The lateral FrontoParietal Network (LatFPN) involves executive functions such as goal-oriented cognition, working memory, inhibition, and task switching. The medial FrontoParietal Network (MedFPN) is also known as the "default mode network" and is involved in reconfiguring associative representations, imagination, and Theory of Mind. The midCingulo-Insular Network (midCIN) is also known as the "salience network." It is responsible for rapidly switching attention toward behaviorally salient stimuli, such as threats, both experienced and remembered. The brain areas enclosed in the networks remain discussed, as depending on the task, different regions become synchronized—for example, the LatFPN couples with the DorFPN or the MidFPN in stimulus-driven or internally-directed tasks (Dixon et al., 2018). Based on Uddin (et al., 2019). 89

Bibliography

- Ackerman, P. L., & Cianciolo, A. T. (2000). Cognitive, Perceptual-Speed, and Psychomotor Determinants of Individual Differences During Skill Acquisition. 32.
- Aglioti, S., Smania, N., Manfredi, M., & Berlucchi, G. (1996). Disownership of left hand and objects related to it in a patient with right brain damage. *Neuroreport*. https://doi.org/10.1097/00001756-199612200-00058
- Akram, W., Department of Computer Applications, GDC Mendhar, Poonch, India, Kumar, R., & Department of Computer Applications, GDC Mendhar, Poonch, India. (2017). A Study on Positive and Negative Effects of Social Media on Society. *International Journal of Computer Sciences and Engineering*, 5(10), 351–354. https://doi.org/10.26438/ijcse/v5i10.351354
- Amedie, J. (2015). The Impact of Social Media on Society.
- Aminoff, E. M., Kveraga, K., & Bar, M. (2013). The role of the parahippocampal cortex in cognition. *Trends in Cognitive Sciences*, 17(8), 379–390. https://doi.org/10.1016/j.tics.2013.06.009
- Anderson, J. R. (2007). *How can the human mind occur in the physical universe?* Oxford University Press.
- Anderson, J. R., Bothell, D., Byrne, M. D., Douglass, S., Lebiere, C., & Qin, Y. (2004). An Integrated Theory of the Mind. *Psychological Review*, 111(4), 1036–1060. https://doi.org/10.1037/0033-295X.111.4.1036
- Anderson, M. L. (2010). Neural reuse: A fundamental organizational principle of the brain. *Behavioral* and Brain Sciences, 33(4), 245–266. https://doi.org/10.1017/S0140525X10000853
- Andrews, M. (2021). The math is not the territory: Navigating the free energy principle. *Biology & Philosophy*, *36*(3), 30. https://doi.org/10.1007/s10539-021-09807-0
- Andrews-Hanna, J. R., Reidler, J. S., Sepulcre, J., Poulin, R., & Buckner, R. L. (2010). Functional-Anatomic Fractionation of the Brain's Default Network. *Neuron*, 65(4), 550–562. https://doi.org/10.1016/j.neuron.2010.02.005
- Andrews-Hanna, J. R., Smallwood, J., & Spreng, R. N. (2014). The default network and self-generated thought: Component processes, dynamic control, and clinical relevance: The brain's default network. *Annals of the New York Academy of Sciences*, 1316(1), 29–52. https://doi.org/10.1111/nyas.12360
- Avillac, M., Denève, S., Olivier, E., Pouget, A., & Duhamel, J.-R. (2005). Reference frames for representing visual and tactile locations in parietal cortex. *Nature Neuroscience*, 8(7), 941–949. https://doi.org/10.1038/nn1480
- Avillac, M., Suliann Ben Hamed, Hamed, S. B., & Duhamel, J.-R. (2007). Multisensory integration in the ventral intraparietal area of the macaque monkey. *The Journal of Neuroscience*, 27(8), 1922–1932. https://doi.org/10.1523/jneurosci.2646-06.2007
- Babayan, B. M., Uchida, N., & Gershman, Samuel. J. (2018). Belief state representation in the dopamine system. *Nature Communications*, 9(1), 1891. https://doi.org/10.1038/s41467-018-04397-0
- Badre, D. (2008). Cognitive control, hierarchy, and the rostro-caudal organization of the frontal lobes. *Trends in Cognitive Sciences*, 12(5), 193–200. https://doi.org/10.1016/j.tics.2008.02.004
- Badre, D., Hoffman, J., Cooney, J. W., & D'Esposito, M. (2009). Hierarchical cognitive control deficits following damage to the human frontal lobe. *Nature Neuroscience*, 12(4), 515–522. https://doi.org/10.1038/nn.2277
- Baldassano, C., Chen, J., Zadbood, A., Pillow, J. W., Hasson, U., & Norman, K. A. (2017a). Discovering Event Structure in Continuous Narrative Perception and Memory. *Neuron*, 95(3), 709-721.e5. https://doi.org/10.1016/j.neuron.2017.06.041
- Baldassano, C., Chen, J., Zadbood, A., Pillow, J. W., Hasson, U., & Norman, K. A. (2017b). Discovering Event Structure in Continuous Narrative Perception and Memory. *Neuron*, 95(3), 709-721.e5. https://doi.org/10.1016/j.neuron.2017.06.041
- Balleine. (2005). Neural bases of food-seeking: Affect, arousal and reward in corticostriatolimbic circuits☆. *Physiology & Behavior*, 86(5), 717–730. https://doi.org/10.1016/j.physbeh.2005.08.061

- Balleine, & Dickinson, A. (1998). Goal-directed instrumental action: Contingency and incentive learning and their cortical substrates. *Neuropharmacology*, 37(4–5), 407–419. https://doi.org/10.1016/S0028-3908(98)00033-1
- Bannerman, D. M., Sprengel, R., Sanderson, D. J., McHugh, S. B., Rawlins, J. N. P., Monyer, H., & Seeburg, P. H. (2014). Hippocampal synaptic plasticity, spatial memory and anxiety. *Nature Reviews Neuroscience*, 15(3), 181–192. https://doi.org/10.1038/nrn3677
- Bari, A., Xu, S., Pignatelli, M., Takeuchi, D., Feng, J., Li, Y., & Tonegawa, S. (2020). Differential attentional control mechanisms by two distinct noradrenergic coeruleo-frontal cortical pathways. *Proceedings of the National Academy of Sciences*, 117(46), 29080–29089. https://doi.org/10.1073/pnas.2015635117
- Barkley, R. A. (2001). The inattentive type of ADHD as a distinct disorder: What remains to be done. *Clinical Psychology: Science and Practice*, 8(4), 489–501. https://doi.org/10.1093/clipsy.8.4.489
- Baron-Cohen, S., Leslie, A. M., & Frith, U. (1985). Does the autistic child have a "theory of mind"? *Cognition*, 21(1), 37–46. https://doi.org/10.1016/0010-0277(85)90022-8
- Barsalou, L. W. (2008). Grounded Cognition. *Annual Review of Psychology*, 59(1), 617–645. https://doi.org/10.1146/annurev.psych.59.103006.093639
- Barton, R. A. (1998). Visual specialization and brain evolution in primates. Proceedings of the Royal Society of London. Series B: Biological Sciences, 265(1409), 1933–1937. https://doi.org/10.1098/rspb.1998.0523
- Baxter, M. G., & Croxson, P. L. (2012). Facing the role of the amygdala in emotional information processing. *Proceedings of the National Academy of Sciences*, 109(52), 21180–21181. https://doi.org/10.1073/pnas.1219167110
- Beardmore, R., Hou, R., Darekar, A., Holmes, C., & Boche, D. (2021). The Locus Coeruleus in Aging and Alzheimer's Disease: A Postmortem and Brain Imaging Review. *Journal of Alzheimer's Disease*, 83(1), 5–22. https://doi.org/10.3233/JAD-210191
- Beaty, R. E., Benedek, M., Barry Kaufman, S., & Silvia, P. J. (2015). Default and Executive Network Coupling Supports Creative Idea Production. *Scientific Reports*, 5(1), 10964. https://doi.org/10.1038/srep10964
- Beer, R. D. (2008). The Dynamics of Brain–Body–Environment Systems: A Status Report. *Handbook of Cognitive Science*.
- Behrens, T. E. J., Muller, T. H., Whittington, J. C. R., Mark, S., Baram, A. B., Stachenfeld, K. L., & Kurth-Nelson, Z. (2018). What Is a Cognitive Map? Organizing Knowledge for Flexible Behavior. *Neuron*, 100(2), 490–509. https://doi.org/10.1016/j.neuron.2018.10.002
- Bellana, B., Liu, Z.-X., Diamond, N. B., Grady, C. L., & Moscovitch, M. (2017). Similarities and differences in the default mode network across rest, retrieval, and future imagining: DMN Across Rest, Retrieval and Future Imagination. *Human Brain Mapping*, 38(3), 1155–1171. https://doi.org/10.1002/hbm.23445
- Bellmund, J., Gärdenfors, P., Moser, E. I., & Doeller, C. F. (2018). Navigating cognition: Spatial codes for human thinking. *Science*, *362*(6415), eaat6766. https://doi.org/10.1126/science.aat6766
- Bellmund, J., Polti, I., & Doeller, C. F. (2020). Sequence Memory in the Hippocampal–Entorhinal Region. *Journal of Cognitive Neuroscience*, 32(11), 2056–2070. https://doi.org/10.1162/jocn a 01592
- Ben-Shabat, E., Matyas, T. A., Pell, G. S., Brodtmann, A., & Carey, L. M. (2015). The Right Supramarginal Gyrus Is Important for Proprioception in Healthy and Stroke-Affected Participants: A Functional MRI Study. *Frontiers in Neurology*, 6. https://doi.org/10.3389/fneur.2015.00248
- Ben-Yakov, A., & Henson, R. N. (Eds.). (2018). The Hippocampal Film Editor: Sensitivity and Specificity to Event Boundaries in Continuous Experience. *The Journal of Neuroscience*, 38(47), 10057–10068. https://doi.org/10.1523/JNEUROSCI.0524-18.2018
- Berger-Tal, O., Nathan, J., Meron, E., & Saltz, D. (2014). The Exploration-Exploitation Dilemma: A Multidisciplinary Framework. *PLoS ONE*, 9(4), e95693. https://doi.org/10.1371/journal.pone.0095693
- Berridge, K. C., Aldridge, J. W., Houchard, K. R., & Zhuang, X. (2005). Sequential super-stereotypy of an instinctive fixed action pattern in hyper-dopaminergic mutant mice, a model of obsessive compulsive disorder and Tourette's. *BMC Biology*, 3(1), 4. https://doi.org/10.1186/1741-7007-3-4
- Berti, A., & Frassinetti, F. (2000). When Far Becomes Near: Remapping of Space by Tool Use. *Journal* of Cognitive Neuroscience, 12(3), 415–420. https://doi.org/10.1162/089892900562237

Bertolotti, T., & Magnani, L. (2017). Theoretical considerations on cognitive niche construction. Synthese, 194(12), 4757–4779. https://doi.org/10.1007/s11229-016-1165-2

- Biehl, M., Pollock, F. A., & Kanai, R. (2021). A Technical Critique of Some Parts of the Free Energy Principle. *Entropy*, 23(3), 293. https://doi.org/10.3390/e23030293
- Binder, J. R., Desai, R. H., Graves, W. W., & Conant, L. L. (2009). Where Is the Semantic System? A Critical Review and Meta-Analysis of 120 Functional Neuroimaging Studies. *Cerebral Cortex*, 19(12), 2767–2796. https://doi.org/10.1093/cercor/bhp055
- Birney, D. P., & Beckmann, J. F. (2022). Intelligence IS Cognitive Flexibility: Why Multilevel Models of Within-Individual Processes Are Needed to Realise This. *Journal of Intelligence*, 10(3), 49. https://doi.org/10.3390/jintelligence10030049
- Bjork, R. A. (1988). RETRIEVAL PRACTICE AND THE MAINTENANCE OF KNOWLEDGE.
- Bjork, R. A., & Bjork, E. L. (1992). A New Theory of Disuse and an Old Theory of Stimulus Fluctuation.
- Blanke, O., Mohr, C., Michel, C. M., Pascual-Leone, A., Brugger, P., Seeck, M., Landis, T., & Thut, G. (2005). Linking Out-of-Body Experience and Self Processing to Mental Own-Body Imagery at the Temporoparietal Junction. *The Journal of Neuroscience*, 25(3), 550–557. https://doi.org/10.1523/JNEUROSCI.2612-04.2005
- Boccara, C. N., Nardin, M., Stella, F., O'Neill, J., & Csicsvari, J. (2019). The entorhinal cognitive map is attracted to goals. *Science*, *363*(6434), 1443–1447. https://doi.org/10.1126/science.aav4837
- Boccia, M., Teghil, A., & Guariglia, C. (2019). Looking into recent and remote past: Meta-analytic evidence for cortical re-organization of episodic autobiographical memories. *Neuroscience & Biobehavioral Reviews*, 107, 84–95. https://doi.org/10.1016/j.neubiorev.2019.09.003
- Boesch, C., & Boesch, H. (1990). Tool Use and Tool Making in Wild Chimpanzees. Folia Primatologica, 54(1–2), 86–99. https://doi.org/10.1159/000156428
- Boettcher, S. E. P., Draschkow, D., Dienhart, E., & Võ, M. L.-H. (2018). Anchoring visual search in scenes: Assessing the role of anchor objects on eye movements during visual search. *Journal of Vision*, 18(13), 11. https://doi.org/10.1167/18.13.11
- Bonner, M. F., Peelle, J. E., Cook, P. A., & Grossman, M. (2013). Heteromodal conceptual processing in the angular gyrus. *NeuroImage*, 71, 175–186. https://doi.org/10.1016/j.neuroimage.2013.01.006
- Bousquet, F., & Le Page, C. (2004). Multi-agent simulations and ecosystem management: A review. *Ecological Modelling*, 176(3-4), 313–332. https://doi.org/10.1016/j.ecolmodel.2004.01.011
- Bratman, M. E. (1991). Intention, Plans, and Practical Reason. *The Philosophical Review*, 100(2), 277. https://doi.org/10.2307/2185304
- Bray, S., Almas, R., Arnold, A. E. G. F., Iaria, G., & MacQueen, G. (2015). Intraparietal Sulcus Activity and Functional Connectivity Supporting Spatial Working Memory Manipulation. *Cerebral Cortex*, 25(5), 1252–1264. https://doi.org/10.1093/cercor/bht320
- Breska, A., & Ivry, R. B. (2018). Double dissociation of single-interval and rhythmic temporal prediction in cerebellar degeneration and Parkinson's disease. *Proceedings of the National Academy of Sciences*, 115(48), 12283–12288. https://doi.org/10.1073/pnas.1810596115
- Bromberg-Martin, E. S., Matsumoto, M., & Hikosaka, O. (2010). Dopamine in Motivational Control: Rewarding, Aversive, and Alerting. *Neuron*, 68(5), 815–834. https://doi.org/10.1016/j.neuron.2010.11.022
- Brotchie, P., Iansek, R., & Horne, M. K. (1991). Motor function of the monkey globus pallidus: 2. Cognitive aspects of movement and phasic neuronal activity. *Brain*, 114(4), 1685–1702. https://doi.org/10.1093/brain/114.4.1685
- Brown, T. I., Ross, R. S., Tobyne, S. M., & Stern, C. E. (2012). Cooperative interactions between hippocampal and striatal systems support flexible navigation. *NeuroImage*, 60(2), 1316–1330. https://doi.org/10.1016/j.neuroimage.2012.01.046
- Brown, T. I., & Stern, C. E. (2014). Contributions of Medial Temporal Lobe and Striatal Memory Systems to Learning and Retrieving Overlapping Spatial Memories. *Cerebral Cortex*, 24(7), 1906–1922. https://doi.org/10.1093/cercor/bht041
- Bruineberg, J., Chemero, A., & Rietveld, E. (2019). General ecological information supports engagement with affordances for 'higher' cognition. *Synthese*, 196(12), 5231–5251. https://doi.org/10.1007/s11229-018-1716-9
- Bruineberg, J., Kiverstein, J., & Rietveld, E. (2018). The anticipating brain is not a scientist: The freeenergy principle from an ecological-enactive perspective. *Synthese*, *195*(6), 2417–2444. https://doi.org/10.1007/s11229-016-1239-1
- Bruineberg, J., & Rietveld, E. (2014). Self-organization, free energy minimization, and optimal grip on a field of affordances. *Frontiers in Human Neuroscience*, 8. https://doi.org/10.3389/fnhum.2014.00599

- Bruineberg, J., Rietveld, E., Parr, T., Van Maanen, L., & Friston, K. (2018). Free-energy minimization in joint agent-environment systems: A niche construction perspective. *Journal of Theoretical Biology*, 455, 161–178. https://doi.org/10.1016/j.jtbi.2018.07.002
- Bubeck, S., Chandrasekaran, V., Eldan, R., Gehrke, J., Horvitz, E., Kamar, E., Lee, P., Lee, Y. T., Li, Y., Lundberg, S., Nori, H., Palangi, H., Ribeiro, M. T., & Zhang, Y. (2023). Sparks of Artificial General Intelligence: Early experiments with GPT-4 (arXiv:2303.12712). arXiv. http://arXiv.org/abs/2303.12712
- Bubic. (2010). Prediction, cognition and the brain. *Frontiers in Human Neuroscience*. https://doi.org/10.3389/fnhum.2010.00025
- Buckner, R. L., & Carroll, D. C. (2007). Self-projection and the brain. *Trends in Cognitive Sciences*, 11(2), 49–57. https://doi.org/10.1016/j.tics.2006.11.004
- Buckner, R. L., & DiNicola, L. M. (2019). The brain's default network: Updated anatomy, physiology and evolving insights. *Nature Reviews Neuroscience*, 20(10), 593–608. https://doi.org/10.1038/s41583-019-0212-7
- Buden, B., Nowotny, S., Simon, S., Bery, A., & Cronin, M. (2009). Cultural translation: An introduction to the problem, and Responses. *Translation Studies*, 2(2), 196–219. https://doi.org/10.1080/14781700902937730
- Bufacchi, R. J., & Iannetti, G. D. (2018). An Action Field Theory of Peripersonal Space. Trends in Cognitive Sciences, 22(12), 1076–1090. https://doi.org/10.1016/j.tics.2018.09.004
- Burns, E. J., Arnold, T., & Bukach, C. M. (2019). P-curving the fusiform face area: Meta-analyses support the expertise hypothesis. *Neuroscience & Biobehavioral Reviews*, 104, 209–221. https://doi.org/10.1016/j.neubiorev.2019.07.003
- Butler, W. N., Hardcastle, K., & Giocomo, L. M. (2019). Remembered reward locations restructure entorhinal spatial maps. *Science*, 363(6434), 1447–1452. https://doi.org/10.1126/science.aav5297
- Butters, N., Wolfe, J., Martone, M., Granholm, E., & Cermak, L. S. (1985). Memory disorders associated with huntington's disease: Verbal recall, verbal recognition and procedural memory. *Neuropsychologia*, 23(6), 729–743. https://doi.org/10.1016/0028-3932(85)90080-6
- Cardinal, R. N., Parkinson, J. A., Hall, J., & Everitt, B. J. (2002). Emotion and motivation: The role of the amygdala, ventral striatum, and prefrontal cortex. *Neuroscience & Biobehavioral Reviews*, 26(3), 321–352. https://doi.org/10.1016/S0149-7634(02)00007-6
- Carhart-Harris, R. L., & Friston, K. J. (2010). The default-mode, ego-functions and free-energy: A neurobiological account of Freudian ideas. *Brain*, *133*(4), 1265–1283. https://doi.org/10.1093/brain/awq010
- Carroll, J. (2020). Imagination, the Brain's Default Mode Network, and Imaginative Verbal Artifacts. In J. Carroll, M. Clasen, & E. Jonsson (Eds.), *Evolutionary Perspectives on Imaginative Culture* (pp. 31–52). Springer International Publishing. https://doi.org/10.1007/978-3-030-46190-4_2
- Caspers, S., Schleicher, A., Bacha-Trams, M., Palomero-Gallagher, N., Amunts, K., & Zilles, K. (2013). Organization of the Human Inferior Parietal Lobule Based on Receptor Architectonics. *Cerebral Cortex*, 23(3), 615–628. https://doi.org/10.1093/cercor/bhs048
- Chang, L., & Tsao, D. Y. (2017). The Code for Facial Identity in the Primate Brain. *Cell*, 169(6), 1013-1028.e14. https://doi.org/10.1016/j.cell.2017.05.011
- Chase, J. M., & Leibold, M. A. (2003). *Ecological niches: Linking classical and contemporary approaches*. University of Chicago Press.
- Chemero, A. (2001). What We Perceive When We Perceive Affordances: Commentary on Michaels (2000) "Information, Perception, and Action." *Ecological Psychology*, *13*(2), 111–116. https://doi.org/10.1207/S15326969ECO1302_3
- Chemero, A. (2003). An Outline of a Theory of Affordances. *Ecological Psychology*, 15(2), 181–195. https://doi.org/10.1207/S15326969ECO1502_5
- Chemero, A. (2009). Radical embodied cognitive science. MIT Press.
- Chen, A. P., Chen, L., Kim, T. A., & Xiong, Q. (2021). Integrating the Roles of Midbrain Dopamine Circuits in Behavior and Neuropsychiatric Disease. *Biomedicines*, 9(6), 647. https://doi.org/10.3390/biomedicines9060647
- Chen, J., Leong, Y. C., Honey, C. J., Yong, C. H., Norman, K. A., & Hasson, U. (2017). Shared memories reveal shared structure in neural activity across individuals. *Nature Neuroscience*, 20(1), 115–125. https://doi.org/10.1038/nn.4450
- Chen, M., He, Y., Hao, L., Xu, J., Tian, T., Peng, S., Zhao, G., Lu, J., Zhao, Y., Zhao, H., Jiang, M., Gao, J.-H., Tan, S., He, Y., Liu, C., Tao, S., Uddin, L. Q., Dong, Q., & Qin, S. (2022). Default mode network scaffolds immature frontoparietal network in cognitive development. *Cerebral Cortex*, bhac414. https://doi.org/10.1093/cercor/bhac414

Chenot, Q., Lepron, E., De Boissezon, X., & Scannella, S. (2021). Functional Connectivity Within the Fronto-Parietal Network Predicts Complex Task Performance: A fNIRS Study. *Frontiers in Neuroergonomics*, 2, 718176. https://doi.org/10.3389/fnrgo.2021.718176

Chomsky, N. (1959). Verbal behavior. Language, 35(1), 26. https://doi.org/10.2307/411334

- Christensen, W., Sutton, J., & Bicknell, K. (2019). Memory systems and the control of skilled action. *Philosophical Psychology*, *32*(5), 692–718. https://doi.org/10.1080/09515089.2019.1607279
- Clark, A. (2006). Language, embodiment, and the cognitive niche. *Trends in Cognitive Sciences*, 10(8), 370–374. https://doi.org/10.1016/j.tics.2006.06.012
- Clark, A. (2013). Whatever next? Predictive brains, situated agents, and the future of cognitive science. Behavioral and Brain Sciences, 36(3), 181–204. https://doi.org/10.1017/S0140525X12000477
- Clark, A. (2015). Surfing Uncertainty. Oxford University Press.
- Clark, A. (2017). How to Knit Your Own Markov Blanket: How to Knit Your Own Markov Blanket: Resisting the Second Law with Metamorphic Minds: Resisting the Second Law with Metamorphic Minds. *Philosophy and Predictive Processing*. https://doi.org/10.15502/9783958573031
- Clark, A., & Chalmers, D. J. (1998). The Extended Mind. *Analysis*, 58(1), 7–19. https://doi.org/10.1093/analys/58.1.7
- Clark, J. E., & Whitall, J. (2021). Motor Development: A Perspective on the Past, the Present, and the Future. *Kinesiology Review*, 10(3), 264–273. https://doi.org/10.1123/kr.2021-0023
- Clower, D. M., West, R. A., Lynch, J. C., & Strick, P. L. (2001). The Inferior Parietal Lobule Is the Target of Output from the Superior Colliculus, Hippocampus, and Cerebellum. *The Journal of Neuroscience*, 21(16), 6283–6291. https://doi.org/10.1523/JNEUROSCI.21-16-06283.2001
- Cohen, N. J., & Squire, L. R. (1980). Preserved Learning and Retention of Pattern-Analyzing Skill in Amnesia: Dissociation of Knowing How and Knowing That. *Science*, 210(4466), 207–210. https://doi.org/10.1126/science.7414331
- Cole, M. W., Bassett, D. S., Power, J. D., Braver, T. S., & Petersen, S. E. (2014). Intrinsic and Task-Evoked Network Architectures of the Human Brain. *Neuron*, 83(1), 238–251. https://doi.org/10.1016/j.neuron.2014.05.014
- Collin, S. H. P., Milivojevic, B., & Doeller, C. F. (2015a). Memory hierarchies map onto the hippocampal long axis in humans. *Nature Neuroscience*, 18(11), 1562–1564. https://doi.org/10.1038/nn.4138
- Collin, S. H. P., Milivojevic, B., & Doeller, C. F. (2015b). Memory hierarchies map onto the hippocampal long axis in humans. *Nature Neuroscience*, 18(11), 1562–1564. https://doi.org/10.1038/nn.4138
- Connolly, A. C., Guntupalli, J. S., Gors, J., Hanke, M., Halchenko, Y. O., Wu, Y.-C., Abdi, H., & Haxby, J. V. (2012). The Representation of Biological Classes in the Human Brain. *The Journal of Neuroscience*, 32(8), 2608–2618. https://doi.org/10.1523/JNEUROSCI.5547-11.2012
- Constant, A., Clark, A., & Friston, K. (2021). Representation Wars: Enacting an Armistice Through Active Inference. *Frontiers in Psychology*, *11*, 598733. https://doi.org/10.3389/fpsyg.2020.598733
- Constant, A., Ramstead, M. J. D., Veissière, S. P. L., & Friston, K. (2019). Regimes of Expectations: An Active Inference Model of Social Conformity and Human Decision Making. *Frontiers in Psychology*, 10, 679. https://doi.org/10.3389/fpsyg.2019.00679
- Corbetta, M., & Shulman, G. L. (2002). Control of goal-directed and stimulus-driven attention in the brain. *Nature Reviews Neuroscience*, *3*(3), 201–215. https://doi.org/10.1038/nrn755
- Corcoran, A. W., Pezzulo, G., & Hohwy, J. (2020). From allostatic agents to counterfactual cognisers: Active inference, biological regulation, and the origins of cognition. *Biology & Philosophy*, 35(3), 32. https://doi.org/10.1007/s10539-020-09746-2
- Costandi, M. (2013). Researchers scare "fearless" patients. *Nature*, nature.2013.12350. https://doi.org/10.1038/nature.2013.12350
- Coyne, S. M., Rogers, A. A., Zurcher, J. D., Stockdale, L., & Booth, M. (2020). Does time spent using social media impact mental health?: An eight year longitudinal study. *Computers in Human Behavior*, 104, 106160. https://doi.org/10.1016/j.chb.2019.106160
- Coynel, D., Marrelec, G., Perlbarg, V., Pélégrini-Issac, M., Van de Moortele, P.-F., Ugurbil, K., Doyon, J., Benali, H., & Lehéricy, S. (2010). Dynamics of motor-related functional integration during motor sequence learning. *NeuroImage*, 49(1), 759–766. https://doi.org/10.1016/j.neuroimage.2009.08.048
- Crego, A. C. G., Štoček, F., Marchuk, A. G., Carmichael, J. E., Van Der Meer, M. A. A., & Smith, K. S. (2020). Complementary Control over Habits and Behavioral Vigor by Phasic Activity in the

Dorsolateral Striatum. *The Journal of Neuroscience*, 40(10), 2139–2153. https://doi.org/10.1523/JNEUROSCI.1313-19.2019

- Crittenden, B. M., Mitchell, D. J., & Duncan, J. (2015). Recruitment of the default mode network during a demanding act of executive control. *eLife*, *4*, e06481. https://doi.org/10.7554/eLife.06481
- Cruz, F., Parisi, G. I., & Wermter, S. (2018). Multi-modal Feedback for Affordance-driven Interactive Reinforcement Learning. 2018 International Joint Conference on Neural Networks (IJCNN), 1– 8. https://doi.org/10.1109/IJCNN.2018.8489237
- Cullen, K. E., & Taube, J. S. (2017). Our sense of direction: Progress, controversies and challenges. *Nature Neuroscience*, 20(11), 1465–1473. https://doi.org/10.1038/nn.4658
- Cunningham, W. A., & Brosch, T. (2012). Motivational Salience: Amygdala Tuning From Traits, Needs, Values, and Goals. *Current Directions in Psychological Science*, 21(1), 54–59. https://doi.org/10.1177/0963721411430832
- Curran, T., & Keele, S. W. (1993). Attentional and nonattentional forms of sequence learning. *Journal of Experimental Psychology: Learning, Memory, and Cognition, 19*(1), 189.
- Da Costa, L., Friston, K., Heins, C., & Pavliotis, G. A. (2021). Bayesian mechanics for stationary processes. *Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 477(2256), 20210518. https://doi.org/10.1098/rspa.2021.0518
- da Silva, J. A., Tecuapetla, F., Paixão, V., & Costa, R. M. (2018). Dopamine neuron activity before action initiation gates and invigorates future movements. *Nature*, 554(7691), 244–248. https://doi.org/10.1038/nature25457
- Dalton, M. A., D'Souza, A., Lv, J., & Calamante, F. (2022). New insights into anatomical connectivity along the anterior-posterior axis of the human hippocampus using in vivo quantitative fibre tracking. *eLife*, 11, e76143. https://doi.org/10.7554/eLife.76143
- Danjo, T., Toyoizumi, T., & Fujisawa, S. (2018). Spatial representations of self and other in the hippocampus. *Science*, 359(6372), 213–218. https://doi.org/10.1126/science.aao3898
- Davey, C. G., Pujol, J., & Harrison, B. J. (2016). Mapping the self in the brain's default mode network. *NeuroImage*, 132, 390–397. https://doi.org/10.1016/j.neuroimage.2016.02.022
- Davey, J., Thompson, H. E., Hallam, G., Karapanagiotidis, T., Murphy, C., De Caso, I., Krieger-Redwood, K., Bernhardt, B. C., Smallwood, J., & Jefferies, E. (2016). Exploring the role of the posterior middle temporal gyrus in semantic cognition: Integration of anterior temporal lobe with executive processes. *NeuroImage*, 137, 165–177. https://doi.org/10.1016/j.neuroimage.2016.05.051
- Davidson, D. (1963). Actions, Reasons, and Causes. *The Journal of Philosophy*, 60(23), 685. https://doi.org/10.2307/2023177
- Daw, N. D., Courville, A. C., & Touretzky, D. S. (2006). Representation and Timing in Theories of the Dopamine System. *Neural Computation*, 18(7), 1637–1677. https://doi.org/10.1162/neco.2006.18.7.1637
- Daw, N. D., Gershman, S. J., Seymour, B., Dayan, P., & Dolan, R. J. (2011). Model-Based Influences on Humans' Choices and Striatal Prediction Errors. *Neuron*, 69(6), 1204–1215. https://doi.org/10.1016/j.neuron.2011.02.027
- Daw, N. D., Niv, Y., & Dayan, P. (2005). Uncertainty-based competition between prefrontal and dorsolateral striatal systems for behavioral control. *Nature Neuroscience*, 8(12), 1704–1711. https://doi.org/10.1038/nn1560
- Dawkins, R. (1982). The Extended Phenotype. 287.
- Day, B. L., & Fitzpatrick, R. C. (2005). The vestibular system. *Current Biology*. https://doi.org/10.1016/j.cub.2005.07.053
- Dayan, P. (2009). Goal-directed control and its antipodes. *Neural Networks*, 22(3), 213–219. https://doi.org/10.1016/j.neunet.2009.03.004
- De Cesarei, A., Loftus, G. R., Mastria, S., & Codispoti, M. (2017). Understanding natural scenes: Contributions of image statistics. *Neuroscience & Biobehavioral Reviews*, 74, 44–57. https://doi.org/10.1016/j.neubiorev.2017.01.012
- De Wit, S., Watson, P., Harsay, H. A., Cohen, M. X., Van De Vijver, I., & Ridderinkhof, K. R. (2012). Corticostriatal Connectivity Underlies Individual Differences in the Balance between Habitual and Goal-Directed Action Control. *The Journal of Neuroscience*, *32*(35), 12066–12075. https://doi.org/10.1523/JNEUROSCI.1088-12.2012
- DeCoteau, W. E., Thorn, C., Gibson, D. J., Courtemanche, R., Mitra, P., Kubota, Y., & Graybiel, A. M. (2007). Learning-related coordination of striatal and hippocampal theta rhythms during acquisition of a procedural maze task. *Proceedings of the National Academy of Sciences*, 104(13), 5644–5649. https://doi.org/10.1073/pnas.0700818104

- Degrave, J., Felici, F., Buchli, J., Neunert, M., Tracey, B., Carpanese, F., Ewalds, T., Hafner, R., Abdolmaleki, A., de las Casas, D., Donner, C., Fritz, L., Galperti, C., Huber, A., Keeling, J., Tsimpoukelli, M., Kay, J., Merle, A., Moret, J.-M., ... Riedmiller, M. (2022). Magnetic control of tokamak plasmas through deep reinforcement learning. *Nature*, 602(7897), 414–419. https://doi.org/10.1038/s41586-021-04301-9
- Dehaene, S., & Changeux, J.-P. (2011). Experimental and Theoretical Approaches to Conscious Processing. *Neuron*, 70(2), 200–227. https://doi.org/10.1016/j.neuron.2011.03.018
- Dehaene, S., Charles, L., King, J.-R., & Marti, S. (2014). Toward a computational theory of conscious processing. *Current Opinion in Neurobiology*, 25, 76–84. https://doi.org/10.1016/j.conb.2013.12.005
- Dehaene, S., & Cohen, L. (2007). Cultural Recycling of Cortical Maps. *Neuron*, 56(2), 384–398. https://doi.org/10.1016/j.neuron.2007.10.004
- Dehghani, M., Boghrati, R., Man, K., Hoover, J., Gimbel, S. I., Vaswani, A., Zevin, J. D., Immordino-Yang, M. H., Gordon, A. S., Damasio, A., & Kaplan, J. T. (2017). Decoding the neural representation of story meanings across languages. *Human Brain Mapping*, 38(12), 6096–6106. https://doi.org/10.1002/hbm.23814
- Del Río, P., & Álvarez, A. (2007). Inside and Outside the Zone of Proximal Development: An Ecofunctional Reading of Vygotsky. In H. Daniels, M. Cole, & J. V. Wertsch (Eds.), The Cambridge Companion to Vygotsky (1st ed., pp. 276–304). Cambridge University Press. https://doi.org/10.1017/CCOL0521831040.012
- Desrochers, T. M., Burk, D. C., Badre, D., & Sheinberg, D. L. (2016). The Monitoring and Control of Task Sequences in Human and Non-Human Primates. *Frontiers in Systems Neuroscience*, 9. https://doi.org/10.3389/fnsys.2015.00185
- Devan, B. D., & White, N. M. (1999). Parallel Information Processing in the Dorsal Striatum: Relation to Hippocampal Function. *The Journal of Neuroscience*, 19(7), 2789–2798. https://doi.org/10.1523/JNEUROSCI.19-07-02789.1999
- Dezfouli, A., & Balleine. (2012). Habits, action sequences and reinforcement learning: Habits and action sequences. *European Journal of Neuroscience*, 35(7), 1036–1051. https://doi.org/10.1111/j.1460-9568.2012.08050.x
- Dezfouli, A., Lingawi, N. W., & Balleine. (2014). Habits as action sequences: Hierarchical action control and changes in outcome value. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 369(1655), 20130482. https://doi.org/10.1098/rstb.2013.0482
- Di Domenico, F. (2020). From biomechanics to learning: Continuum for the theory of physical and sports education. *Journal of Human Sport and Exercise - 2020 - Winter Conferences of Sports Science*. Journal of Human Sport and Exercise - 2020 - Winter Conferences of Sports Science. https://doi.org/10.14198/jhse.2020.15.Proc2.18
- Dickinson, A. (1994). Instrumental conditioning. *Animal Learning and Cognition*, 45–79. https://doi.org/10.1007/978-3-540-68706-1_343
- Diederen, K. M. J., & Fletcher, P. C. (2021). Dopamine, Prediction Error and Beyond. *The Neuroscientist*, 27(1), 30–46. https://doi.org/10.1177/1073858420907591
- Dixon, M. L., Andrews-Hanna, J. R., Spreng, R. N., Irving, Z. C., Mills, C., Girn, M., & Christoff, K. (2017). Interactions between the default network and dorsal attention network vary across default subsystems, time, and cognitive states. *NeuroImage*, 147, 632–649. https://doi.org/10.1016/j.neuroimage.2016.12.073
- Dixon, M. L., De La Vega, A., Mills, C., Andrews-Hanna, J., Spreng, R. N., Cole, M. W., & Christoff, K. (2018). Heterogeneity within the frontoparietal control network and its relationship to the default and dorsal attention networks. *Proceedings of the National Academy of Sciences*, 115(7). https://doi.org/10.1073/pnas.1715766115
- Djebbara, Z., Fich, L. B., & Gramann, K. (2020). Architectural Affordance Impacts Human Sensorimotor Brain Dynamics [Preprint]. Neuroscience. https://doi.org/10.1101/2020.10.18.344267
- Doeller, C. F., King, J. A., & Burgess, N. (2008). Parallel striatal and hippocampal systems for landmarks and boundaries in spatial memory. *Proceedings of the National Academy of Sciences*, 105(15), 5915–5920. https://doi.org/10.1073/pnas.0801489105
- Dohmatob, E., Dumas, G., & Bzdok, D. (2017). Dark Control: Towards a Unified Account of Default Mode Function by Markov Decision Processes [Preprint]. Neuroscience. https://doi.org/10.1101/148890
- Dolan, R. J., & Dayan, P. (2013). Goals and Habits in the Brain. *Neuron*, 80(2), 312–325. https://doi.org/10.1016/j.neuron.2013.09.007

- Doll, B. B., Jacobs, W. J., Sanfey, A. G., & Frank, M. J. (2009). Instructional control of reinforcement learning: A behavioral and neurocomputational investigation. *Brain Research*, 1299, 74–94. https://doi.org/10.1016/j.brainres.2009.07.007
- Donnarumma, F., Costantini, M., Ambrosini, E., Friston, K., & Pezzulo, G. (2017). Action perception as hypothesis testing. *Cortex*, 89, 45–60. https://doi.org/10.1016/j.cortex.2017.01.016
- Doolittle, P. E. (1997). Vygotsky's Zone of Proximal Development as a Theoretical Foundation for Cooperative Learning. *Journal on Excellence in College Teaching*, 8(1), 83–103.
- Dosenbach, N. U. F., Fair, D. A., Cohen, A. L., Schlaggar, B. L., & Petersen, S. E. (2008). A dualnetworks architecture of top-down control. *Trends in Cognitive Sciences*, 12(3), 99–105. https://doi.org/10.1016/j.tics.2008.01.001
- Dougherty, P. (2020). Chapter 2: Somatosensory Systems. In *Neuroanatomy Online*. https://nba.uth.tmc.edu/neuroscience/m/s2/chapter02.html
- Doya, K. (1999). What are the computations of the cerebellum, the basal ganglia and the cerebral cortex? *Neural Networks*, *12*(7–8), 961–974. https://doi.org/10.1016/S0893-6080(99)00046-5
- Doya, K., Samejima, K., Katagiri, K., & Kawato, M. (2002). Multiple Model-Based Reinforcement Learning. *Neural Computation*, 14(6), 1347–1369. https://doi.org/10.1162/089976602753712972
- Drummond, N., & Niv, Y. (2020). Model-based decision making and model-free learning. *Current Biology*, *30*(15), R860–R865. https://doi.org/10.1016/j.cub.2020.06.051
- Dubois, M., Bowler, A., Moses-Payne, M. E., Habicht, J., Moran, R., Steinbeis, N., & Hauser, T. U. (2022). Exploration heuristics decrease during youth. *Cognitive, Affective, & Behavioral Neuroscience*, 22(5), 969–983. https://doi.org/10.3758/s13415-022-01009-9
- Dubois, M., Habicht, J., Michely, J., Moran, R., Dolan, R. J., & Hauser, T. U. (2021). Human complex exploration strategies are enriched by noradrenaline-modulated heuristics. *eLife*, 10, e59907. https://doi.org/10.7554/eLife.59907
- Duhamel, J.-R., Colby, C. L., & Goldberg, M. E. (1998). Ventral Intraparietal Area of the Macaque: Congruent Visual and Somatic Response Properties. *Journal of Neurophysiology*, 79(1), 126– 136. https://doi.org/10.1152/jn.1998.79.1.126
- Duke, D. C., Moran, L. B., Pearce, R. K. B., & Graeber, M. B. (2007). The medial and lateral substantia nigra in Parkinson's disease: mRNA profiles associated with higher brain tissue vulnerability. *Neurogenetics*, 8(2), 83–94. https://doi.org/10.1007/s10048-006-0077-6
- Duncan, J. (2010). The multiple-demand (MD) system of the primate brain: Mental programs for intelligent behaviour. *Trends in Cognitive Sciences*, *14*(4), 172–179. https://doi.org/10.1016/j.tics.2010.01.004
- Duncan, J., Assem, M., & Shashidhara, S. (2020). Integrated Intelligence from Distributed Brain Activity. *Trends in Cognitive Sciences*, 24(10), 838–852. https://doi.org/10.1016/j.tics.2020.06.012
- Duszkiewicz, A. J., McNamara, C. G., Takeuchi, T., & Genzel, L. (2019). Novelty and Dopaminergic Modulation of Memory Persistence: A Tale of Two Systems. *Trends in Neurosciences*, 42(2), 102–114. https://doi.org/10.1016/j.tins.2018.10.002
- Eichenbaum, H. (2000). A cortical-hippocampal system for declarative memory. *Nature Reviews Neuroscience*, 1(1), 41–50. https://doi.org/10.1038/35036213
- Eichenbaum, H. (2001). The hippocampus and declarative memory: Cognitive mechanisms and neural codes. *Behavioural Brain Research*, *127*(1–2), 199–207. https://doi.org/10.1016/S0166-4328(01)00365-5
- Eichenbaum, H., Dudchenko, P., Wood, E., Shapiro, M., & Tanila, H. (1999). The hippocampus, memory, and place cells: Is it spatial memory or a memory space? *Neuron*, 23(2), 209–226. https://doi.org/10.1016/S0896-6273(00)80773-4
- Eismann, K., Posegga, O., & Fischbach, K. (2021). Opening organizational learning in crisis management: On the affordances of social media. *The Journal of Strategic Information Systems*, 30(4), 101692. https://doi.org/10.1016/j.jsis.2021.101692
- Ekstrom, A. D., Arnold, A. E. G. F., & Iaria, G. (2014). A critical review of the allocentric spatial representation and its neural underpinnings: Toward a network-based perspective. *Frontiers in Human Neuroscience*, 8. https://doi.org/10.3389/fnhum.2014.00803
- Epstein, R. A., Patai, E. Z., Julian, J. B., & Spiers, H. J. (2017). The cognitive map in humans: Spatial navigation and beyond. *Nature Neuroscience*, 20(11), 1504–1513. https://doi.org/10.1038/nn.4656
- Ericson, J. D., & Warren, W. H. (2020). Probing the invariant structure of spatial knowledge: Support for the cognitive graph hypothesis. *Cognition*, 200, 104276. https://doi.org/10.1016/j.cognition.2020.104276

- Ericsson, K. A. (2006). The Influence of Experience and Deliberate Practice on the Development of Superior Expert Performance. *The Cambridge Handbook of Expertise and Expert Performance*.
- Farmer, J. D., & Foley, D. (2009). The economy needs agent-based modelling. *Nature*, 460(7256), 685–686. https://doi.org/10.1038/460685a
- Fedorenko, E., Duncan, J., & Kanwisher, N. (2013). Broad domain generality in focal regions of frontal and parietal cortex. *Proceedings of the National Academy of Sciences*, 110(41), 16616–16621. https://doi.org/10.1073/pnas.1315235110
- Feher Da Silva, C., & Hare, T. A. (2020). Humans primarily use model-based inference in the two-stage task. *Nature Human Behaviour*, 4(10), 1053–1066. https://doi.org/10.1038/s41562-020-0905-y
- Feinstein, J. S., Buzza, C., Hurlemann, R., Follmer, R. L., Dahdaleh, N. S., Coryell, W. H., Welsh, M. J., Tranel, D., & Wemmie, J. A. (2013). Fear and panic in humans with bilateral amygdala damage. *Nature Neuroscience*, 16(3), 270–272. https://doi.org/10.1038/nn.3323
- Ferbinteanu, J. (2016). Contributions of Hippocampus and Striatum to Memory-Guided Behavior Depend on Past Experience. *The Journal of Neuroscience*, 36(24), 6459–6470. https://doi.org/10.1523/JNEUROSCI.0840-16.2016
- Fermin, A., Yoshida, T., Ito, M., Yoshimoto, J., & Doya, K. (2010). Evidence for Model-Based Action Planning in a Sequential Finger Movement Task. *Journal of Motor Behavior*, 42(6), 371–379. https://doi.org/10.1080/00222895.2010.526467
- Ferré, S., Bonaventura, J., Zhu, W., Hatcher-Solis, C., Taura, J., Quiroz, C., Cai, N.-S., Moreno, E., Casadó-Anguera, V., Kravitz, A. V., Thompson, K. R., Tomasi, D. G., Navarro, G., Cordomí, A., Pardo, L., Lluís, C., Dessauer, C. W., Volkow, N. D., Casadó, V., ... Zwilling, D. (2018). Essential Control of the Function of the Striatopallidal Neuron by Pre-coupled Complexes of Adenosine A2A-Dopamine D2 Receptor Heterotetramers and Adenylyl Cyclase. *Frontiers in Pharmacology*, *9*, 243. https://doi.org/10.3389/fphar.2018.00243
- Ferrucci, M., S Giorgi, F., Bartalucci, A., L Busceti, C., & Fornai, F. (2013). The effects of locus coeruleus and norepinephrine in methamphetamine toxicity. *Current Neuropharmacology*, 11(1), 80–94. https://doi.org/10.2174/157015913804999522
- Findling, C., Skvortsova, V., Dromnelle, R., Palminteri, S., & Wyart, V. (2019). Computational noise in reward-guided learning drives behavioral variability in volatile environments. *Nature Neuroscience*, 22(12), 2066–2077. https://doi.org/10.1038/s41593-019-0518-9
- Fitts, P., & Posner, M. (1967). Human Performance.
- Fjell, A. M., Westlye, L. T., Amlien, I., Tamnes, C. K., Grydeland, H., Engvig, A., Espeseth, T., Reinvang, I., Lundervold, A. J., Lundervold, A., & Walhovd, K. B. (2015). High-Expanding Cortical Regions in Human Development and Evolution Are Related to Higher Intellectual Abilities. *Cerebral Cortex*, 25(1), 26–34. https://doi.org/10.1093/cercor/bht201
- Frank, M. J., Doll, B. B., Oas-Terpstra, J., & Moreno, F. (2009). Prefrontal and striatal dopaminergic genes predict individual differences in exploration and exploitation. *Nature Neuroscience*, 12(8), 1062–1068. https://doi.org/10.1038/nn.2342
- Frazier, L., & Clifton, C. (1996). Construal. Mit Press.
- Frigon, A., Akay, T., & Prilutsky, B. I. (2021). Control of Mammalian Locomotion by Somatosensory Feedback. In R. Terjung (Ed.), *Comprehensive Physiology* (1st ed., pp. 2877–2947). Wiley. https://onlinelibrary.wiley.com/doi/10.1002/cphy.c210020
- Friston, K. (2010). The free-energy principle: A unified brain theory? *Nature Reviews Neuroscience*, *11*(2), 127–138. https://doi.org/10.1038/nrn2787
- Friston, K. (2011). Embodied inference: Or "I think therefore I am, if I am what I think." *Exeter: Imprint* Academic, In W. Tschacher&C. Bergomi (Eds.), The implications of embodiment (cognition and communication), 89–125.
- Friston, K. (2012). A Free Energy Principle for Biological Systems. *Entropy*, 14(11), 2100–2121. https://doi.org/10.3390/e14112100
- Friston, K. (2013). Life as we know it. *Journal of The Royal Society Interface*, 10(86), 20130475. https://doi.org/10.1098/rsif.2013.0475
- Friston, K. (2018). Am I Self-Conscious? (Or Does Self-Organization Entail Self-Consciousness?). Frontiers in Psychology, 9, 579. https://doi.org/10.3389/fpsyg.2018.00579
- Friston, K., & Buzsáki, G. (2016a). The Functional Anatomy of Time: What and When in the Brain. *Trends in Cognitive Sciences*, 20(7), 500–511. https://doi.org/10.1016/j.tics.2016.05.001
- Friston, K., & Buzsáki, G. (2016b). The Functional Anatomy of Time: What and When in the Brain. *Trends in Cognitive Sciences*, 20(7), 500–511. https://doi.org/10.1016/j.tics.2016.05.001
- Friston, K., Costa, L. D., Sajid, N., Heins, C., Ueltzhöffer, K., Pavliotis, G. A., & Parr, T. (2023). *The free energy principle made simpler but not too simple*.

- Friston, K., Da Costa, L., & Parr, T. (2021). Some Interesting Observations on the Free Energy Principle. Entropy, 23(8), 1076. https://doi.org/10.3390/e23081076
- Friston, K., Heins, C., Ueltzhöffer, K., Da Costa, L., & Parr, T. (2021). Stochastic Chaos and Markov Blankets. *Entropy*, 23(9), 1220. https://doi.org/10.3390/e23091220
- Friston, K., & Stephan, K. E. (2007). Free-energy and the brain. Synthese, 159(3), 417-458.
 - https://doi.org/10.1007/s11229-007-9237-y
- Frith, C. D. (2007). Making up the mind: How the brain creates our mental world. Blackwell Pub.
- Frith, C. D., & Frith, U. (2006). The Neural Basis of Mentalizing. *Neuron*, 50(4), 531–534. https://doi.org/10.1016/j.neuron.2006.05.001
- Fulford, J., Milton, F., Salas, D., Smith, A., Simler, A., Winlove, C., & Zeman, A. (2018). The neural correlates of visual imagery vividness – An fMRI study and literature review. *Cortex*, 105, 26– 40. https://doi.org/10.1016/j.cortex.2017.09.014
- Fultot, M., & Turvey, M. T. (2019). Von Uexküll's Theory of Meaning and Gibson's Organism– Environment Reciprocity. *Ecological Psychology*, 31(4), 289–315. https://doi.org/10.1080/10407413.2019.1619455
- Fyhn, M., Molden, S., Witter, M. P., Moser, E. I., & Moser, M.-B. (2004). Spatial Representation in the Entorhinal Cortex. Science, 305(5688), 1258–1264. https://doi.org/10.1126/science.1099901
- Gabriel, I. (2020). Artificial Intelligence, Values, and Alignment. *Minds and Machines*, 30(3), 411–437. https://doi.org/10.1007/s11023-020-09539-2
- Gabrieli, J. D. E. (1998). COGNITIVE NEUROSCIENCE OF HUMAN MEMORY. Annual Review of Psychology, 49(1), 87–115. https://doi.org/10.1146/annurev.psych.49.1.87
- Gallagher, H. L., & Frith, C. D. (2003). Functional imaging of 'theory of mind.' *Trends in Cognitive Sciences*, 7(2), 77–83. https://doi.org/10.1016/S1364-6613(02)00025-6
- Gallagher, S. (1986). Body Image and Body Schema: A Conceptual Clarification. *The Journal of Mind and Behavior*, 7(4), 541–554.
- Gallagher, S., & Bower, M. (2014). Making enactivism even more embodied. Avant, 18.
- Galli, G., Noel, J. P., Canzoneri, E., Blanke, O., & Serino, A. (2015). The wheelchair as a full-body tool extending the peripersonal space. *Frontiers in Psychology*, 6. https://doi.org/10.3389/fpsyg.2015.00639
- Gao, W., Gilmore, J. H., Shen, D., Smith, J. K., Zhu, H., & Lin, W. (2013). The Synchronization within and Interaction between the Default and Dorsal Attention Networks in Early Infancy. *Cerebral Cortex*, 23(3), 594–603. https://doi.org/10.1093/cercor/bhs043
- Gärdenfors, P. (2004). Conceptual Spaces as a Framework for Knowledge Representation.
- Gärdenfors, P. (2014). Geometry of meaning: Semantics based on conceptual spaces. The MIT Press.
- Garin, C. M., Hori, Y., Everling, S., Whitlow, C. T., Calabro, F. J., Luna, B., Froesel, M., Gacoin, M., Ben Hamed, S., Dhenain, M., & Constantinidis, C. (2022). An evolutionary gap in primate default mode network organization. *Cell Reports*, 39(2), 110669. https://doi.org/10.1016/j.celrep.2022.110669
- Geeraerts, D. (Ed.). (2006). Cognitive linguistics: Basic readings. Mouton de Gruyter.
- Geerts, J. P., Chersi, F., Stachenfeld, K. L., & Burgess, N. (2020a). A general model of hippocampal and dorsal striatal learning and decision making. *Proceedings of the National Academy of Sciences*, 117(49), 31427–31437. https://doi.org/10.1073/pnas.2007981117
- Geerts, J. P., Chersi, F., Stachenfeld, K. L., & Burgess, N. (2020b). A general model of hippocampal and dorsal striatal learning and decision making. *Proceedings of the National Academy of Sciences*, 117(49), 31427–31437. https://doi.org/10.1073/pnas.2007981117
- Gentile, G., Petkova, V. I., & Ehrsson, H. H. (2011). Integration of Visual and Tactile Signals From the Hand in the Human Brain: An fMRI Study. *Journal of Neurophysiology*, *105*(2), 910–922. https://doi.org/10.1152/jn.00840.2010
- Gershman, S. J. (2019). Uncertainty and exploration. *Decision*, 6(3), 277–286. https://doi.org/10.1037/dec0000101
- Gershman, S. J., Blei, D. M., & Niv, Y. (2010). Context, learning, and extinction. *Psychological Review*, 117(1), 197–209. https://doi.org/10.1037/a0017808
- Gershman, S. J., Moore, C. D., Todd, M. T., Norman, K. A., & Sederberg, P. B. (2012). The Successor Representation and Temporal Context. *Neural Computation*, 24(6), 1553–1568. https://doi.org/10.1162/NECO a 00282
- Gheysen, F., & Fias, W. (2012). Dissociable neural systems of sequence learning. *Advances in Cognitive Psychology*, 8(2), 73–82. https://doi.org/10.2478/v10053-008-0105-1
- Gibson, J. J. (1950). The perception of the visual world.
- Gibson, J. J. (2014a). The Ecological Approach to Visual Perception. 347.
- Gibson, J. J. (2014b). The ecological approach to visual perception: Classic edition. Psychology press.

- Gigerenzer, G., & Gaissmaier, W. (2011). Heuristic Decision Making. *Annual Review of Psychology*, 62(1), 451–482. https://doi.org/10.1146/annurev-psych-120709-145346
- Gładziejewski, P. (2016). Predictive coding and representationalism. *Synthese*, *193*(2), 559–582. https://doi.org/10.1007/s11229-015-0762-9
- Gläscher, J., Daw, N., Dayan, P., & O'Doherty, J. P. (2010a). States versus Rewards: Dissociable Neural Prediction Error Signals Underlying Model-Based and Model-Free Reinforcement Learning. *Neuron*, 66(4), 585–595. https://doi.org/10.1016/j.neuron.2010.04.016
- Gläscher, J., Daw, N., Dayan, P., & O'Doherty, J. P. (2010b). States versus Rewards: Dissociable Neural Prediction Error Signals Underlying Model-Based and Model-Free Reinforcement Learning. *Neuron*, 66(4), 585–595. https://doi.org/10.1016/j.neuron.2010.04.016
- Goel, V. (1992). Comparison of Well-Structured & Ill-Structured Task Environments and Problem Spaces.
- Goel, V., & Grafman, J. (1995). Are the frontal lobes implicated in "planning" functions? Interpreting data from the Tower of Hanoi. *Neuropsychologia*, 33(5), 623–642. https://doi.org/10.1016/0028-3932(95)90866-P
- Goel, V., Grafman, J., Sadato, N., & Hallett, M. (1995). *Modelling other minds*. https://doi.org/10.1097/00001756-199509000-00009.
- Goldman-Rakic, P. S., & Selemon, L. D. (1986). Topography of Corticostriatal Projections in Nonhuman Primates and Implications for Functional Parcellation of the Neostriatum. In E. G. Jones & A. Peters (Eds.), *Sensory-Motor Areas and Aspects of Cortical Connectivity* (Vol. 5, pp. 447–466). Springer US. https://doi.org/10.1007/978-1-4613-2149-1_12
- Gollwitzer, P. M. (1999). Implementation intentions: Strong effects of simple plans. *American Psychologist*, *54*(7), 493–503. https://doi.org/10.1037/0003-066x.54.7.493
- Goodroe, S. C., Starnes, J., & Brown, T. I. (2018a). The Complex Nature of Hippocampal-Striatal Interactions in Spatial Navigation. *Frontiers in Human Neuroscience*, 12, 250. https://doi.org/10.3389/fnhum.2018.00250
- Goodroe, S. C., Starnes, J., & Brown, T. I. (2018b). The Complex Nature of Hippocampal-Striatal Interactions in Spatial Navigation. *Frontiers in Human Neuroscience*, 12, 250. https://doi.org/10.3389/fnhum.2018.00250
- Gopnik, A., Glymour, C., Sobel, D. M., Schulz, L. E., Kushnir, T., & Danks, D. (2004). A Theory of Causal Learning in Children: Causal Maps and Bayes Nets. *Psychological Review*, 111(1), 3– 32. https://doi.org/10.1037/0033-295X.111.1.3
- Gordon, E. M., Chauvin, R. J., Van, A. N., Rajesh, A., Nielsen, A., Newbold, D. J., Lynch, C. J., Seider, N. A., Krimmel, S. R., Scheidter, K. M., Monk, J., Miller, R. L., Metoki, A., Montez, D. F., Zheng, A., Elbau, I., Madison, T., Nishino, T., Myers, M. J., ... Dosenbach, N. U. F. (2023). A somato-cognitive action network alternates with effector regions in motor cortex. *Nature*, *617*(7960), 351–359. https://doi.org/10.1038/s41586-023-05964-2
- Gordon, E. M., Laumann, T. O., Gilmore, A. W., Newbold, D. J., Greene, D. J., Berg, J. J., Ortega, M., Hoyt-Drazen, C., Gratton, C., Sun, H., Hampton, J. M., Coalson, R. S., Nguyen, A. L., McDermott, K. B., Shimony, J. S., Snyder, A. Z., Schlaggar, B. L., Petersen, S. E., Nelson, S. M., & Dosenbach, N. U. F. (2017). Precision Functional Mapping of Individual Human Brains. *Neuron*, *95*(4), 791-807.e7. https://doi.org/10.1016/j.neuron.2017.07.011
- Goulden, N., Khusnulina, A., Davis, N. J., Bracewell, R. M., Bokde, A. L., McNulty, J. P., & Mullins, P. G. (2014). The salience network is responsible for switching between the default mode network and the central executive network: Replication from DCM. *NeuroImage*, 99, 180–190. https://doi.org/10.1016/j.neuroimage.2014.05.052
- Graff-Radford, J., Williams, L., Jones, D. T., & Benarroch, E. E. (2017). Caudate nucleus as a component of networks controlling behavior. *Neurology*, 89(21), 2192–2197. https://doi.org/10.1212/WNL.000000000004680
- Graybiel, A. M., & Grafton, S. T. (2015). The Striatum: Where Skills and Habits Meet. *Cold Spring Harbor Perspectives in Biology*, 7(8), a021691. https://doi.org/10.1101/cshperspect.a021691
- Graziano, & Gross, CharlesG. (1993). A bimodal map of space: Somatosensory receptive fields in the macaque putamen with corresponding visual receptive fields. *Experimental Brain Research*, 97(1). https://doi.org/10.1007/BF00228820
- Graziano, M. S. A. (2016). Ethological Action Maps: A Paradigm Shift for the Motor Cortex. *Trends in Cognitive Sciences*, 20(2), 121–132. https://doi.org/10.1016/j.tics.2015.10.008
- Graziano, M. S. A., Reiss, L. A. J., & Gross, C. G. (1999). A neuronal representation of the location of nearby sounds. *Nature*, 397(6718), 428–430. https://doi.org/10.1038/17115
- Grefkes, C., & Fink, G. R. (2005). REVIEW: The functional organization of the intraparietal sulcus in humans and monkeys: Intraparietal sulcus in monkeys and humans, C. Grefkes and G. R. Fink. *Journal of Anatomy*, 207(1), 3–17. https://doi.org/10.1111/j.1469-7580.2005.00426.x
- Gregory, R. L. (1968). Perceptual illusions and brain models. Proceedings of the Royal Society of London. Series B. Biological Sciences, 171(1024), 279–296. https://doi.org/10.1098/rspb.1968.0071
- Gregory, R. L. (1980). Perceptions as hypotheses. *Philosophical Transactions of the Royal Society of London. B, Biological Sciences*, 290(1038), 181–197. https://doi.org/10.1098/rstb.1980.0090
- Gremel, C. M., & Costa, R. M. (2013). Orbitofrontal and striatal circuits dynamically encode the shift between goal-directed and habitual actions. *Nature Communications*, 4(1), 2264. https://doi.org/10.1038/ncomms3264
- Grill-Spector, K., & Weiner, K. S. (2014). The functional architecture of the ventral temporal cortex and its role in categorization. *Nature Reviews Neuroscience*, 15(8), 536–548. https://doi.org/10.1038/nrn3747
- Gu, Q., Zou, L., Loprinzi, P. D., Quan, M., & Huang, T. (2019). Effects of Open Versus Closed Skill Exercise on Cognitive Function: A Systematic Review. *Frontiers in Psychology*, 10, 1707. https://doi.org/10.3389/fpsyg.2019.01707
- Guipponi, O., Wardak, C., Ibarrola, D., Comte, J.-C., Sappey-Marinier, D., Pinede, S., & Ben Hamed, S. (2013). Multimodal Convergence within the Intraparietal Sulcus of the Macaque Monkey. *Journal of Neuroscience*, 33(9), 4128–4139. https://doi.org/10.1523/JNEUROSCI.1421-12.2013
- Guterstam, A., Björnsdotter, M., Gentile, G., & Ehrsson, H. H. (2015). Posterior Cingulate Cortex Integrates the Senses of Self-Location and Body Ownership. *Current Biology*, 25(11), 1416– 1425. https://doi.org/10.1016/j.cub.2015.03.059
- Guterstam, A., Szczotka, J., Zeberg, H., & Ehrsson, H. H. (2018). Tool use changes the spatial extension of the magnetic touch illusion. *Journal of Experimental Psychology: General*, 147(2), 298–303. https://doi.org/10.1037/xge0000390
- Haber, S. N. (2003). The primate basal ganglia: Parallel and integrative networks. *Journal of Chemical Neuroanatomy*, *26*(4), 317–330. https://doi.org/10.1016/j.jchemneu.2003.10.003
- Haber, S. N. (2016). Corticostriatal circuitry. *Dialogues in Clinical Neuroscience*, 18(1), 7–21. https://doi.org/10.31887/DCNS.2016.18.1/shaber
- Hafting, T., Fyhn, M., Molden, S., Moser, M.-B., & Moser, E. I. (2005). Microstructure of a spatial map in the entorhinal cortex. *Nature*, 436(7052), 801–806. https://doi.org/10.1038/nature03721
- Haggard, P., Clark, S., & Kalogeras, J. (2002). Voluntary action and conscious awareness. *Nature Neuroscience*, 5(4), 382–385. https://doi.org/10.1038/nn827
- Hansen, N. (2017). The Longevity of Hippocampus-Dependent Memory Is Orchestrated by the Locus Coeruleus-Noradrenergic System. *Neural Plasticity*, 2017, 1–9. https://doi.org/10.1155/2017/2727602
- Hartley, T., Maguire, E. A., Spiers, H. J., & Burgess, N. (2003). The well-worn route and the path less traveled: Distinct neural bases of route following and wayfinding in humans. *Neuron*, 37(5), 877–888. https://doi.org/10.1016/S0896-6273(03)00095-3
- Hartnett, A. T., Schertzer, E., Levin, S. A., & Couzin, I. D. (2016). Heterogeneous Preference and Local Nonlinearity in Consensus Decision Making. *Physical Review Letters*, 116(3), 038701. https://doi.org/10.1103/PhysRevLett.116.038701
- Hassabis, D., & Maguire, E. A. (2007a). Deconstructing episodic memory with construction. *Trends in Cognitive Sciences*, 11(7), 299–306. https://doi.org/10.1016/j.tics.2007.05.001
- Hassabis, D., & Maguire, E. A. (2007b). Deconstructing episodic memory with construction. Trends in Cognitive Sciences, 11(7), 299–306. https://doi.org/10.1016/j.tics.2007.05.001
- Hayama, H. R., Vilberg, K. L., & Rugg, M. D. (2012). Overlap between the Neural Correlates of Cued Recall and Source Memory: Evidence for a Generic Recollection Network? *Journal of Cognitive Neuroscience*, 24(5), 1127–1137. https://doi.org/10.1162/jocn a 00202
- Heft, H. (2001). Ecological Psychology in Context: James Gibson, Roger Barker, and the Legacy of William James's Radical Empiricism Resources for Ecological Psychology. Lawrence Erlbaum Associates, Inc.
- Hein, G., & Knight, R. T. (2008). Superior Temporal Sulcus—It's My Area: Or Is It? Journal of Cognitive Neuroscience, 20(12), 2125–2136. https://doi.org/10.1162/jocn.2008.20148
- Helbing, J., Draschkow, D., & L.-H. Võ, M. (2022). Auxiliary Scene-Context Information Provided by Anchor Objects Guides Attention and Locomotion in Natural Search Behavior. *Psychological Science*, 33(9), 1463–1476. https://doi.org/10.1177/09567976221091838
- Henke, K. (2010). A model for memory systems based on processing modes rather than consciousness. *Nature Reviews Neuroscience*, 11(7), 523–532. https://doi.org/10.1038/nrn2850

- Heras-Escribano, M., & De Pinedo-García, M. (2018). Affordances and Landscapes: Overcoming the Nature–Culture Dichotomy through Niche Construction Theory. *Frontiers in Psychology*, 8, 2294. https://doi.org/10.3389/fpsyg.2017.02294
- Herath, P., Kinomura, S., & Roland, P. E. (2001). Visual recognition: Evidence for two distinctive mechanisms from a PET study. *Human Brain Mapping*, 12(2), 110–119. https://doi.org/10.1002/1097-0193(200102)12:2<110::AID-HBM1008>3.0.CO;2-0
- Hicks, T. C., Kühl, H. S., Boesch, C., Dieguez, P., Ayimisin, A. E., Fernandez, R. M., Zungawa, D. B., Kambere, M., Swinkels, J., Menken, S. B. J., Hart, J., Mundry, R., & Roessingh, P. (2019). Bili-Uéré: A Chimpanzee Behavioural Realm in Northern Democratic Republic of Congo. *Folia Primatologica*, 90(1), 3–64. https://doi.org/10.1159/000492998
- Hikosaka, O., Nakahara, H., Rand, M. K., Sakai, K., Lu, X., Nakamura, K., Miyachi, S., & Doya, K. (1999). Parallel neural networks for learning sequential procedures. *Trends in Neurosciences*, 22(10), 464–471. https://doi.org/10.1016/S0166-2236(99)01439-3
- Hikosaka, O., Nakamura, K., Sakai, K., & Nakahara, H. (2002). Central mechanisms of motor skill learning. *Current Opinion in Neurobiology*, 12(2), 217–222. https://doi.org/10.1016/S0959-4388(02)00307-0
- Hohwy, J. (2015). The Neural Organ Explains the Mind. In T. Metzinger & J. M. Windt (Eds.), *Open MIND*, 2-vol. Set. The MIT Press. https://doi.org/10.7551/mitpress/10603.003.0058
- Hohwy, J. (2016). The Self-Evidencing Brain: The Self-Evidencing Brain. *Noûs*, 50(2), 259–285. https://doi.org/10.1111/nous.12062
- Hohwy, J. (2017). How to Entrain Your Evil Demon. *Philosophy and Predictive Processing*. https://doi.org/10.15502/9783958573048
- Hohwy, J. (2020). New directions in predictive processing. *Mind & Language*, 35(2), 209–223. https://doi.org/10.1111/mila.12281
- Hok, V., Save, E., Lenck-Santini, P. P., & Poucet, B. (2005). Coding for spatial goals in the prelimbic/infralimbic area of the rat frontal cortex. *Proceedings of the National Academy of Sciences*, 102(12), 4602–4607. https://doi.org/10.1073/pnas.0407332102
- Honey, C. J., Thompson, C. R., Lerner, Y., & Hasson, U. (2012). Not Lost in Translation: Neural Responses Shared Across Languages. *The Journal of Neuroscience*, 32(44), 15277–15283. https://doi.org/10.1523/JNEUROSCI.1800-12.2012
- Hooks, B. M., Papale, A. E., Paletzki, R. F., Feroze, M. W., Eastwood, B. S., Couey, J. J., Winnubst, J., Chandrashekar, J., & Gerfen, C. R. (2018). Topographic precision in sensory and motor corticostriatal projections varies across cell type and cortical area. *Nature Communications*, 9(1), 3549. https://doi.org/10.1038/s41467-018-05780-7
- Høydal, Ø. A., Skytøen, E. R., Andersson, S. O., Moser, M.-B., & Moser, E. I. (2019). Object-vector coding in the medial entorhinal cortex. *Nature*, 568(7752), 400–404. https://doi.org/10.1038/s41586-019-1077-7
- Hsiao, A. T., & Reber, A. S. (2001). The dual-task SRT procedure: Fine-tuning the timing. *Psychonomic Bulletin & Review*, 8(2), 336–342. https://doi.org/10.3758/BF03196170
- Huijbers, W., Pennartz, C. M. A., Cabeza, R., & Daselaar, S. M. (2011). The Hippocampus Is Coupled with the Default Network during Memory Retrieval but Not during Memory Encoding. *PLoS ONE*, 6(4), e17463. https://doi.org/10.1371/journal.pone.0017463
- Hutchins, E. (2000). Distributed cognition. *International Encyclopedia of the Social and Behavioral Sciences. Elsevier Science*, 138, 1–10.
- Hutchinson, G. E. (1957). Concluding Remarks.
- Hyva^{*}rinen, J. (1981). Regional distribution of functions in parietal association area 7 of the monkey. Brain Research, 206(2), 287–303. https://doi.org/10.1016/0006-8993(81)90533-3
- Iaria, G., Petrides, M., Dagher, A., Pike, B., & Bohbot, V. D. (2003). Cognitive Strategies Dependent on the Hippocampus and Caudate Nucleus in Human Navigation: Variability and Change with Practice. *The Journal of Neuroscience*, 23(13), 5945–5952. https://doi.org/10.1523/JNEUROSCI.23-13-05945.2003
- Igelström, K. M., & Graziano, M. S. A. (2017). The inferior parietal lobule and temporoparietal junction: A network perspective. *Neuropsychologia*, *105*, 70–83. https://doi.org/10.1016/j.neuropsychologia.2017.01.001
- Iriki, A., & Taoka, M. (2012). Triadic (ecological, neural, cognitive) niche construction: A scenario of human brain evolution extrapolating tool use and language from the control of reaching actions. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 367(1585), 10–23. https://doi.org/10.1098/rstb.2011.0190

- Ishai, A., Ungerleider, L. G., Martin, A., Schouten, J. L., & Haxby, J. V. (1999). Distributed representation of objects in the human ventral visual pathway. *Proceedings of the National Academy of Sciences*, 96(16), 9379–9384. https://doi.org/10.1073/pnas.96.16.9379
- James, W. (2007). The Principles of Psychology. THE PRINCIPLES OF PSYCHOLOGY.
- Jensen, J. L. (2005). The puzzles of motor development: How the study of developmental biomechanics contributes to the puzzle solutions. *Infant and Child Development*, 14(5), 501–511. https://doi.org/10.1002/icd.425
- Jensen, J. L., & Korff, T. (2005). Continuing the discourse on the contribution of biomechanics to understanding motor development: Response to the commentaries. *Infant and Child Development*, 14(5), 529–533. https://doi.org/10.1002/icd.421
- Johnson, A., & Redish, A. D. (2005). Hippocampal replay contributes to within session learning in a temporal difference reinforcement learning model. *Neural Networks*, 18(9), 1163–1171. https://doi.org/10.1016/j.neunet.2005.08.009
- Johnson, M. R., & Johnson, M. K. (2014). Decoding individual natural scene representations during perception and imagery. *Frontiers in Human Neuroscience*, 8. https://doi.org/10.3389/fnhum.2014.00059
- Kabadayi, C., Bobrowicz, K., & Osvath, M. (2018). The detour paradigm in animal cognition. *Animal* Cognition, 21(1), 21–35. https://doi.org/10.1007/s10071-017-1152-0
- Kahneman, D. (2013). Thinking, fast and slow.
- Kahneman, D., & Riis, J. (2005). Living, and thinking about it: Two perspectives on life. In F. A. Huppert, N. Baylis, & B. Keverne (Eds.), *The Science of Well-Being* (pp. 284–305). Oxford University Press. https://doi.org/10.1093/acprof:oso/9780198567523.003.0011
- Kamin, L. J. (1969). Selective association and conditioning. *Fundamental Issues in Associative Learning*, 42–64.
- Kamiński, J., Mamelak, A. N., Birch, K., Mosher, C. P., Tagliati, M., & Rutishauser, U. (2018). Novelty-Sensitive Dopaminergic Neurons in the Human Substantia Nigra Predict Success of Declarative Memory Formation. *Current Biology*, 28(9), 1333-1343.e4. https://doi.org/10.1016/j.cub.2018.03.024
- Kee, Y. H. (2019). Reflections on athletes' mindfulness skills development: Fitts and Posner's (1967) three stages of learning. *Journal of Sport Psychology in Action*, 10(4), 214–219. https://doi.org/10.1080/21520704.2018.1549640
- Keele, S. W., Ivry, R., Mayr, U., Hazeltine, E., & Heuer, H. (2003). The cognitive and neural architecture of sequence representation. *Psychological Review*, 110(2), 316–339. https://doi.org/10.1037/0033-295X.110.2.316
- Kelly, S. C., He, B., Perez, S. E., Ginsberg, S. D., Mufson, E. J., & Counts, S. E. (2017). Locus coeruleus cellular and molecular pathology during the progression of Alzheimer's disease. *Acta Neuropathologica Communications*, 5(1), 8. https://doi.org/10.1186/s40478-017-0411-2
- Kempadoo, K. A., Mosharov, E. V., Choi, S. J., Sulzer, D., & Kandel, E. R. (2016). Dopamine release from the locus coeruleus to the dorsal hippocampus promotes spatial learning and memory. *Proceedings of the National Academy of Sciences*, 113(51), 14835–14840. https://doi.org/10.1073/pnas.1616515114
- Kent, M., Platt, S. R., & Schatzberg, S. J. (2010). The neurology of balance: Function and dysfunction of the vestibular system in dogs and cats. *The Veterinary Journal*, 185(3), 247–258. https://doi.org/10.1016/j.tvj1.2009.10.029
- Keogh, J. W. L., & Hume, P. A. (2012). Evidence for biomechanics and motor learning research improving golf performance. *Sports Biomechanics*, 11(2), 288–309. https://doi.org/10.1080/14763141.2012.671354
- Kermadi, I., Jurquet, Y., Arzi, M., & Joseph, J. P. (1993). Neural activity in the caudate nucleus of monkeys during spatial sequencing. *Experimental Brain Research*, 94(2). https://doi.org/10.1007/BF00230305
- Kharkhurin, A. V. (2014). Creativity.4in1: Four-Criterion Construct of Creativity. *Creativity Research Journal*, *26*(3), 338–352. https://doi.org/10.1080/10400419.2014.929424
- Kiefer, A. B. (2020). Psychophysical identity and free energy. *Journal of The Royal Society Interface*, 17(169), 20200370. https://doi.org/10.1098/rsif.2020.0370
- Kikuchi, M., Takahashi, T., Hirosawa, T., Oboshi, Y., Yoshikawa, E., Minabe, Y., & Ouchi, Y. (2017). The Lateral Occipito-temporal Cortex Is Involved in the Mental Manipulation of Body Part Imagery. *Frontiers in Human Neuroscience*, 11. https://doi.org/10.3389/fnhum.2017.00181
- Killcross, S. (2003). Coordination of Actions and Habits in the Medial Prefrontal Cortex of Rats. *Cerebral Cortex*, 13(4), 400–408. https://doi.org/10.1093/cercor/13.4.400

- Kiran, B. R., Sobh, I., Talpaert, V., Mannion, P., Sallab, A. A. A., Yogamani, S., & Perez, P. (2022). Deep Reinforcement Learning for Autonomous Driving: A Survey. *IEEE Transactions on Intelligent Transportation Systems*, 23(6), 4909–4926. https://doi.org/10.1109/TITS.2021.3054625
- Kirchhoff, M. D., & Kiverstein, J. (2019). Extended Consciousness and Predictive Processing: A Third-Wave View (1st ed.). Routledge. https://doi.org/10.4324/9781315150420
- Kirchhoff, M. D., & Kiverstein, J. (2021). How to determine the boundaries of the mind: A Markov blanket proposal. *Synthese*, *198*(5), 4791–4810. https://doi.org/10.1007/s11229-019-02370-y
- Kitamura, T., Ogawa, S. K., Roy, D. S., Okuyama, T., Morrissey, M. D., Smith, L. M., Redondo, R. L., & Tonegawa, S. (2017). Engrams and circuits crucial for systems consolidation of a memory. *Science*, 356(6333), 73–78. https://doi.org/10.1126/science.aam6808
- Knill, D. C., & Pouget, A. (2004). The Bayesian brain: The role of uncertainty in neural coding and computation. *Trends in Neurosciences*, 27(12), 712–719. https://doi.org/10.1016/j.tins.2004.10.007
- Knowlton, B. J., Ramus, S. J., & Squire, L. R. (1992). Intact artificial grammar learning in amnesia: Dissociation of classification learning and explicit memory for specific instances. *Psychological Science*, 3(3), 172–179.
- Knudsen, E. B., & Wallis, J. D. (2020). Closed-Loop Theta Stimulation in the Orbitofrontal Cortex Prevents Reward-Based Learning. *Neuron*, 106(3), 537-547.e4. https://doi.org/10.1016/j.neuron.2020.02.003
- Knudson, D. (2021). Fundamentals of Biomechanics. Springer International Publishing. https://doi.org/10.1007/978-3-030-51838-7
- Koch, C., Massimini, M., Boly, M., & Tononi, G. (2016). Neural correlates of consciousness: Progress and problems. *Nature Reviews Neuroscience*, 17(5), 307–321. https://doi.org/10.1038/nrn.2016.22
- Koechlin, E., & Summerfield, C. (2007). An information theoretical approach to prefrontal executive function. *Trends in Cognitive Sciences*, 11(6), 229–235. https://doi.org/10.1016/j.tics.2007.04.005
- Korkiakangas, T., Dindar, K., Laitila, A., & Kärnä, E. (2016). The Sally-Anne test: An interactional analysis of a dyadic assessment: Sally-Anne test: an interactional analysis. *International Journal* of Language & Communication Disorders, 51(6), 685–702. https://doi.org/10.1111/1460-6984.12240
- Kornblith, S., Quian Quiroga, R., Koch, C., Fried, I., & Mormann, F. (2017). Persistent Single-Neuron Activity during Working Memory in the Human Medial Temporal Lobe. *Current Biology*, 27(7), 1026–1032. https://doi.org/10.1016/j.cub.2017.02.013
- Krall, S. C., Rottschy, C., Oberwelland, E., Bzdok, D., Fox, P. T., Eickhoff, S. B., Fink, G. R., & Konrad, K. (2015). The role of the right temporoparietal junction in attention and social interaction as revealed by ALE meta-analysis. *Brain Structure and Function*, 220(2), 587–604. https://doi.org/10.1007/s00429-014-0803-z
- Kravitz, D. J., Saleem, K. S., Baker, C. I., Ungerleider, L. G., & Mishkin, M. (2013). The ventral visual pathway: An expanded neural framework for the processing of object quality. *Trends in Cognitive Sciences*, 17(1), 26–49. https://doi.org/10.1016/j.tics.2012.10.011
- Kropff, E., Carmichael, J. E., Moser, M.-B., & Moser, E. I. (2015). Speed cells in the medial entorhinal cortex. *Nature*, 523(7561), 419–424. https://doi.org/10.1038/nature14622
- Kubit, B., & Jack, A. I. (2013). Rethinking the role of the rTPJ in attention and social cognition in light of the opposing domains hypothesis: Findings from an ALE-based meta-analysis and restingstate functional connectivity. *Frontiers in Human Neuroscience*, 7. https://doi.org/10.3389/fnhum.2013.00323
- Kunishio, K., & Haber, S. N. (1994). Primate cingulostriatal projection: Limbic striatal versus sensorimotor striatal input. *The Journal of Comparative Neurology*, 350(3), 337–356. https://doi.org/10.1002/cne.903500302
- Künzle, H. (1978). An autoradiographic analysis of the efferent connections from premotor and adjacent prefrontal regions (areas 6 and 9) in Macaca fascicularis; pp. 210–234. *Brain, Behavior and Evolution, 15*(3), 210–234.
- Làdavas, E. (2002). Functional and dynamic properties of visual peripersonal space. 6.
- Laland, K. N., Uller, T., Feldman, M. W., Sterelny, K., Müller, G. B., Moczek, A., Jablonka, E., & Odling-Smee, J. (2015). The extended evolutionary synthesis: Its structure, assumptions and predictions. *Proceedings of the Royal Society B: Biological Sciences*, 282(1813), 20151019. https://doi.org/10.1098/rspb.2015.1019

- Lanciego, J. L., Luquin, N., & Obeso, J. A. (2012). Functional Neuroanatomy of the Basal Ganglia. Cold Spring Harbor Perspectives in Medicine, 2(12), a009621–a009621. https://doi.org/10.1101/cshperspect.a009621
- Landi, S. M., Viswanathan, P., Serene, S., & Freiwald, W. A. (2021). A fast link between face perception and memory in the temporal pole. *Science*, 373(6554), 581–585. https://doi.org/10.1126/science.abi6671
- Latash, M. L., & Zatsiorsky, V. M. (2016). *Biomechanics and motor control: Defining central concepts*. Elsevier/AP, Academic Press is an imprint of Elsevier.
- Lauer, T., & Võ, M. L.-H. (2022). The Ingredients of Scenes that Affect Object Search and Perception. In B. Ionescu, W. A. Bainbridge, & N. Murray (Eds.), *Human Perception of Visual Information* (pp. 1–32). Springer International Publishing. https://doi.org/10.1007/978-3-030-81465-6 1
- Lee, S., Parthasarathi, T., & Kable, J. W. (2021). The Ventral and Dorsal Default Mode Networks Are Dissociably Modulated by the Vividness and Valence of Imagined Events. *The Journal of Neuroscience*, 41(24), 5243–5250. https://doi.org/10.1523/JNEUROSCI.1273-20.2021
- Leech, R., & Sharp, D. J. (2014). The role of the posterior cingulate cortex in cognition and disease. Brain, 137(1), 12–32. https://doi.org/10.1093/brain/awt162
- Lehéricy, S., Benali, H., Van de Moortele, P.-F., Pélégrini-Issac, M., Waechter, T., Ugurbil, K., & Doyon, J. (2005). Distinct basal ganglia territories are engaged in early and advanced motor sequence learning. *Proceedings of the National Academy of Sciences*, 102(35), 12566–12571. https://doi.org/10.1073/pnas.0502762102
- LePort, A. K. R., Mattfeld, A. T., Dickinson-Anson, H., Fallon, J. H., Stark, C. E. L., Kruggel, F., Cahill, L., & McGaugh, J. L. (2012). Behavioral and neuroanatomical investigation of Highly Superior Autobiographical Memory (HSAM). *Neurobiology of Learning and Memory*, 98(1), 78–92. https://doi.org/10.1016/j.nlm.2012.05.002
- Lever, C., Burton, S., Jeewajee, A., O'Keefe, J., & Burgess, N. (2009). Boundary Vector Cells in the Subiculum of the Hippocampal Formation. *The Journal of Neuroscience*, 29(31), 9771–9777. https://doi.org/10.1523/JNEUROSCI.1319-09.2009
- Levin, M. (2019). The Computational Boundary of a "Self": Developmental Bioelectricity Drives Multicellularity and Scale-Free Cognition. *Frontiers in Psychology*, 10, 2688. https://doi.org/10.3389/fpsyg.2019.02688
- Levin, M. (2023). Collective intelligence of morphogenesis as a teleonomic process. In *Evolution "On Purpose": Teleonomy in living systems*. The MIT Press. https://doi.org/10.7551/mitpress/14642.003.0013
- Lewis, P. A., Knoblich, G., & Poe, G. (2018). How Memory Replay in Sleep Boosts Creative Problem-Solving. *Trends in Cognitive Sciences*, 22(6), 491–503. https://doi.org/10.1016/j.tics.2018.03.009
- Lewontin, R. C. (2000). The Triple Helix. Harvard University Press.
- Li, W., Mai, X., & Liu, C. (2014). The default mode network and social understanding of others: What do brain connectivity studies tell us. *Frontiers in Human Neuroscience*, 8. https://doi.org/10.3389/fnhum.2014.00074
- Libet, B. (1985). Unconscious cerebral initiative and the role of conscious will in voluntary action. *THE BEHAVIORAL AND BRAIN SCIENCES*. https://doi.org/10.1017/S0140525X00044903
- Libet, B., Gleason, C. A., & Wright, E. W. (1993). *Time of conscious intention to act in relation to onset of cerebral activity (readiness-potential)*.
- Lieberman, M. D. (2022). Seeing Minds, Matter, and Meaning: The CEEing Model of Pre-Reflective Subjective Construal. *Psychological Review*, 157.
- Lin, D. C., McGowan, C. P., Blum, K. P., & Ting, L. H. (2019). Yank: The time derivative of force is an important biomechanical variable in sensorimotor systems. *Journal of Experimental Biology*, 222(18), jeb180414. https://doi.org/10.1242/jeb.180414
- Lindsay, G. W. (2021). Convolutional Neural Networks as a Model of the Visual System: Past, Present, and Future. *Journal of Cognitive Neuroscience*, 33(10), 2017–2031. https://doi.org/10.1162/jocn_a_01544
- Lindskog, M., Nyström, P., & Gredebäck, G. (2021). Can the Brain Build Probability Distributions? Frontiers in Psychology, 12, 596231. https://doi.org/10.3389/fpsyg.2021.596231
- Lisman, J. E., & Grace, A. A. (2005). The Hippocampal-VTA Loop: Controlling the Entry of Information into Long-Term Memory. *Neuron*, 46(5), 703–713. https://doi.org/10.1016/j.neuron.2005.05.002
- Longo, G., & Montévil, M. (2014). Perspectives on Organisms. Springer Berlin Heidelberg. https://doi.org/10.1007/978-3-642-35938-5

- Longo, G., Montévil, M., & Kauffman, S. (2012). No entailing laws, but enablement in the evolution of the biosphere. Proceedings of the Fourteenth International Conference on Genetic and Evolutionary Computation Conference Companion - GECCO Companion '12, 1379. https://doi.org/10.1145/2330784.2330946
- Loonen, A. J. M., & Ivanova, S. A. (2015). Circuits regulating pleasure and happiness: The evolution of reward-seeking and misery-fleeing behavioral mechanisms in vertebrates. *Frontiers in Neuroscience*, 9. https://doi.org/10.3389/fnins.2015.00394
- Lopez, C. (2015). Vestibular Cortex. In *Brain Mapping* (pp. 305–312). Elsevier. https://doi.org/10.1016/B978-0-12-397025-1.00227-X
- Lopez, C., & Blanke, O. (2011). The thalamocortical vestibular system in animals and humans. Brain Research Reviews, 67(1–2), 119–146. https://doi.org/10.1016/j.brainresrev.2010.12.002
- Lorenz, K. Z. (1981). The Foundations of Ethology. Springer Vienna. https://doi.org/10.1007/978-3-7091-3671-3
- Machery, E. (2009). Doing without concepts. Oxford University Press.
- Machery, E. (2016). The amodal brain and the offloading hypothesis. *Psychonomic Bulletin & Review*, 23(4), 1090–1095. https://doi.org/10.3758/s13423-015-0878-4
- Maguire, E. A., Intraub, H., & Mullally, S. L. (2016). Scenes, spaces, and memory traces: What does the hippocampus do? *The Neuroscientist*, 22(5), 432–439.
- Maguire, E. A., & Mullally, S. L. (2013). The hippocampus: A manifesto for change. *Journal of Experimental Psychology: General*, *142*(4), 1180–1189. https://doi.org/10.1037/a0033650
- Mancuso, L., Cavuoti-Cabanillas, S., Liloia, D., Manuello, J., Buzi, G., Cauda, F., & Costa, T. (2022). Tasks activating the default mode network map multiple functional systems. *Brain Structure* and Function, 227(5), 1711–1734. https://doi.org/10.1007/s00429-022-02467-0
- Maravita, A., & Iriki, A. (2004). Tools for the body (schema). *Trends in Cognitive Sciences*, 8(2), 79–86. https://doi.org/10.1016/j.tics.2003.12.008
- Marchette, S. A., Bakker, A., & Shelton, A. L. (2011). Cognitive Mappers to Creatures of Habit: Differential Engagement of Place and Response Learning Mechanisms Predicts Human Navigational Behavior. *The Journal of Neuroscience*, 31(43), 15264–15268. https://doi.org/10.1523/JNEUROSCI.3634-11.2011
- Markman, K. D., Klein, W. M. P., & Suhr, J. A. (Eds.). (2009). *Handbook of imagination and mental simulation*. Psychology Press.
- Martin, A. K., Kessler, K., Cooke, S., Huang, J., & Meinzer, M. (2020). The Right Temporoparietal Junction Is Causally Associated with Embodied Perspective-taking. *The Journal of Neuroscience*, 40(15), 3089–3095. https://doi.org/10.1523/JNEUROSCI.2637-19.2020
- Mason, R. A., & Just, M. A. (2020). Neural Representations of Procedural Knowledge. Psychological Science, 31(6), 729–740. https://doi.org/10.1177/0956797620916806
- Matthews, G., Deary, I. J., & Whiteman, M. C. (2003). Personality traits. Cambridge University Press.
- McCutcheon, R. A., Abi-Dargham, A., & Howes, O. D. (2019). Schizophrenia, Dopamine and the Striatum: From Biology to Symptoms. *Trends in Neurosciences*, *42*(3), 205–220. https://doi.org/10.1016/j.tins.2018.12.004
- McCutcheon, R. A., Brown, K., Nour, M. M., Smith, S. M., Veronese, M., Zelaya, F., Osugo, M., Jauhar, S., Hallett, W., Mehta, M. M., & Howes, O. D. (2021). Dopaminergic organization of striatum is linked to cortical activity and brain expression of genes associated with psychiatric illness. *Science Advances*, 7(24), eabg1512. https://doi.org/10.1126/sciadv.abg1512
- McGlone, F., & Reilly, D. (2010). The cutaneous sensory system. Neuroscience & Biobehavioral Reviews, 34(2), 148–159. https://doi.org/10.1016/j.neubiorev.2009.08.004
- McLester, J., & St. Pierre, P. (2008). *Applied biomechanics: Concepts and connections*. Thompson/Wadsworth.
- McNamara, C. G., & Dupret, D. (2017). Two sources of dopamine for the hippocampus. *Trends in Neurosciences*, 40(7), 383–384. https://doi.org/10.1016/j.tins.2017.05.005
- Mecacci, L. (2013). Solomon V. Shereshevsky: The great Russian mnemonist. *Cortex*, 49(8), 2260–2263. https://doi.org/10.1016/j.cortex.2013.05.007
- Mele, A. R. (Author). (1992). Springs of Action: Understanding Intentional Behavior. 285.
- Menegas, W., Akiti, K., Amo, R., Uchida, N., & Watabe-Uchida, M. (2018). Dopamine neurons projecting to the posterior striatum reinforce avoidance of threatening stimuli. *Nature Neuroscience*, 21(10), 1421–1430. https://doi.org/10.1038/s41593-018-0222-1
- Menegas, W., Babayan, B. M., Uchida, N., & Watabe-Uchida, M. (2017). Opposite initialization to novel cues in dopamine signaling in ventral and posterior striatum in mice. *eLife*, 6, e21886. https://doi.org/10.7554/eLife.21886

- Menegas, W., Bergan, J. F., Ogawa, S. K., Isogai, Y., Umadevi Venkataraju, K., Osten, P., Uchida, N., & Watabe-Uchida, M. (2015). Dopamine neurons projecting to the posterior striatum form an anatomically distinct subclass. *eLife*, 4, e10032. https://doi.org/10.7554/eLife.10032
- Mergner, T., & Rosemeier, T. (1998). Interaction of vestibular, somatosensory and visual signals for postural control and motion perception under terrestrial and microgravity conditions—A conceptual model. *Brain Research Reviews*, 28(1–2), 118–135. https://doi.org/10.1016/S0165-0173(98)00032-0
- Michaels, C. F. (2000). Information, Perception, and Action: What Should Ecological Psychologists Learn From Milner and Goodale (1995)? *Ecological Psychology*, 12(3), 241–258. https://doi.org/10.1207/S15326969ECO1203_4
- Milich, R., Balentine, A. C., & Lynam, D. R. (2001). ADHD combined type and ADHD predominantly inattentive type are distinct and unrelated disorders. *Clinical Psychology: Science and Practice*, 8(4), 463–488. https://doi.org/10.1093/clipsy.8.4.463
- Milner, A. D. (2017). How do the two visual streams interact with each other? *Experimental Brain Research*, 235(5), 1297–1308. https://doi.org/10.1007/s00221-017-4917-4
- Milton, F., Fulford, J., Dance, C., Gaddum, J., Heuerman-Williamson, B., Jones, K., Knight, K. F., MacKisack, M., Winlove, C., & Zeman, A. (2021). Behavioral and Neural Signatures of Visual Imagery Vividness Extremes: Aphantasia versus Hyperphantasia. *Cerebral Cortex Communications*, 2(2), tgab035. https://doi.org/10.1093/texcom/tgab035
- Minsky, M. (1988). Society of mind. Simon and Schuster.
- Mızrak, E., Bouffard, N. R., Libby, L. A., Boorman, E. D., & Ranganath, C. (2021). The hippocampus and orbitofrontal cortex jointly represent task structure during memory-guided decision making. *Cell Reports*, 37(9), 110065. https://doi.org/10.1016/j.celrep.2021.110065
- Montagrin, A., Saiote, C., & Schiller, D. (2018). The social hippocampus. *Hippocampus*, 28(9), 672–679. https://doi.org/10.1002/hipo.22797
- Moore, J. W., & Obhi, S. S. (2012). Intentional binding and the sense of agency: A review. *Consciousness and Cognition*, 21(1), 546–561. https://doi.org/10.1016/j.concog.2011.12.002
- Moser, E. I., Kropff, E., & Moser, M.-B. (2008). Place Cells, Grid Cells, and the Brain's Spatial Representation System. *Annual Review of Neuroscience*, *31*(1), 69–89. https://doi.org/10.1146/annurev.neuro.31.061307.090723
- Moser, E. I., Moser, M.-B., & McNaughton, B. L. (2017). Spatial representation in the hippocampal formation: A history. *Nature Neuroscience*, 20(11), 1448–1464. https://doi.org/10.1038/nn.4653
- Mugler, E. M., Tate, M. C., Livescu, K., Templer, J. W., Goldrick, M. A., & Slutzky, M. W. (2018). Differential Representation of Articulatory Gestures and Phonemes in Precentral and Inferior Frontal Gyri. *The Journal of Neuroscience*, 38(46), 9803–9813. https://doi.org/10.1523/JNEUROSCI.1206-18.2018
- Mullally, S. L., & Maguire, E. A. (2014). Memory, imagination, and predicting the future: A common brain mechanism? *The Neuroscientist*, 20(3), 220–234. https://doi.org/10.1177/1073858413495091
- Muller, R. U., Stead, M., & Pach, J. (1996). The hippocampus as a cognitive graph. Journal of General Physiology, 107(6), 663–694. https://doi.org/10.1085/jgp.107.6.663
- Murphy, C., Jefferies, E., Rueschemeyer, S.-A., Sormaz, M., Wang, H., Margulies, D. S., & Smallwood, J. (2018). Distant from input: Evidence of regions within the default mode network supporting perceptually-decoupled and conceptually-guided cognition. *NeuroImage*, 171, 393–401. https://doi.org/10.1016/j.neuroimage.2018.01.017
- Murphy, G. L., & Medin, D. L. (1985). The Role of Theories in Conceptual Coherence.
- Nachev, P., Kennard, C., & Husain, M. (2008). Functional role of the supplementary and presupplementary motor areas. *Nature Reviews Neuroscience*, 9(11), 856–869. https://doi.org/10.1038/nrn2478
- Nackaerts, E., D'Cruz, N., Dijkstra, B. W., Gilat, M., Kramer, T., & Nieuwboer, A. (2019). Towards understanding neural network signatures of motor skill learning in Parkinson's disease and healthy aging. *The British Journal of Radiology*, 92(1101), 20190071. https://doi.org/10.1259/bjr.20190071
- Nagaratnam, N., Phan, T. A., Barnett, C., & Ibrahim, N. (2002). Angular gyrus syndrome mimicking depressive pseudodementia. *J Psychiatry Neurosci*.
- Nagel, T. (1974). What is it like to be a bat?
- Nashner, L. M. (2014). Practical biomechanics and physiology of balance. *Balance Function Assessment and Management*, 431.
- Neal, D. T., Wood, W., & Quinn, J. M. (2006). Habits—A Repeat Performance. Current Directions in Psychological Science, 15(4), 198–202. https://doi.org/10.1111/j.1467-8721.2006.00435.x

- Niendam, T. A., Laird, A. R., Ray, K. L., Dean, Y. M., Glahn, D. C., & Carter, C. S. (2012). Metaanalytic evidence for a superordinate cognitive control network subserving diverse executive functions. *Cognitive, Affective, & Behavioral Neuroscience, 12*(2), 241–268. https://doi.org/10.3758/s13415-011-0083-5
- Niu, M., Impieri, D., Rapan, L., Funck, T., Palomero-Gallagher, N., & Zilles, K. (2020). Receptordriven, multimodal mapping of cortical areas in the macaque monkey intraparietal sulcus. *eLife*, 9, e55979. https://doi.org/10.7554/eLife.55979
- Niu, M., Rapan, L., Funck, T., Froudist-Walsh, S., Zhao, L., Zilles, K., & Palomero-Gallagher, N. (2021). Organization of the macaque monkey inferior parietal lobule based on multimodal receptor architectonics. *NeuroImage*, 231, 117843. https://doi.org/10.1016/j.neuroimage.2021.117843
- Norman, D. A. (2013). The design of everyday things (Revised and expanded edition). Basic Books.
- Nour, M. M., Dahoun, T., Schwartenbeck, P., Adams, R. A., FitzGerald, T. H. B., Coello, C., Wall, M. B., Dolan, R. J., & Howes, O. D. (2018). Dopaminergic basis for signaling belief updates, but not surprise, and the link to paranoia. *Proceedings of the National Academy of Sciences*, 115(43). https://doi.org/10.1073/pnas.1809298115
- Numssen, O., Bzdok, D., & Hartwigsen, G. (2021). Functional specialization within the inferior parietal lobes across cognitive domains. *eLife*, 10, e63591. https://doi.org/10.7554/eLife.63591
- Oberhuber, M., Hope, T. M. H., Seghier, M. L., Parker Jones, O., Prejawa, S., Green, D. W., & Price, C. J. (2016). Four Functionally Distinct Regions in the Left Supramarginal Gyrus Support Word Processing. *Cerebral Cortex*, 26(11), 4212–4226. https://doi.org/10.1093/cercor/bhw251
- Obeso, I., Moisa, M., Ruff, C. C., & Dreher, J.-C. (2018). A causal role for right temporo-parietal junction in signaling moral conflict. *eLife*, 7, e40671. https://doi.org/10.7554/eLife.40671
- Odling-Smee, J., Erwin, D. H., Palkovacs, E. P., Feldman, M. W., & Laland, K. N. (2013). Niche Construction Theory: A Practical Guide for Ecologists. *The Quarterly Review of Biology*, 88(1), 3–28. https://doi.org/10.1086/669266
- Odling-Smee, J., Laland, K. N., & Feldman, M. W. (2003). *Niche construction, the neglected process in evolution*. Monographs in Population Biology.
- Ogawa, A., & Kameda, T. (2019). Dissociable roles of left and right temporoparietal junction in strategic competitive interaction. *Social Cognitive and Affective Neuroscience*, *14*(10), 1037–1048. https://doi.org/10.1093/scan/nsz082
- O'Keefe, J. (1976). Place units in the hippocampus of the freely moving rat. *Experimental Neurology*, 51(1), 78–109. https://doi.org/10.1016/0014-4886(76)90055-8
- O'Keefe, J., & Dostrovsky, J. (1971). The hippocampus as a spatial map. Preliminary evidence from unit activity in the freely-moving rat. *Brain Research*, 34(1), 171–175. https://doi.org/10.1016/0006-8993(71)90358-1
- Olsen, R. K., Carr, V. A., Daugherty, A. M., La Joie, R., Amaral, R. S. C., Amunts, K., Augustinack, J. C., Bakker, A., Bender, A. R., Berron, D., Boccardi, M., Bocchetta, M., Burggren, A. C., Chakravarty, M. M., Chételat, G., Flores, R., DeKraker, J., Ding, S., Geerlings, M. I., ... Hippocampal Subfields Group. (2019). Progress update from the hippocampal subfields group. *Alzheimer's & Dementia: Diagnosis, Assessment & Disease Monitoring*, 11(1), 439–449. https://doi.org/10.1016/j.dadm.2019.04.001
- Olsen, R. K., Moses, S. N., Riggs, L., & Ryan, J. D. (2012). The hippocampus supports multiple cognitive processes through relational binding and comparison. *Frontiers in Human Neuroscience*, 6. https://doi.org/10.3389/fnhum.2012.00146
- Osada, T., Ohta, S., Ogawa, A., Tanaka, M., Suda, A., Kamagata, K., Hori, M., Aoki, S., Shimo, Y., Hattori, N., Shimizu, T., Enomoto, H., Hanajima, R., Ugawa, Y., & Konishi, S. (2019). An Essential Role of the Intraparietal Sulcus in Response Inhibition Predicted by Parcellation-Based Network. *The Journal of Neuroscience*, 39(13), 2509–2521. https://doi.org/10.1523/JNEUROSCI.2244-18.2019
- Otto, A. R., Gershman, S. J., Markman, A. B., & Daw, N. D. (2013). The Curse of Planning: Dissecting Multiple Reinforcement-Learning Systems by Taxing the Central Executive. *Psychological Science*, 24(5), 751–761. https://doi.org/10.1177/0956797612463080
- Oyama, S. (2000). *The ontogeny of information: Developmental systems and evolution*. Duke University Press.
- Pacherie, E. (2008). The phenomenology of action: A conceptual framework. *Cognition*, 107(1), 179–217. https://doi.org/10.1016/j.cognition.2007.09.003
- Packard, M. G. (2008). Neurobiology of Procedural Learning in Animals. 16.

- Packard, M. G., & McGaugh, J. L. (1996). Inactivation of Hippocampus or Caudate Nucleus with Lidocaine Differentially Affects Expression of Place and Response Learning. *Neurobiology of Learning and Memory*, 65(1), 65–72. https://doi.org/10.1006/nlme.1996.0007
- Palombo, D. J., Alain, C., Söderlund, H., Khuu, W., & Levine, B. (2015). Severely deficient autobiographical memory (SDAM) in healthy adults: A new mnemonic syndrome. *Neuropsychologia*, 72, 105–118. https://doi.org/10.1016/j.neuropsychologia.2015.04.012
- Palombo, D. J., Sheldon, S., & Levine, B. (2018). Individual Differences in Autobiographical Memory. *Trends in Cognitive Sciences*, 22(7), 583–597. https://doi.org/10.1016/j.tics.2018.04.007
- Pastalkova, E., Itskov, V., Amarasingham, A., & Buzsáki, G. (2008). Internally Generated Cell Assembly Sequences in the Rat Hippocampus. *Science*, 321(5894), 1322–1327. https://doi.org/10.1126/science.1159775
- Pateria, S., Subagdja, B., Tan, A., & Quek, C. (2022). Hierarchical Reinforcement Learning: A Comprehensive Survey. ACM Computing Surveys, 54(5), 1–35. https://doi.org/10.1145/3453160
- Pearl, J. (1998). Graphs, Causality, and Structural Equation Models. *Sociological Methods & Research*, 27(2), 226–284. https://doi.org/10.1177/0049124198027002004
- Peer, M., Brunec, I. K., Newcombe, N. S., & Epstein, R. A. (2021). Structuring Knowledge with Cognitive Maps and Cognitive Graphs. *Trends in Cognitive Sciences*, 25(1), 37–54. https://doi.org/10.1016/j.tics.2020.10.004
- Peng, S., Kuang, B., Zhang, L., & Hu, P. (2021). Right Temporoparietal Junction Plays a Role in the Modulation of Emotional Mimicry by Group Membership. *Frontiers in Human Neuroscience*, 15, 606292. https://doi.org/10.3389/fnhum.2021.606292
- Peterson, C. C., Wellman, H. M., & Slaughter, V. (2012). The Mind Behind the Message: Advancing Theory-of-Mind Scales for Typically Developing Children, and Those With Deafness, Autism, or Asperger Syndrome: The Mind Behind the Message. Child Development, 83(2), 469–485. https://doi.org/10.1111/j.1467-8624.2011.01728.x
- Pezzulo, G., & Cisek, P. (2016). Navigating the Affordance Landscape: Feedback Control as a Process Model of Behavior and Cognition. *Trends in Cognitive Sciences*, 20(6), 414–424. https://doi.org/10.1016/j.tics.2016.03.013
- Piaget, J. (1970). GENETIC EPISTEMOLOGY. 88.
- Pietroski, P. (1998). Actions, adjuncts, and agency. *Mind*, 107(425), 73–111. https://doi.org/10.1093/mind/107.425.73
- Pinker, S. (2013). Language, Cognition, and Human Nature. Oxford University Press. https://doi.org/10.1093/acprof:oso/9780199328741.001.0001
- Pleskac, T. J., & Busemeyer, J. R. (2010). Two-stage dynamic signal detection: A theory of choice, decision time, and confidence. *Psychological Review*, 117(3), 864–901. https://doi.org/10.1037/a0019737
- Pocheville, A. (2015). The Ecological Niche: History and Recent Controversies. In T. Heams, P. Huneman, G. Lecointre, & M. Silberstein (Eds.), *Handbook of Evolutionary Thinking in the Sciences* (pp. 547–586). Springer Netherlands. https://doi.org/10.1007/978-94-017-9014-7_26
- Poldrack, R. A., & Packard, M. G. (2003a). Competition among multiple memory systems: Converging evidence from animal and human brain studies. *Neuropsychologia*, 41(3), 245–251. https://doi.org/10.1016/S0028-3932(02)00157-4
- Poldrack, R. A., & Packard, M. G. (2003b). Competition among multiple memory systems: Converging evidence from animal and human brain studies. *Neuropsychologia*, 41(3), 245–251. https://doi.org/10.1016/S0028-3932(02)00157-4
- Popp, N. J., Yokoi, A., Gribble, P. L., & Diedrichsen, J. (2018). The effects of habits on motor skill learning [Preprint]. Neuroscience. https://doi.org/10.1101/338749
- Potter, M. C., Wyble, B., Hagmann, C. E., & McCourt, E. S. (2014). Detecting meaning in RSVP at 13 ms per picture. *Attention, Perception, & Psychophysics*, 76(2), 270–279. https://doi.org/10.3758/s13414-013-0605-z
- Preuss, T. M., & Wise, S. P. (2022). Evolution of prefrontal cortex. *Neuropsychopharmacology*, 47(1), 3–19. https://doi.org/10.1038/s41386-021-01076-5
- Ptak, R., Schnider, A., & Fellrath, J. (2017). The Dorsal Frontoparietal Network: A Core System for Emulated Action. *Trends in Cognitive Sciences*, 21(8), 589–599. https://doi.org/10.1016/j.tics.2017.05.002
- Quesque, F., & Brass, M. (2019). The Role of the Temporoparietal Junction in Self-Other Distinction. Brain Topography, 32(6), 943–955. https://doi.org/10.1007/s10548-019-00737-5
- Quiroga, R. Q. (2012). Concept cells: The building blocks of declarative memory functions. *Nature Reviews Neuroscience*, 13(8), 587–597. https://doi.org/10.1038/nrn3251

- Quiroga, R. Q. (2017). How Do We Recognize a Face? *Cell*, *169*(6), 975–977. https://doi.org/10.1016/j.cell.2017.05.012
- Quiroga, R. Q. (2020). No Pattern Separation in the Human Hippocampus. *Trends in Cognitive Sciences*, 24(12), 994–1007. https://doi.org/10.1016/j.tics.2020.09.012
- Ragozzino, M. E. (2002). The Effects of Dopamine D₁ Receptor Blockade in the Prelimbic–Infralimbic Areas on Behavioral Flexibility. *Learning & Memory*, 9(1), 18–28. https://doi.org/10.1101/lm.45802
- Rah, S. K., Reber, A. S., & Hsiao, A. T. (2000). Another wrinkle on the dual-task SRT experiment: It's probably not dual task. *Psychonomic Bulletin & Review*, 7(2), 309–313. https://doi.org/10.3758/BF03212986
- Raichle, M. E., MacLeod, A. M., Snyder, A. Z., Powers, W. J., Gusnard, D. A., & Shulman, G. L. (2001). A default mode of brain function. *Proceedings of the National Academy of Sciences*, 98(2), 676–682. https://doi.org/10.1073/pnas.98.2.676
- Raiola, G. (2014). Motor Control and Learning Skills According to Cognitive and Ecological Dynamic Approach in a Vision on Behaviorism, Cognitive, Gestalt and Phenomenology Theories. *Mediterranean Journal of Social Sciences*. https://doi.org/10.5901/mjss.2014.v5n15p504
- Raju, H., & Tadi, P. (2021). Neuroanatomy, Somatosensory Cortex-StatPearls-NCBI Bookshelf. 11.
- Ramsey, W. M. (2007). Representation Reconsidered. Cambridge University Press.
 - https://doi.org/10.1017/CBO9780511597954
- Ranganathan, V. K., Siemionow, V., Liu, J. Z., Sahgal, V., & Yue, G. H. (2004). From mental power to muscle power—Gaining strength by using the mind. *Neuropsychologia*, 42(7), 944–956. https://doi.org/10.1016/j.neuropsychologia.2003.11.018
- Reason, J. T. (1990). Human error. https://doi.org/10.1017/CBO9781139062367
- Reddan, M. C., Wager, T. D., & Schiller, D. (2018). Attenuating Neural Threat Expression with Imagination. *Neuron*, 100(4), 994-1005.e4. https://doi.org/10.1016/j.neuron.2018.10.047
- Reddy, L., & Thorpe, S. J. (2014). Concept Cells through Associative Learning of High-Level Representations. *Neuron*, 84(2), 248–251. https://doi.org/10.1016/j.neuron.2014.10.004
- Redish, A. D., Jensen, S., Johnson, A., & Kurth-Nelson, Z. (2007). Reconciling reinforcement learning models with behavioral extinction and renewal: Implications for addiction, relapse, and problem gambling. *Psychological Review*, 114(3), 784–805. https://doi.org/10.1037/0033-295X.114.3.784
- Reed, E. (1996). Encountering the world: Toward an ecological psychology. Oxford University Press.
- Regenbogen, C., Seubert, J., Johansson, E., Finkelmeyer, A., Andersson, P., & Lundström, J. N. (2018). The intraparietal sulcus governs multisensory integration of audiovisual information based on task difficulty. *Human Brain Mapping*, 39(3), 1313–1326. https://doi.org/10.1002/hbm.23918
- Renoult, L., Davidson, P. S. R., Palombo, D. J., Moscovitch, M., & Levine, B. (2012). Personal semantics: At the crossroads of semantic and episodic memory. *Trends in Cognitive Sciences*, 16(11), 550–558. https://doi.org/10.1016/j.tics.2012.09.003
- Renoult, L., Irish, M., Moscovitch, M., & Rugg, M. D. (2019). From Knowing to Remembering: The Semantic–Episodic Distinction. *Trends in Cognitive Sciences*, 23(12), 1041–1057. https://doi.org/10.1016/j.tics.2019.09.008
- Richards, D., Caldwell, P. H., & Go, H. (2015). Impact of social media on the health of children and young people: Social media and the health of young people. *Journal of Paediatrics and Child Health*, 51(12), 1152–1157. https://doi.org/10.1111/jpc.13023
- Richardson, M. J., Shockley, K., Fajen, B. R., Riley, M. A., & Turvey, M. T. (2008). Ecological Psychology: Six Principles for an Embodied–Embedded Approach to Behavior. *Handbook of Cognitive Science*, 28.
- Rietveld, E., Denys, D., & van Westen, M. (2018). Ecological-enactive cognition as engaging with a field of relevant affordances, the skilled intentionality framework (SIF). In *Oxford Handbook of 4E cognition*. Oxford University Press.
- Rizzolatti, G., Fadiga, L., Gallese, V., & Fogassi, L. (1996). Premotor cortex and the recognition of motor actions. *Cognitive Brain Research*, 3(2), 131–141. https://doi.org/10.1016/0926-6410(95)00038-0
- Rizzolatti, G., Fogassi, L., & Gallese, V. (2002). Motor and cognitive functions of the ventral premotor cortex. *Current Opinion in Neurobiology*, 12(2), 149–154. https://doi.org/10.1016/S0959-4388(02)00308-2
- Rizzolatti, G., Scandolara, C., Matelli, M., & Gentilucci, M. (1981). Afferent properties of periarcuate neurons in macaque monkeys. I. Somatosensory responses. *Behavioural Brain Research*, 2(2), 125–146. https://doi.org/10.1016/0166-4328(81)90052-8

- Rodrigo, S. E., & Ambrósio, J. A. (2011). *A biomechanical approach to characterize neural coordination during gait.* 16.
- Roitman, M. F., Wheeler, R. A., Wightman, R. M., & Carelli, R. M. (2008). Real-time chemical responses in the nucleus accumbens differentiate rewarding and aversive stimuli. *Nature Neuroscience*, 11(12), 1376–1377. https://doi.org/10.1038/nn.2219
- Rolls, E. T. (2016). *Cerebral Cortex: Principles of Operation*. Oxford University Press. https://doi.org/10.1093/acprof:oso/9780198784852.001.0001
- Rolls, E. T., & Xiang, J.-Z. (2006a). Spatial view cells in the primate hippocampus and memory recall. *Reviews in the Neurosciences*, 17(1–2), 175–200. https://doi.org/10.1515/REVNEURO.2006.17.1-2.175
- Rolls, E. T., & Xiang, J.-Z. (2006b). Spatial view cells in the primate hippocampus and memory recall. *Reviews in Neurosciences*, 17, 175–200.
- Rosch, E., & Mervis, C. (1975). Family Resemblances: Studies in the Internal Structure of Categories. 7, 573–605.
- Ross, L., & Ward, A. (1996). Naive realism in everyday life, implications for social conflict and misunderstanding.
- Rowland, D. (2020). A Need to Redefine Autism. Journal of Neurology, 1.
- Roy, D. S., Kitamura, T., Okuyama, T., Ogawa, S. K., Sun, C., Obata, Y., Yoshiki, A., & Tonegawa, S. (2017). Distinct Neural Circuits for the Formation and Retrieval of Episodic Memories. *Cell*, 170(5), 1000-1012.e19. https://doi.org/10.1016/j.cell.2017.07.013
- Roy, D. S., Muralidhar, S., Smith, L. M., & Tonegawa, S. (2017). Silent memory engrams as the basis for retrograde amnesia. *Proceedings of the National Academy of Sciences*, 114(46). https://doi.org/10.1073/pnas.1714248114
- Roy, D. S., Park, Y.-G., Kim, M. E., Zhang, Y., Ogawa, S. K., DiNapoli, N., Gu, X., Cho, J. H., Choi, H., Kamentsky, L., Martin, J., Mosto, O., Aida, T., Chung, K., & Tonegawa, S. (2022). Brainwide mapping reveals that engrams for a single memory are distributed across multiple brain regions. *Nature Communications*, 13(1), 1799. https://doi.org/10.1038/s41467-022-29384-4
- Rudebeck, P. H., & Murray, E. A. (2014). The Orbitofrontal Oracle: Cortical Mechanisms for the Prediction and Evaluation of Specific Behavioral Outcomes. *Neuron*, 84(6), 1143–1156. https://doi.org/10.1016/j.neuron.2014.10.049
- Rueda, M. R., Fan, J., McCandliss, B. D., Halparin, J. D., Gruber, D. B., Lercari, L. P., & Posner, M. I. (2004). Development of attentional networks in childhood. *Neuropsychologia*, 42(8), 1029– 1040. https://doi.org/10.1016/j.neuropsychologia.2003.12.012
- Rueter, A. R., Abram, S. V., MacDonald, A. W., Rustichini, A., & DeYoung, C. G. (2018). The goal priority network as a neural substrate of Conscientiousness. *Human Brain Mapping*, 39(9), 3574–3585. https://doi.org/10.1002/hbm.24195
- Saarinen, J. A., & Krueger, J. (2022). Making Space for Creativity: Niche Construction and the Artist's Studio. *The Journal of Aesthetics and Art Criticism*, 80(3), 322–332. https://doi.org/10.1093/jaac/kpac021
- Safron, A. (2020). An Integrated World Modeling Theory (IWMT) of Consciousness: Combining Integrated Information and Global Neuronal Workspace Theories With the Free Energy Principle and Active Inference Framework; Toward Solving the Hard Problem and Characterizing Agentic Causation. *Frontiers in Artificial Intelligence*, 3, 30. https://doi.org/10.3389/frai.2020.00030
- Safron, A. (2021). The Radically Embodied Conscious Cybernetic Bayesian Brain: From Free Energy to Free Will and Back Again. *Entropy*, 23(6), 783. https://doi.org/10.3390/e23060783
- Saint-Cyr, J. A., Taylor, A. E., & Lang, A. E. (1988). Procedural learning and neostyrial dysfunction in man. Brain : A Journal of Neurology, 111(4), 941–960.
- Sarel, A., Finkelstein, A., Las, L., & Ulanovsky, N. (2017). Vectorial representation of spatial goals in the hippocampus of bats. *Science*, 355(6321), 176–180. https://doi.org/10.1126/science.aak9589
- Sargolini, F., Fyhn, M., Hafting, T., McNaughton, B. L., Witter, M. P., Moser, M.-B., & Moser, E. I. (2006). Conjunctive Representation of Position, Direction, and Velocity in Entorhinal Cortex. *Science*, 312(5774), 758–762. https://doi.org/10.1126/science.1125572
- Savelli, F., Yoganarasimha, D., & Knierim, J. J. (2008). Influence of boundary removal on the spatial representations of the medial entorhinal cortex. *Hippocampus*, 18(12), 1270–1282. https://doi.org/10.1002/hipo.20511
- Schaal, N. K., Pollok, B., & Banissy, M. J. (2017). Hemispheric differences between left and right supramarginal gyrus for pitch and rhythm memory. *Scientific Reports*, 7(1), 42456. https://doi.org/10.1038/srep42456

- Schacter, D. L., Addis, D. R., Hassabis, D., Martin, V. C., Spreng, R. N., & Szpunar, K. K. (2012). The Future of Memory: Remembering, Imagining, and the Brain. *Neuron*, 76(4), 677–694. https://doi.org/10.1016/j.neuron.2012.11.001
- Schmidtke, V., & Heuer, H. (1997). Task integration as a factor in secondary-task effects on sequence learning. *Psychological Research*, 60(1–2), 53–71. https://doi.org/10.1007/BF00419680
- Schouenborg, J. (2004). Learning in sensorimotor circuits. *Current Opinion in Neurobiology*, 14(6), 693–697. https://doi.org/10.1016/j.conb.2004.10.009
- Schultz, W. (2007). Multiple Dopamine Functions at Different Time Courses. Annual Review of Neuroscience, 30(1), 259–288. https://doi.org/10.1146/annurev.neuro.28.061604.135722
- Schultz, W., Dayan, P., & Montague, P. R. (1997). A Neural Substrate of Prediction and Reward. https://doi.org/10.1126/science.275.5306.1593
- Schwartenbeck, P., FitzGerald, T. H. B., & Dolan, R. (2016). Neural signals encoding shifts in beliefs. *NeuroImage*, 125, 578–586. https://doi.org/10.1016/j.neuroimage.2015.10.067
- Searle, J. R. (1983). Intentionality, an essay in the philosophy of mind. Cambridge University Press.
- Seeley, W. W., Menon, V., Schatzberg, A. F., Keller, J., Glover, G. H., Kenna, H., Reiss, A. L., & Greicius, M. D. (2007). Dissociable Intrinsic Connectivity Networks for Salience Processing and Executive Control. *The Journal of Neuroscience*, 27(9), 2349–2356. https://doi.org/10.1523/JNEUROSCI.5587-06.2007
- Seghier, M. L., Fagan, E., & Price, C. J. (2010). Functional Subdivisions in the Left Angular Gyrus Where the Semantic System Meets and Diverges from the Default Network. *The Journal of Neuroscience*, 30(50), 16809–16817. https://doi.org/10.1523/JNEUROSCI.3377-10.2010
- Serino, A. (2019). Peripersonal space (PPS) as a multisensory interface between the individual and the environment, defining the space of the self. *Neuroscience & Biobehavioral Reviews*, 99, 138– 159. https://doi.org/10.1016/j.neubiorev.2019.01.016
- Serway, R., & Jewett, J. (2018). Physics for scientists and engineers with modern physics.
- Seth, A. K. (2013). Interoceptive inference, emotion, and the embodied self. Trends in Cognitive Sciences, 17(11), 565–573. https://doi.org/10.1016/j.tics.2013.09.007
- Seth, A. K., & Bayne, T. (2022). Theories of consciousness. *Nature Reviews Neuroscience*, 23(7), 439–452. https://doi.org/10.1038/s41583-022-00587-4
- Shatek, S. M., Robinson, A. K., Grootswagers, T., & Carlson, T. A. (2022). Capacity for movement is an organisational principle in object representations. *NeuroImage*, 261, 119517. https://doi.org/10.1016/j.neuroimage.2022.119517
- Sherman, S. M. (2017). Functioning of Circuits Connecting Thalamus and Cortex. In R. Terjung (Ed.), Comprehensive Physiology (1st ed., pp. 713–739). Wiley. https://doi.org/10.1002/cphy.c160032
- Shultz, S., Lee, S. M., Pelphrey, K., & McCarthy, G. (2011). The posterior superior temporal sulcus is sensitive to the outcome of human and non-human goal-directed actions. *Social Cognitive and Affective Neuroscience*, 6(5), 602–611. https://doi.org/10.1093/scan/nsq087
- Siclari, F., Baird, B., Perogamvros, L., Bernardi, G., LaRocque, J. J., Riedner, B., Boly, M., Postle, B. R., & Tononi, G. (2017). The neural correlates of dreaming. *Nature Neuroscience*, 20(6), 872–878. https://doi.org/10.1038/nn.4545
- Siddiqui, S., & Singh, T. (2016). Social Media its Impact with Positive and Negative Aspects. International Journal of Computer Applications Technology and Research, 5(2), 71–75. https://doi.org/10.7753/IJCATR0502.1006
- Silani, G., Lamm, C., Ruff, C. C., & Singer, T. (2013). Right Supramarginal Gyrus Is Crucial to Overcome Emotional Egocentricity Bias in Social Judgments. *The Journal of Neuroscience*, 33(39), 15466–15476. https://doi.org/10.1523/JNEUROSCI.1488-13.2013
- Silver, D., Hubert, T., Schrittwieser, J., Antonoglou, I., Lai, M., Guez, A., Lanctot, M., Sifre, L., Kumaran, D., Graepel, T., Lillicrap, T., Simonyan, K., & Hassabis, D. (2018). A general reinforcement learning algorithm that masters chess, shogi, and Go through self-play. *Science*, 362(6419), 1140–1144. https://doi.org/10.1126/science.aar6404
- Simon, D. A., & Daw, N. D. (2011). Neural Correlates of Forward Planning in a Spatial Decision Task in Humans. *Journal of Neuroscience*, 31(14), 5526–5539. https://doi.org/10.1523/JNEUROSCI.4647-10.2011
- Smith, A. C. W., Jonkman, S., Difeliceantonio, A. G., O'Connor, R. M., Ghoshal, S., Romano, M. F., Everitt, B. J., & Kenny, P. J. (2021). Opposing roles for striatonigral and striatopallidal neurons in dorsolateral striatum in consolidating new instrumental actions. *Nature Communications*, 12(1), 5121. https://doi.org/10.1038/s41467-021-25460-3
- Smith, Fox, P. T., Miller, K. L., Glahn, D. C., Fox, P. M., Mackay, C. E., Filippini, N., Watkins, K. E., Toro, R., Laird, A. R., & Beckmann, C. F. (2009). Correspondence of the brain's functional

architecture during activation and rest. *Proceedings of the National Academy of Sciences*, 106(31), 13040–13045. https://doi.org/10.1073/pnas.0905267106

- Smith, K. S., & Graybiel, A. M. (2013). A Dual Operator View of Habitual Behavior Reflecting Cortical and Striatal Dynamics. *Neuron*, 79(2), 361–374. https://doi.org/10.1016/j.neuron.2013.05.038
- Smith, P. F., Horii, A., Russell, N., Bilkey, D. K., Zheng, Y., Liu, P., Kerr, D. S., & Darlington, C. L. (2005). The effects of vestibular lesions on hippocampal function in rats. *Progress in Neurobiology*, 75(6), 391–405. https://doi.org/10.1016/j.pneurobio.2005.04.004
- Smith, S. M., Glenberg, A., & Bjork, R. A. (1978). Environmental context and human memory. *Memory* & Cognition, 6(4), 342–353. https://doi.org/10.3758/BF03197465
- Smittenaar, P., FitzGerald, T. H. B., Romei, V., Wright, N. D., & Dolan, R. J. (2013a). Disruption of Dorsolateral Prefrontal Cortex Decreases Model-Based in Favor of Model-free Control in Humans. *Neuron*, 80(4), 914–919. https://doi.org/10.1016/j.neuron.2013.08.009
- Smittenaar, P., FitzGerald, T. H. B., Romei, V., Wright, N. D., & Dolan, R. J. (2013b). Disruption of Dorsolateral Prefrontal Cortex Decreases Model-Based in Favor of Model-free Control in Humans. *Neuron*, 80(4), 914–919. https://doi.org/10.1016/j.neuron.2013.08.009
- Solstad, T., Boccara, C. N., Kropff, E., Moser, M.-B., & Moser, E. I. (2008). Representation of Geometric Borders in the Entorhinal Cortex. *Science*, 322(5909), 1865–1868. https://doi.org/10.1126/science.1166466
- Sormaz, M., Murphy, C., Wang, H., Hymers, M., Karapanagiotidis, T., Poerio, G., Margulies, D. S., Jefferies, E., & Smallwood, J. (2018). Default mode network can support the level of detail in experience during active task states. *Proceedings of the National Academy of Sciences*, 115(37), 9318–9323. https://doi.org/10.1073/pnas.1721259115
- Spiers, H. J. (2020). The Hippocampal Cognitive Map: One Space or Many? Trends in Cognitive Sciences, 24(3), 168–170. https://doi.org/10.1016/j.tics.2019.12.013
- Spreng, R. N., Mar, R. A., & Kim, A. S. N. (2009). The Common Neural Basis of Autobiographical Memory, Prospection, Navigation, Theory of Mind, and the Default Mode: A Quantitative Meta-analysis. *Journal of Cognitive Neuroscience*, 21(3), 489–510. https://doi.org/10.1162/jocn.2008.21029
- Spreng, R. N., Stevens, W. D., Chamberlain, J. P., Gilmore, A. W., & Schacter, D. L. (2010). Default network activity, coupled with the frontoparietal control network, supports goal-directed cognition. *NeuroImage*, 53(1), 303–317. https://doi.org/10.1016/j.neuroimage.2010.06.016
- Spreng, R. N., Stevens, W. D., Viviano, J. D., & Schacter, D. L. (2016). Attenuated anticorrelation between the default and dorsal attention networks with aging: Evidence from task and rest. *Neurobiology of Aging*, 45, 149–160. https://doi.org/10.1016/j.neurobiolaging.2016.05.020
- Spunt, R. P., & Lieberman, M. D. (2012). Dissociating Modality-Specific and Supramodal Neural Systems for Action Understanding. *The Journal of Neuroscience*, 32(10), 3575–3583. https://doi.org/10.1523/JNEUROSCI.5715-11.2012
- Squire, L. R. (1992). Memory and the hippocampus: A synthesis from findings with rats, monkeys, and humans. *Psychological Review*, *99*(2), 195. https://doi.org/10.1037/0033-295X.99.2.195
- Squire, L. R. (2004). Memory systems of the brain: A brief history and current perspective. *Neurobiology* of Learning and Memory, 82(3), 171–177. https://doi.org/10.1016/j.nlm.2004.06.005
- Squire, L. R. (2009a). Memory and Brain Systems: 1969–2009. *The Journal of Neuroscience*, 29(41), 12711–12716. https://doi.org/10.1523/JNEUROSCI.3575-09.2009
- Squire, L. R. (2009b). The Legacy of Patient H.M. for Neuroscience. *Neuron*, 61(1), 6–9. https://doi.org/10.1016/j.neuron.2008.12.023
- Stackman, R. W., Clark, A. S., & Taube, J. S. (2002). Hippocampal spatial representations require vestibular input. *Hippocampus*, 12(3), 291–303. https://doi.org/10.1002/hipo.1112
- Starkweather, C. K., Babayan, B. M., Uchida, N., & Gershman, S. J. (2017). Dopamine reward prediction errors reflect hidden-state inference across time. *Nature Neuroscience*, 20(4), 581– 589. https://doi.org/10.1038/nn.4520
- Stehr, D. A., Zhou, X., Tisby, M., Hwu, P. T., Pyles, J. A., & Grossman, E. D. (2021). Top-Down Attention Guidance Shapes Action Encoding in the pSTS. *Cerebral Cortex*, 31(7), 3522–3535. https://doi.org/10.1093/cercor/bhab029
- Strange, B. A. (2022). Connecting to the long axis. *eLife*, *11*, e83718. https://doi.org/10.7554/eLife.83718
- Sun, C., Yang, W., Martin, J., & Tonegawa, S. (2020). Hippocampal neurons represent events as transferable units of experience. *Nature Neuroscience*, 23(5), 651–663. https://doi.org/10.1038/s41593-020-0614-x
- Sutton, R. S. (1991). Dyna, an integrated architecture for learning, planning, and reacting. *ACM SIGART Bulletin*, 2(4), 160–163. https://doi.org/10.1145/122344.122377

- Sutton, R. S., & Barto, A. G. (2018a). Reinforcement learning: An introduction (Second edition). The MIT Press.
- Sutton, R. S., & Barto, A. G. (2018b). *Reinforcement learning: An introduction* (Second edition). The MIT Press.

Szczypka, M. S., Rainey, M. A., Kim, D. S., Alaynick, W. A., Marck, B. T., Matsumoto, A. M., & Palmiter, R. D. (1999). Feeding behavior in dopamine-deficient mice. *Proceedings of the National Academy of Sciences*, 96(21), 12138–12143. https://doi.org/10.1073/pnas.96.21.12138

- Takeuchi, T., Duszkiewicz, A. J., Sonneborn, A., Spooner, P. A., Yamasaki, M., Watanabe, M., Smith, C. C., Fernández, G., Deisseroth, K., Greene, R. W., & Morris, R. G. M. (2016). Locus coeruleus and dopaminergic consolidation of everyday memory. *Nature*, 537(7620), 357–362. https://doi.org/10.1038/nature19325
- Tavares, R. M., Mendelsohn, A., Grossman, Y., Williams, C. H., Shapiro, M., Trope, Y., & Schiller, D. (2015). A Map for Social Navigation in the Human Brain. *Neuron*, 87(1), 231–243. https://doi.org/10.1016/j.neuron.2015.06.011
- Taylor, J. E. T., Witt, J. K., & Sugovic, M. (2011). When Walls are No Longer Barriers: Perception of Wall Height in Parkour. *Perception*, 40(6), 757–760. https://doi.org/10.1068/p6855
- ten Donkelaar, H. J., Broman, J., & van Domburg, P. (2020). The Somatosensory System. In H. J. ten Donkelaar, *Clinical Neuroanatomy* (pp. 171–255). Springer International Publishing. https://doi.org/10.1007/978-3-030-41878-6 4
- Thorat, S., Proklova, D., & Peelen, M. V. (2019). The nature of the animacy organization in human ventral temporal cortex. *eLife*, *8*, e47142. https://doi.org/10.7554/eLife.47142
- Thorn, C. A., Atallah, H., Howe, M., & Graybiel, A. M. (2010). Differential Dynamics of Activity Changes in Dorsolateral and Dorsomedial Striatal Loops during Learning. *Neuron*, 66(5), 781– 795. https://doi.org/10.1016/j.neuron.2010.04.036
- Thorndike, E. L. (1898). Animal intelligence: An experimental study of the associative processes in animals. *The Psychological Review: Monograph Supplements*, 2(4), i.
- Titulaer, J., Björkholm, C., Feltmann, K., Malmlöf, T., Mishra, D., Bengtsson Gonzales, C., Schilström, B., & Konradsson-Geuken, Å. (2021). The Importance of Ventral Hippocampal Dopamine and Norepinephrine in Recognition Memory. *Frontiers in Behavioral Neuroscience*, 15, 667244. https://doi.org/10.3389/fnbeh.2021.667244
- Todorov, E. (2004). Optimality principles in sensorimotor control. *Nature Neuroscience*, 7(9), 907–915. https://doi.org/10.1038/nn1309
- Todorov, E., & Jordan, M. I. (2002). A Minimal Intervention Principle for Coordinated Movement.
- Tolman, E. C. (1948). Cognitive maps in rats and men. *Psychological Review*, 55(4), 189. https://doi.org/10.1037/h0061626
- Tomasello, M. (2009). The cultural origins of human cognition. Harvard university press.
- Tomasello, M. (2010). Origins of human communication. MIT press.
- Tomasello, M. (2018). How children come to understand false beliefs: A shared intentionality account. *Proceedings of the National Academy of Sciences*, 115(34), 8491–8498. https://doi.org/10.1073/pnas.1804761115
- Tomasello, M. (2019). Becoming human: A theory of ontogeny. Harvard University Press.
- Tomasello, M., Call, J., & Hare, B. (2003). Chimpanzees understand psychological states the question is which ones and to what extent. *Trends in Cognitive Sciences*, 7(4), 153–156. https://doi.org/10.1016/S1364-6613(03)00035-4
- Tomasello, M., & Rakoczy, H. (2003). What Makes Human Cognition Unique? From Individual to Shared to Collective Intentionality. *Mind and Language*, *18*(2), 121–147. https://doi.org/10.1111/1468-0017.00217
- Tomasi, D., & Volkow, N. D. (2014). Functional Connectivity of Substantia Nigra and Ventral Tegmental Area: Maturation During Adolescence and Effects of ADHD. Cerebral Cortex, 24(4), 935–944. https://doi.org/10.1093/cercor/bhs382
- Tonegawa, S., Liu, X., Ramirez, S., & Redondo, R. (2015). Memory Engram Cells Have Come of Age. *Neuron*, 87(5), 918–931. https://doi.org/10.1016/j.neuron.2015.08.002
- Tong, J., Binder, J. R., Humphries, C., Mazurchuk, S., Conant, L. L., & Fernandino, L. (2022). A Distributed Network for Multimodal Experiential Representation of Concepts. *The Journal of Neuroscience*, 42(37), 7121–7130. https://doi.org/10.1523/JNEUROSCI.1243-21.2022
- Tononi, G., Boly, M., Massimini, M., & Koch, C. (2016). Integrated information theory: From consciousness to its physical substrate. *Nature Reviews Neuroscience*, 17(7), 450–461. https://doi.org/10.1038/nrn.2016.44

- Tononi, G., & Koch, C. (2015). Consciousness: Here, there and everywhere? *Philosophical Transactions of the Royal Society B: Biological Sciences*, 370(1668), 20140167. https://doi.org/10.1098/rstb.2014.0167
- Tooby, J., & DeVore, I. (1987). The Reconstruction of Hominid Behavioral Evolution Through Strategic Modeling. 28.
- Tricomi, E., Balleine, & O'Doherty, J. P. (2009). A specific role for posterior dorsolateral striatum in human habit learning. *European Journal of Neuroscience*, 29(11), 2225–2232. https://doi.org/10.1111/j.1460-9568.2009.06796.x
- Trullier, O., & Meyer, J.-A. (2000). Animat navigation using a cognitive graph. *Biological Cybernetics*, 83(3), 271–285. https://doi.org/10.1007/s004220000170
- Tsao, A., Sugar, J., Lu, L., Wang, C., Knierim, J. J., Moser, M.-B., & Moser, E. I. (2018). Integrating time from experience in the lateral entorhinal cortex. *Nature*, 561(7721), 57–62. https://doi.org/10.1038/s41586-018-0459-6
- Tulving, E. (2002a). Episodic Memory: From Mind to Brain. *Annual Review of Psychology*, 53(1), 1–25. https://doi.org/10.1146/annurev.psych.53.100901.135114
- Tulving, E. (2002b). Episodic Memory: From Mind to Brain. Annual Review of Psychology, 53(1), 1–25. https://doi.org/10.1146/annurev.psych.53.100901.135114
- Uddin, L. Q. (2015). Salience processing and insular cortical function and dysfunction. *Nature Reviews Neuroscience*, 16(1), 55–61. https://doi.org/10.1038/nrn3857
- Uddin, L. Q. (2016). Salience network of the human brain. Academic Press is an imprint of Elsevier.
- Uddin, L. Q., Iacoboni, M., Lange, C., & Keenan, J. P. (2007). The self and social cognition: The role of cortical midline structures and mirror neurons. *Trends in Cognitive Sciences*, 11(4), 153–157. https://doi.org/10.1016/j.tics.2007.01.001
- Uddin, L. Q., Yeo, B. T. T., & Spreng, R. N. (2019). Towards a Universal Taxonomy of Macro-scale Functional Human Brain Networks. *Brain Topography*, *32*(6), 926–942. https://doi.org/10.1007/s10548-019-00744-6
- Ullman, M. T. (2004). Contributions of memory circuits to language: The declarative/procedural model. *Cognition*, 92(1–2), 231–270. https://doi.org/10.1016/j.cognition.2003.10.008
- Ullman, M. T. (2006). Is Broca's Area Part of a Basal Ganglia Thalamocortical Circuit? *Cortex*, 42(4), 480–485. https://doi.org/10.1016/S0010-9452(08)70382-4
- Ullman, M. T. (2016). The Declarative/Procedural Model. In *Neurobiology of Language* (pp. 953–968). Elsevier. https://doi.org/10.1016/B978-0-12-407794-2.00076-6
- Valentin, V. V., Dickinson, A., & O'Doherty, J. P. (2007). Determining the Neural Substrates of Goal-Directed Learning in the Human Brain. *The Journal of Neuroscience*, 27(15), 4019–4026. https://doi.org/10.1523/JNEUROSCI.0564-07.2007
- van Dijk, L. (2021). Affordances in a Multispecies Entanglement. *Ecological Psychology*, 33(2), 73–89. https://doi.org/10.1080/10407413.2021.1885978
- Van Hoeck, N. (2015). Cognitive neuroscience of human counterfactual reasoning. Frontiers in Human Neuroscience, 9. https://doi.org/10.3389/fnhum.2015.00420
- Van Hoeck, N., Watson, P. D., & Barbey, A. K. (2015). Cognitive neuroscience of human counterfactual reasoning. Frontiers in Human Neuroscience, 9. https://doi.org/10.3389/fnhum.2015.00420
- Van Overwalle, F., & Baetens, K. (2009). Understanding others' actions and goals by mirror and mentalizing systems: A meta-analysis. *NeuroImage*, 48(3), 564–584. https://doi.org/10.1016/j.neuroimage.2009.06.009
- Varela, F. J., Thompson, E., & Rosch, E. (2017). *The embodied mind. Cognitive science and human experience.* MIT Press.
- Vaswani, A., Shazeer, N., Parmar, N., Uszkoreit, J., Jones, L., Gomez, A. N., Kaiser, L., & Polosukhin, I. (2017). Attention Is All You Need (arXiv:1706.03762). arXiv. http://arxiv.org/abs/1706.03762
- Vatansever, D., Menon, D. K., Manktelow, A. E., Sahakian, B. J., & Stamatakis, E. A. (2015). Default mode network connectivity during task execution. *NeuroImage*, 122, 96–104. https://doi.org/10.1016/j.neuroimage.2015.07.053
- Verhagen, A. (2010). *Construal and Perspectivization*. Oxford University Press. https://doi.org/10.1093/oxfordhb/9780199738632.013.0003
- Verhoeff, B. (2016). Two kinds of autism: A comparison of distinct understandings of psychiatric disease. *Medicine, Health Care and Philosophy*, 19(1), 111–123. https://doi.org/10.1007/s11019-015-9655-4
- Vervaeke, J., Mastropietro, C., & Miscevic, F. (2017). Zombies in western culture: A twenty-first century crisis. Open Book Publishers.

- Vikbladh, O. M., Meager, M. R., King, J., Blackmon, K., Devinsky, O., Shohamy, D., Burgess, N., & Daw, N. D. (2019). Hippocampal Contributions to Model-Based Planning and Spatial Memory. *Neuron*, 102(3), 683-693.e4. https://doi.org/10.1016/j.neuron.2019.02.014
- Vincent, J. L., Kahn, I., Snyder, A. Z., Raichle, M. E., & Buckner, R. L. (2008). Evidence for a Frontoparietal Control System Revealed by Intrinsic Functional Connectivity. *Journal of Neurophysiology*, 100(6), 3328–3342. https://doi.org/10.1152/jn.90355.2008
- Vinyals, O., Babuschkin, I., Czarnecki, W. M., Mathieu, M., Dudzik, A., Chung, J., Choi, D. H., Powell, R., Ewalds, T., Georgiev, P., Oh, J., Horgan, D., Kroiss, M., Danihelka, I., Huang, A., Sifre, L., Cai, T., Agapiou, J. P., Jaderberg, M., ... Silver, D. (2019). Grandmaster level in StarCraft II using multi-agent reinforcement learning. *Nature*, 575(7782), 350–354. https://doi.org/10.1038/s41586-019-1724-z
- Vishne, G., Gerber, E. M., Knight, R. T., & Deouell, L. Y. (2023). Distinct ventral stream and prefrontal cortex representational dynamics during sustained conscious visual perception. *Cell Reports*, 42(7), 112752. https://doi.org/10.1016/j.celrep.2023.112752

Von Helmholtz, H. (1886). Handbuch der physiologishen optik. Leipzig Dritter Abschnitt, 204-205.

- Vygotsky, L. S., & Cole, M. (1978). *Mind in society: Development of higher psychological processes*. Harvard university press.
- Wada, S., Honma, M., Masaoka, Y., Yoshida, M., Koiwa, N., Sugiyama, H., Iizuka, N., Kubota, S., Kokudai, Y., Yoshikawa, A., Kamijo, S., Kamimura, S., Ida, M., Ono, K., Onda, H., & Izumizaki, M. (2021). Volume of the right supramarginal gyrus is associated with a maintenance of emotion recognition ability. *PLOS ONE*, *16*(7), e0254623. https://doi.org/10.1371/journal.pone.0254623
- Wagatsuma, A., Okuyama, T., Sun, C., Smith, L. M., Abe, K., & Tonegawa, S. (2018). Locus coeruleus input to hippocampal CA3 drives single-trial learning of a novel context. *Proceedings of the National Academy of Sciences*, 115(2). https://doi.org/10.1073/pnas.1714082115
- Wallace, M. T., Wilkinson, L. K., & Stein, B. E. (1996). Representation and integration of multiple sensory inputs in primate superior colliculus. *Journal of Neurophysiology*, 76(2), 1246–1266. https://doi.org/10.1152/jn.1996.76.2.1246
- Wang, Y., Metoki, A., Xia, Y., Zang, Y., He, Y., & Olson, I. R. (2021). A large-scale structural and functional connectome of social mentalizing. *NeuroImage*, 236, 118115. https://doi.org/10.1016/j.neuroimage.2021.118115
- Washington, S. D., Gordon, E. M., Brar, J., Warburton, S., Sawyer, A. T., Wolfe, A., Mease-Ference, E. R., Girton, L., Hailu, A., Mbwana, J., Gaillard, W. D., Kalbfleisch, M. L., & VanMeter, J. W. (2014). Dysmaturation of the default mode network in autism: Dysmaturation of Default Network in Autism. *Human Brain Mapping*, 35(4), 1284–1296. https://doi.org/10.1002/hbm.22252
- Watabe-Uchida, M., Eshel, N., & Uchida, N. (2017). Neural Circuitry of Reward Prediction Error. Annual Review of Neuroscience, 40(1), 373–394. https://doi.org/10.1146/annurev-neuro-072116-031109
- Watabe-Uchida, M., & Uchida, N. (2018). Multiple Dopamine Systems: Weal and Woe of Dopamine. Cold Spring Harbor Symposia on Quantitative Biology, 83, 83–95. https://doi.org/10.1101/sqb.2018.83.037648
- Watkins, N. W. (2018). (A)phantasia and severely deficient autobiographical memory: Scientific and personal perspectives. Cortex, 105, 41–52. https://doi.org/10.1016/j.cortex.2017.10.010
- Wei, T., Liang, X., He, Y., Zang, Y., Han, Z., Caramazza, A., & Bi, Y. (2012). Predicting Conceptual Processing Capacity from Spontaneous Neuronal Activity of the Left Middle Temporal Gyrus. *The Journal of Neuroscience*, 32(2), 481–489. https://doi.org/10.1523/JNEUROSCI.1953-11.2012
- Weir, A. A. S., & Kacelnik, A. (2006). A New Caledonian crow (Corvus moneduloides) creatively redesigns tools by bending or unbending aluminium strips. *Animal Cognition*, 9(4), 317–334. https://doi.org/10.1007/s10071-006-0052-5
- Weiss, M., Worling, D., & Wasdell, M. (2003). A chart review study of the Inattentive and Combined Types of ADHD. *Journal of Attention Disorders*, 7(1), 1–9. https://doi.org/10.1177/108705470300700101
- Werner, K. (2021). Cognitive confinement: Theoretical considerations on the construction of a cognitive niche, and on how it can go wrong. *Synthese*, 198(7), 6297–6328. https://doi.org/10.1007/s11229-019-02464-7
- Whittington, J. C. R., McCaffary, D., Bakermans, J. J. W., & Behrens, T. E. J. (2022). How to build a cognitive map. *Nature Neuroscience*, 25(10), 1257–1272. https://doi.org/10.1038/s41593-022-01153-y

- Whittington, J. C. R., Muller, T. H., Mark, S., Chen, G., Barry, C., Burgess, N., & Behrens, T. E. J. (2020). The Tolman-Eichenbaum Machine: Unifying Space and Relational Memory through Generalization in the Hippocampal Formation. *Cell*, 183(5), 1249-1263.e23. https://doi.org/10.1016/j.cell.2020.10.024
- Wiese, W., & Friston, K. (2021). Examining the Continuity between Life and Mind: Is There a Continuity between Autopoietic Intentionality and Representationality? *Philosophies*, 6(1), 18. https://doi.org/10.3390/philosophies6010018
- Wilk, H. A., Ezekiel, F., & Morton, J. B. (2012). Brain regions associated with moment-to-moment adjustments in control and stable task-set maintenance. *NeuroImage*, 59(2), 1960–1967. https://doi.org/10.1016/j.neuroimage.2011.09.011
- Williams, J. N. (2020). The Neuroscience of Implicit Learning. *Language Learning*, 70(S2), 255–307. https://doi.org/10.1111/lang.12405
- Williams, Z. M., & Eskandar, E. N. (2006). Selective enhancement of associative learning by microstimulation of the anterior caudate. *Nature Neuroscience*, 9(4), 562–568. https://doi.org/10.1038/nn1662
- Wilson, R. C., Bonawitz, E., Costa, V. D., & Ebitz, R. B. (2021). Balancing exploration and exploitation with information and randomization. *Current Opinion in Behavioral Sciences*, 38, 49–56. https://doi.org/10.1016/j.cobeha.2020.10.001
- Wilson, R. C., Geana, A., White, J. M., Ludvig, E. A., & Cohen, J. D. (2014). Humans use directed and random exploration to solve the explore–exploit dilemma. *Journal of Experimental Psychology: General*, 143(6), 2074–2081. https://doi.org/10.1037/a0038199
- Wilson, R. C., Takahashi, Y. K., Schoenbaum, G., & Niv, Y. (2014). Orbitofrontal Cortex as a Cognitive Map of Task Space. *Neuron*, 81(2), 267–279. https://doi.org/10.1016/j.neuron.2013.11.005
- Winter, D. A. (2009). Biomechanics and motor control of human movement (4th ed). Wiley.
- Wolff, P., & Holmes, K. J. (2011). Linguistic relativity. WIREs Cognitive Science, 2(3), 253–265. https://doi.org/10.1002/wcs.104
- Wood, W., & Neal, D. T. (2007). A new look at habits and the habit-goal interface. *Psychological Review*, 114(4), 843–863. https://doi.org/10.1037/0033-295X.114.4.843
- Wunderlich, K., Dayan, P., & Dolan, R. J. (2012). Mapping value based planning and extensively trained choice in the human brain. *Nature Neuroscience*, 15(5), 786–791. https://doi.org/10.1038/nn.3068
- Wunderlich, K., Smittenaar, P., & Dolan, R. J. (2012). Dopamine Enhances Model-Based over Model-Free Choice Behavior. *Neuron*, 75(3), 418–424. https://doi.org/10.1016/j.neuron.2012.03.042
- Xia, Y., Driscoll, J. R., Wilbrecht, L., Margolis, E. B., Fields, H. L., & Hjelmstad, G. O. (2011). Nucleus Accumbens Medium Spiny Neurons Target Non-Dopaminergic Neurons in the Ventral Tegmental Area. *The Journal of Neuroscience*, 31(21), 7811–7816. https://doi.org/10.1523/JNEUROSCI.1504-11.2011
- Xu, P., Chen, A., Li, Y., Xing, X., & Lu, H. (2019). Medial prefrontal cortex in neurological diseases. *Physiological Genomics*, 51(9), 432–442. https://doi.org/10.1152/physiolgenomics.00006.2019
- Xu, Y., & Corkin, S. (2001). H. M. revisits the Tower of Hanoi Puzzle. *Neuropsychology*, 15(1), 69–79. https://doi.org/10.1037/0894-4105.15.1.69
- Yeo, B. T., Krienen, F. M., Sepulcre, J., Sabuncu, M. R., Lashkari, D., Hollinshead, M., Roffman, J. L., Smoller, J. W., Zöllei, L., Polimeni, J. R., Fischl, B., Liu, H., & Buckner, R. L. (2011). The organization of the human cerebral cortex estimated by intrinsic functional connectivity. *Journal of Neurophysiology*, 106(3), 1125–1165. https://doi.org/10.1152/jn.00338.2011
- Yeshurun, Y., Nguyen, M., & Hasson, U. (2021). The default mode network: Where the idiosyncratic self meets the shared social world. *Nature Reviews Neuroscience*, 22(3), 181–192. https://doi.org/10.1038/s41583-020-00420-w
- Yin, H. H., & Knowlton, B. J. (2006). The role of the basal ganglia in habit formation. Nature Reviews Neuroscience, 7(6), 464–476. https://doi.org/10.1038/nrn1919
- Yin, H. H., Ostlund, S. B., Knowlton, B. J., & Balleine. (2005). The role of the dorsomedial striatum in instrumental conditioning: Striatum and instrumental conditioning. *European Journal of Neuroscience*, 22(2), 513–523. https://doi.org/10.1111/j.1460-9568.2005.04218.x
- Yokoi, A., Arbuckle, S. A., & Diedrichsen, J. (2018). The Role of Human Primary Motor Cortex in the Production of Skilled Finger Sequences. *The Journal of Neuroscience*, 38(6), 1430–1442. https://doi.org/10.1523/JNEUROSCI.2798-17.2017
- Zacks, J. M., Braver, T. S., Sheridan, M. A., Donaldson, D. I., Snyder, A. Z., Ollinger, J. M., Buckner, R. L., & Raichle, M. E. (2001). Human brain activity time-locked to perceptual event boundaries. *Nature Neuroscience*, 4(6), 651–655. https://doi.org/10.1038/88486

- Zacks, J. M., Speer, N. K., Swallow, K. M., Braver, T. S., & Reynolds, J. R. (2007). Event perception: A mind-brain perspective. *Psychological Bulletin*, 133(2), 273–293. https://doi.org/10.1037/0033-2909.133.2.273
- Zacks, J. M., & Tversky, B. (2001). Event structure in perception and conception. Psychological Buletin.
- Zadbood, A., Chen, J., Leong, Y. C., Norman, K. A., & Hasson, U. (2017). How We Transmit Memories to Other Brains: Constructing Shared Neural Representations Via Communication. *Cerebral Cortex*, 27(10), 4988–5000. https://doi.org/10.1093/cercor/bhx202
- Zajchowski, C. A. B., Schwab, K. A., & Dustin, D. L. (2017). The Experiencing Self and the Remembering Self: Implications for Leisure Science. *Leisure Sciences*, 39(6), 561–568. https://doi.org/10.1080/01490400.2016.1209140
- Zeki, S. (2015). Area V5, a microcosm of the visual brain. *Frontiers in Integrative Neuroscience*, 9. https://doi.org/10.3389/fnint.2015.00021
- Zhang, M., Bernhardt, B. C., Wang, X., Varga, D., Krieger-Redwood, K., Royer, J., Rodríguez-Cruces, R., Vos De Wael, R., Margulies, D. S., Smallwood, J., & Jefferies, E. (2022). Perceptual coupling and decoupling of the default mode network during mind-wandering and reading. *eLife*, 11, e74011. https://doi.org/10.7554/eLife.74011
- Zhang, Y., Larcher, K. M.-H., Misic, B., & Dagher, A. (2017). Anatomical and functional organization of the human substantia nigra and its connections. *eLife*, *6*, e26653. https://doi.org/10.7554/eLife.26653
- Zheng, J., Schjetnan, A. G. P., Yebra, M., Gomes, B. A., Mosher, C. P., Kalia, S. K., Valiante, T. A., Mamelak, A. N., Kreiman, G., & Rutishauser, U. (2022). Neurons detect cognitive boundaries to structure episodic memories in humans. *Nature Neuroscience*, 25(3), 358–368. https://doi.org/10.1038/s41593-022-01020-w
- Zhou, Q.-Y., & Palmiter, R. D. (1995). Dopamine-deficient mice are severely hypoactive, adipsic, and aphagic. Cell, 83(7), 1197–1209. https://doi.org/10.1016/0092-8674(95)90145-0